THE SCHOOL BOARD OF BROWARD COUNTY, FLORIDA OFFICE OF THE SUPERINTENDENT

ROBERT W. RUNCIE SUPERINTENDENT OF SCHOOLS

April 13, 2020

TO:

School Board Members

FROM:

Frank Girardi

Executive Director

Capital Programs

VIA:

Robert W. Runcie

Superintendent of Schools

SUBJECT: Revision to Item 8, The Office of Capital Program is Requesting Approval of the PM RFQ Form and Authorization to Advertise to Award One or More Program Managers - RFQ FY20-192, for the April 14, 2020, Special School Board

Meeting

Below is a synopsis of changes that have been made to Exhibit #3, RFQ FY20-192 - ONLINE:

Section 6.0 Scope of Services:

- Last paragraph completed list of areas
- 6.2.1 Last sentence added clarifying language to make consistent with contract language regarding Principal-in-Charge and PM Lead
- 6.3.4 Added "and/or Principal-in-Charge" to second paragraph
- 6.3.6 Added clarifying language related to roofing

Agreement revisions:

The draft agreement has also been revised. A red-line copy of this revised agreement is also included to reflect where changes have occurred.

RWR/FLG:djc Attachments

c: Senior Leadership Team



REQUEST FOR QUALIFICATIONS

RFQ # FY20-192

PROGRAM MANAGER - OWNER'S REPRESENTATIVE SERVICES



The School Board of Broward County, Florida

Donna P. Korn, Chair Dr. Rosalind Osgood, Vice Chair

Lori Alhadeff Robin Bartleman Heather P. Brinkworth Patricia Good Laurie Rich Levinson Ann Murray Nora Rupert

Robert W. Runcie, Superintendent of Schools

Frank L. Girardi, Executive Director, Capital Programs

Mary C. Coker, Director, Procurement & Warehousing Services

Barbara J. Myrick, General Counsel

Office of Facilities and Construction Procurement and Warehousing Services

NOTICES TO ALL SUBMITTING FIRMS

Cone of Silence and Lobbyist Activities

For Cone of Silence requirements and for Lobbyist Activity restrictions refer to Section 1, Introduction and General Information lines labeled Cone of Silence and Lobbyist Activities. Line numbers 1.10 and 1.11 and to School Board Policies 3320 and 1100B.

THE SCHOOL BOARD OF BROWARD COUNTY, FLORIDA
Procurement and Warehousing Services Department
7720 W. Oakland Park Boulevard, Suite 323
Sunrise, Florida 33351-6704

REQUEST FOR QUALIFICATIONS (RFQ) RFQ # FY20-192

PROGRAM MANAGER – OWNER'S REPRESENTATIVE SERVICES

Public Announcement

Notice is hereby given that SEALED QUALIFICATIONS STATEMENTS will be received by THE SCHOOL BOARD OF BROWARD COUNTY, FLORIDA, at the Procurement & Warehousing Services Department, 7720 W. Oakland Park Blvd., Sunrise, FL 33351, Suite 323 up to 02:00 PM on May 19, 2020, and opened immediately thereafter in Bid Room #1, #2, #3 and/or #4, for provision of all materials and labor necessary for Program Manager – Owner's Representative Services as described herein. A **MANDATORY** presubmittal conference will take place **VIRTUALLY** utilizing the Microsoft Teams Platform at 10:00 AM on April 22, 2020. All interested parties are invited to attend. See the RFQ Section 2.0 – Calendar for additional information on the mandatory pre-submittal conference.

Bid details can be found online by visiting the following website:

https://www.demandstar.com/supplier/bids/agency_inc/bid_list.asp?f=search&LP=BB&mi=10150

To be advertised on April 15, 2020

REQUEST FOR QUALIFICATIONS (RFQ) RFQ # FY20-192

PROGRAM MANAGER – OWNER'S REPRESENTATIVE SERVICES

RFQ Issue Date: April 15, 2020

Description of Scope:

The District has determined that engaging the assistance of a team of construction industry professionals organized in one (1) or two (2) Program Manager - Owner's Representative (PMOR) agents is the most flexible, cost-effective means to achieve the coordination and synergy necessary to meet the demands of the Schools. A PMOR offers flexibility through its capacity to engage professionals with specific expertise at appropriate points in the planning and implementation of projects. The detailed scope of services can be found in Section 6.0 of this RFQ.

The SBBC intends to enter into an agreement with one (1) or two (2) qualified firms to perform Program Manager – Owner's Representative Services. Services shall include all types and manner of Program Manager – Owner's Representative Services related to Owner's Design and Construction programs as required and directed by the District.

NOTICES TO ALL SUBMITTING FIRMS

Cone of Silence and Lobbyist Activities

For Cone of Silence requirements and for Lobbyist Activity restrictions refer to Section 1, Introduction and General Information lines labeled Cone of Silence and Lobbyist Activities. Line numbers 1.10 and 1.11 and to School Board Policies 3320 and 1100B.

THE SCHOOL BOARD OF BROWARD COUNTY, FLORIDA Procurement and Warehousing Services Department 7720 W. Oakland Park Boulevard, Suite 323 Sunrise, Florida 33351-6704

TABLE OF CONTENTS

<u>Section</u>	<u>P</u>	<u>Page</u>
	RFQ Cover Sheet	0
	Notice of RFQ	1
	RFQ – Description of Scope	2
	Table of Contents	3
1.0	Introduction and General Information	5
2.0	Calendar	8
3.0	RFQ Conditions	. 10
4.0	Required Qualifications Statement Format and Response Information	. 17
5.0	Evaluation of Qualifications	. 26
6.0	Scope of Services	. 37
<u>Attachmen</u>	nts_	
Attachment		
Attachment	<u>t A.1</u> Small/Minority/Women Business Enterprise (S/M/WBE) Subcontractor Parti	cipation Sche

Attachment A	Supplier Diversity Outreach Program Guidelines (Document 00467)
Attachment A.1	Small/Minority/Women Business Enterprise (S/M/WBE) Subcontractor Participation Schedule
Attachment A.2	Statement of Intent to Perform as an S/M/WBE Subcontractor
Attachment A.3	S/M/WBE Monthly Subcontractor Utilization Report
Attachment A.4	S/M/WBE Good Faith Effort Form
Attachment B	Required Response Form
Attachment C	Proposer's Request for Information Form (for use by Submitting Firm)
Attachment D	Relevant Program / Project / Reference Form
Attachment E	Conflict of Interest
Attachment F	Authorization to Proceed Form
Attachment G	ACH Payment Agreement Form
Attachment H	Sample Program Manager – Owner's Representative (PMOR) Agreement
Attachment I	Broward County Public Schools Capital Program - Active Projects by Phase
Attachment J	Sample Program Manager – Owner's Representative (PMOR) Evaluation

RFQ Online Attachment Links

Submitting Firms need to be familiar

Online Documents: The following documents shall be downloaded by the Program Manager ("Submitting Firm" or "PMOR") and are considered as RFQ and Contract Attachments, and requirements of the Contract:

Division 0 - Specifications

http://www.broward.k12.fl.us/constructioncontracts/D0docs.html

Design Standards

http://www.broward.k12.fl.us/constructioncontracts/DivisiononeDesignStandards.html

- SBBC Design Criteria
 - Specifications Division 1 through 17
 - Document Submittal Checklist for Plan Review
 - o Design Guidelines

Note the following:

For SBBC's Design and Materials Standards downloaded from SBBC's website – the word 'Contractor' shall be interpreted as the word 'Construction Manager' in every instance that the word Contractor is used in these standards.

 State Requirements for Educational Facilities (SREF) latest edition: http://www.fldoe.org/edfacil/sref.asp

SECTION 1.0 - INTRODUCTION AND GENERAL INFORMATION

- The School Board of Broward County, Florida (hereinafter referred to as "SBBC") desires to receive Qualifications Statements for Program Manager Owner's Representative Services for program management services for educational facilities projects district-wide, to include all types and manner of program management services related to design and construction programs as required and directed by the District, from Submitting Firms qualified and interested in providing professional management services as described herein. The SBBC intends to enter into an agreement with one (1) or two (2) qualified firms to perform Program Manager Owner's Representative Services. Submitting Firms may be referred to as "Program Manager" or "PMOR" throughout this document.
- 1.2 Contract Term and Type: SBBC intends to establish a three (3) year contract term commencing in or about June 2020 through June 2023. The term of the contract may, by mutual agreement between SBBC and the Awardee, be extended for two (2) additional one-year periods and one (1) additional six-month period beyond the expiration date of the renewal period. Procurement & Warehousing Services Department, will, if considering renewing, request a letter of intent to renew from each Awardee, before the end of the current contract period. The Awardee will be notified when the recommendation has been acted upon by the School Board.
 - 1.2.1 Should the current, active PMOR be recommended for award, the new contract term shall not commence prior to August 19, 2020 to align with the current, active Program Manager Owner's Representative Services Agreement.
- Evaluation and Award: All Qualifications Statements will be evaluated by the Qualifications Selection Evaluation Committee (QSEC) and District staff based upon the information submitted by Submitting Firms in response to this RFQ and in accordance with the evaluation criteria established in Section 5.0. Based upon the evaluation of Qualifications Statements, QSEC will recommend one (1) or two (2) Submitting Firm(s) to the Superintendent of Schools who may then recommend one (1) or two (2) Submitting Firm(s) to SBBC for award.
- Questions and Interpretations: Any questions, requests for clarifications or interpretations regarding any portion of this RFQ (or reporting errors, inconsistencies, or ambiguities) during the Submittal preparation period shall be received no later than the date stated in Section 2.0 (which will not be less than ten (10) business days prior to the deadline due date for submitting the completed RFQ response), and no questions or clarifications will be considered after this date. Program Managers must submit Document 00220 Proposer's Request for Information (RFI) Form (Attachment C) for all RFI requests and all RFI requests must be typed or printed. Questions not submitted on Document 00220 will not be considered. RFI requests must be for a single RFQ, RFI requests for multiple RFQs may not be considered. Any questions which require a response which amends the RFQ document in any manner will be answered via Addendum by the Procurement and Warehousing Services Department and provided to all Submitting Firms. Address questions or requests for clarifications in writing to:

Luis E. Perez
Purchasing Agent III
Procurement & Warehousing Services
7720 West Oakland Park Blvd - Suite 323
Sunrise, FL 33351
754-321-0936 Fax

E-mail: Luis.Perez@browardschools.com

- 1. 5 Any verbal or written information, which is obtained other than by information in this RFQ document or by Addenda, shall not be binding on SBBC.
- 1.6 **Qualifications Statement Format:** Any response to the RFQ shall follow the format and include the information as identified in Section 4.0 of this RFQ.
- 1.7 **Qualifications Statement Package Requirements:**
 - One (1) complete, original printed hard-copy Qualifications Statement (clearly labeled as "original").

RFQ FY20-192 Page 5 of 54
Program Manager - Owner's Representative Services

Version Date: 04/14/2020

- One (1) complete, original electronic version (clearly labeled as "original"), which must be identical to the hard-copy original. Electronic version should clearly identify the RFQ # and Firm Name.
- Two (2) complete, electronic version copies (clearly **labeled as "copy"**), which must be identical to the original electronic version. Electronic version should clearly identify the RFQ # and Firm Name.
- Six (6) additional printed hard-copies (which must be identical to the original Qualifications Statement except they shall be labeled as "copy").
- Qualifications Statements shall be submitted in a sealed envelope (package, box, etc.) with the RFQ number and description clearly identified by label on the Package along with the Submitting Firms Name.
- 1.8 Preparation Cost of Qualifications Statement: The Submitting Firm is solely responsible for any and all costs associated with responding to this RFQ. SBBC will not reimburse any Submitting Firm for any costs associated with the preparation and submittal of any Qualifications Statement, or for any travel and per diem costs that are incurred by any Submitting Firm.
- 1.9 It is the sole responsibility of the Submitting Firm to assure it has received the entire Qualifications Statement and any and all Addenda.
- 1.10 It is the sole responsibility of the Submitting Firm to assure that its Qualifications Statement is time stamped in the Procurement and Warehousing Services Department on or before the date and time stated in Section 2.0.
- 1.11 No submissions made after the Qualifications Statement opening, amending or supplementing the Qualifications Statement, will be considered.
- 1.12 Withdrawal of Qualifications Statement: A Qualifications Statement may be withdrawn at any time, in writing, prior to the start of QSEC Part A for the applicable RFQ.
- 1.13 **Gratuities:** Submitting Firms shall not provide any gratuities, favors, or anything of monetary value to any official, employee, or agent of SBBC; including any School Board Member, Superintendent of Schools and any QSEC Members, for any reason during this entire procurement process.
- 1.14 **Cone of Silence:** Any Submitting Firm, or lobbyist for a Submitting Firm, is prohibited from having any communications (except as provided in this rule) concerning any solicitation for a competitive procurement with any School Board Member, the Superintendent, any QSEC Member, or any other school district employee after the Procurement and Warehousing Services Department releases a solicitation to the general public. All communications regarding this solicitation shall be directed to the designated Purchasing Agent unless otherwise notified by the Procurement and Warehousing Services Department. This "Cone of Silence" period shall go into effect upon the time of release of the solicitation and shall remain in effect until SBBC either awards a contract or rejects all Qualifications Statements. Further, any Submitting Firm, its principals, or its lobbyists shall not offer campaign contributions to School Board Members or offer contributions to School Board Members for campaigns of other candidates for political office during the period in which the vendor is attempting to sell goods or services to SBBC. This period of limitation of offering campaign contributions shall commence at the time of the "cone of silence" period for any solicitation for a competitive procurement as described by School Board Policy 3320, Part II, Section GG as well as School Board Policy 1007, Section 5.4 - Campaign Contribution Fundraising. Any Submitting Firm or its lobbyist who violates this provision shall cause the Submitting Firm's Qualifications Statement to be considered non-responsive and therefore be ineligible for award. This prohibition does not apply to:
 - Telephone calls to the Procurement & Warehousing Services staff to request copies of this RFQ, to confirm attendance, or request directions regarding an interview notification received;
 - Delivery of the Respondent's Submittal;
 - Discussion at the interview:
 - Delivery of written questions about the RFQ:
 - Review of background/contract documents at the staff offices;
 - Public Records Requests; and/or
 - Bid protest proceedings.

RFQ FY20-192 Version Date: 04/14/2020

- Lobbyist Activities: In accordance with School Board Policy 1100B, as currently enacted or as amended from time to time, persons acting as lobbyists must state, at the beginning of their presentation, letter, telephone call, e-mail or facsimile transmission to School Board Members, Superintendent or Members of Senior Management, the group, association, organization or business interest she/he is representing.
 - 1.15.1 A lobbyist is defined as a person who, for immediate or subsequent compensation (e.g., monetary profit/personal gain), represents a public or private group, association, organization or business interest and engages in efforts to influence School Board Members on matters within their official jurisdiction.
 - 1.15.2 A lobbyist is not considered to be a person representing school allied groups (e.g., PTA, DAC, Band Booster Associations, etc.) nor a public official acting in her/his official capacity.
 - 1.15.3 Lobbyists shall annually (July 1) disclose in each instance and for each client prior to any lobbying activities, their identity and activities by completing the lobbyist statement form which can be obtained from official School Board Records, School Board Member's Offices or the Superintendent's Office and will be recorded on SBBC's website, www.browardschools.com.
 - 1.15.4 The lobbyist must disclose any direct business association with any current elected or appointed official or employee of SBBC or any immediate family member of such elected or appointed official or employee of SBBC.
 - 1.15.5 Senior-level employees and/or School Board Members are prohibited from lobbying activities for two (2) years after resignation or retirement or expiration of their term of office.
 - 1.15.6 The Office of the Superintendent shall keep a current list of persons who have submitted the lobbyist statement form.
- 1.16 Certification Regarding Scrutinized Activities: By submitting its bid, proposal or qualifications statement, the Submitting Firm certifies that it is not participating in a boycott of Israel, it is not on the List of Scrutinized Companies that Boycott Israel; it is not on the List of Scrutinized Companies with Activities in Sudan; it is not on the List of Scrutinized Companies with Activities in the Iran Petroleum Energy Sector; and that it is not engaged in business operations in Cuba or Syria.

SECTION 2.0 CALENDAR

04/15/2020 Issuance of RFQ # - FY20-192 – Program Manager – Owner's Representative Services

04/22/2020 MANDATORY Pre-Submittal Conference

At 10:00 a.m. ET*. through

the Microsoft Teams Virtual Platform. To register, EMAIL the names, email addresses,

and phone numbers of the individuals that will be participating to:

Luis.Perez@browardschools.com

05/05/2020 Written questions due:

On or before 5:00 p.m. ET

Procurement and Warehousing Services Department

7720 West Oakland Park Blvd., Suite 323,

Sunrise, Florida 33351-6704

Or via email to: Luis.Perez@browardschools.com

05/19/2020 Qualifications Statements due:

At or before 2:00 p.m. ET at

Procurement and Warehousing Services Department.

7720 West Oakland Park Blvd., Suite 323,

Sunrise, Florida 33351-6704

06/01/2020 Submitting Firm Presentations Final Draft due:

At or before 2:00 p.m. ET

Via email to Luis.Perez@browardschools.com

06/03/2020 QSEC Meeting** - Selection Committee reviews Qualifications and develops list of

shortlisted firms:

At 9:00 a.m. ET* at

Procurement and Warehousing Services Department 7720 West Oakland Park Boulevard, Suite 323

Sunrise, Florida 33351-6704

Approximately 5 business days

after Committee Review

Posting of Evaluation Committee's Recommendation to negotiate. Recommendation to

award shall be posted after the completion of negotiations.

06/10/2020 & 06/11/2020 Negotiation Meeting*** (Time - **TBD**)

At Procurement and Warehousing Services Department

7720 West Oakland Park Boulevard, Suite 323

Sunrise, Florida 33351-6704***

06/23/2020 Tentative School Board Award Date.

Individuals with disabilities requesting accommodations under the Americans with Disabilities Act (ADA) may call Equal Educational Opportunities (EEO) at 754-321-2150 or Teletype Machine (TTY) 754-321-2158.

RFQ FY20-192 Page 8 of 54

^{*}These are public meetings. SBBC prohibits any policy or procedure which results in discrimination on the basis of age, color, disability, gender identity, gender expression, national origin, marital status, race, religion, sex or sexual orientation. Individuals who wish to file a discrimination and/or harassment complaint may call the Executive Director, Benefits & EEO Compliance at 754-321-2150 or Teletype Machine (TTY) 754-321-2158.

** QSEC Meeting – The date of the selection committee review meeting is approximate and may be changed via Addendum or Notice. Pursuant to the requirement of Section 287.055(4)(a), Florida Statutes, the District shall evaluate Qualifications Statements regarding the proposed projects and shall conduct discussions with no fewer than three (3) Submitting Firms based on the approach, experience, and ability to furnish the required services presented in their Qualifications Statement. The QSEC Meeting will be broken down into two parts: Part A and Part B.

Part A: Committee members will first review and score the Qualifications Statements. Submitting Firms may be present during this portion of the meeting, but no additional handouts, paperwork, or presentations from Submitting Firms will be permitted.

Part B: Following the review of the Qualifications Statements, committee members will conduct and score discussions with all firms. It is estimated that the discussions portion of the meeting will be comprised of each firm giving a 20-minute presentation to QSEC followed by Questions from the committee members. Failure of a Submitting Firm to be present for its scheduled presentation time slot shall result in the disqualification of the Submitting Firm and entire evaluation of the firm will not take place. Accordingly, District staff shall establish a meeting schedule including specific randomly selected time slots for Submitting Firm presentations to QSEC. No additional handouts or paperwork from Submitting Firms, except the printout of the PowerPoint presentation, will be permitted.

NOTE: This meeting schedule and assigned time slots are only estimates and are subject to change at the sole discretion of QSEC depending upon the progress of the meeting. As such, Submitting Firms are strongly encouraged, but not required, to have a representative be physically present at the designated meeting site location for the entire meeting (in addition to its scheduled time slot) to be available for presentations and to attend the opening and closing portions of the meeting.

***Representatives from the selected Firm or Firm(s) selected shall be available for negotiations on both days. Actual meeting date and time will be assigned after QSEC meeting. Location is subject to change, but the meeting will take place in Broward County, FL.

SECTION 3.0 - RFQ CONDITIONS

- 3.1 Evaluation Committees and Qualifications Statements: SBBC complies with Sections 119.071 and 286.0113, Florida Statutes, in the review and award of contracts.
- 3.2 **Public Record:** Each Submitting Firm acknowledges that all information contained within its Qualifications Statement is part of the public domain as defined by the State of Florida Sunshine and Public Record Laws and is not exempt from public inspection or copying.
- 3.3 **Governing Law:** This RFQ, and any award(s) resulting from it, shall be interpreted and construed in accordance with the laws of the State of Florida. Any protests arising from this RFQ shall be subject to Section 120.57(3), Florida Statutes. Any disputes or controversies arising out of a contract awarded under this RFQ shall be submitted to the jurisdiction of the Circuit Court of the Seventeenth Judicial Circuit in and for Broward County, Florida or to the United States District Court for the Southern District of Florida.
- **Advertising:** By submitting a response to this RFQ, Submitting Firm agrees not to use any award under this RFQ as a part of 3.4 any commercial advertising without prior written approval of SBBC.
- 3.5 **Billing Instructions and Payment:** All payments made to the Submitting Firm shall be made by ACH (Automated Clearing House). Monthly invoices shall be submitted to the Director, Construction for review and approval. Refer to the Agreement for the specifics. The ACH Payment Agreement Form. This form shall be submitted by the awardee at the time of the execution of the Contract.
- 3.6 Contract Value: No quarantee is given or implied as to the total dollar value or work as a result of this RFQ. SBBC is not obligated to place any order for services performed as a result of this award. Order placement will be based upon the needs and in the best interest of SBBC.
- 3.7 Conflict of Interest and Conflicting Employment or Contractual Relationship: Section 112.313 (3) and (7), Florida Statutes, sets forth restrictions upon the ability of SBBC employees acting in a private capacity to rent, lease, or sell any realty. goods, or services to SBBC. It also places restrictions upon SBBC employees concerning outside employment or contractual relationships with any business entity which is doing business with SBBC. Each Submitting Firm is to disclose any employees it has who are also SBBC employees. Refer to the Conflict of Interest Form. This executed document shall be submitted at the time of Contract Execution. Any employees identified by the Submitting Firm should obtain legal advice as to their obligations and restrictions under Section 112.313 (3) and (7), Florida Statutes.
- 3.8 **Disputes:**
 - In the event of a conflict between the documents, the order of priority of the documents shall be as follows:
 - 1. The Agreement resulting from the award of this RFQ (if applicable); then
 - 2. Addenda released for this RFQ, with the latest Addendum taking precedence; then
 - 3. the RFQ Documents: then
 - 4. Awardee's Qualifications Statement.
 - 3.8.2 In case of any other doubt or difference of opinion, the decision of SBBC shall be put in writing and shall be final and binding on both parties.
- 3.9 **Insurance and Liability:** Refer to Submitting Firm Agreement for Professional Insurance Requirements.
- 3.10 Public Entity Crimes: Section 287.133(2)(a), Florida Statutes, as currently enacted or as amended from time to time, states that a person or affiliate who has been placed on the convicted vendor list following a conviction for a public entity crime may not submit a Qualifications Statement on a contract to provide any goods or services to a public entity, may not submit a Qualifications Statement on a contract with a public entity for the construction or repair of a public building or public work, may not submit a Qualifications Statement on leases of real property to a public entity, may not be awarded or perform work as a contractor, supplier, subcontractor, or consultant under a contract with any public entity, and may not transact business with any public entity in excess of the threshold amount provided in Section 287.017, Florida Statutes, for CATEGORY TWO [currently \$25,000] for a period of thirty-six (36) months from the date of being placed on the convicted vendor list.

RFQ FY20-192 Page 10 of 54

Program Manager - Owner's Representative Services

3.10.1 By submitting a Qualifications Statement, the proposer represents that restrictions related to public entity crimes stated in Section 287.133(2) (a), Florida Statutes, do not apply to either his own company, or that of his subcontractors or suppliers.

3.11 Supplier Diversity Outreach Program (SDOP) Guidelines and Standard Operating Procedures Information:

SBBC has implemented School Board Policy 3330 Supplier Diversity Outreach Program. The provisions of the Policy shall apply to all competitive solicitations for construction, professional services, commodities and other contractual services, and any resulting contract documents including change orders, and amendments. **Refer to SDOP Guidelines (See Attachment A).**

This RFQ includes participation goals for SBBC-certified M/WBEs. Refer to Section 4.2.6 and Section 5.0 Evaluation of Qualification – 5.2.3 Scoring of M/WBE Participation – Section E for additional information.

For information regarding M/WBE certification, or to obtain information on locating certified M/WBEs, contact Supplier Diversity Outreach Program (SDOP) at 754-321-0505 or visit BrowardSchools.com/SDOP.

3.11.1 <u>Nondiscrimination</u>:

Each Submitting Firm hereby certifies and agrees that the following information is correct: In preparing its response to this project, the Submitting Firm has considered all proposals submitted from qualified, potential subcontractors and suppliers, and has not engaged in "discrimination" as defined in School Board Policy 3330 – Supplier Diversity Outreach Program, Section D.1.; to wit: discrimination in the solicitation, selection, or commercial treatment of any subcontractor, vendor, supplier, or commercial customer on the basis of race, color, religion, ancestry or national origin, sex, age, marital status, sexual orientation or on the basis of disability or other unlawful forms of discrimination. Without limiting the foregoing, "discrimination" also includes retaliating against any person or other entity for reporting any incident of "discrimination." Without limiting any other provision of the solicitation for responses on this project, it is understood and agreed that, if this certification is false, such false certification will constitute grounds for SBBC to reject the response submitted by the Submitting Firm on this project, and terminate any contract awarded based on the response. As a condition of submitting a response to SBBC, the Submitting Firm agrees to comply with SBBC's Commercial Nondiscrimination Policy as described under its School Board Policy 3330 – Supplier Diversity Outreach Program, Section D.1.

3.11.2 **Certification Application Instructions:**

To ensure that firms seeking to participate in the Supplier Diversity Outreach Program (SDOP) qualify as Small Business Enterprises (SBEs), or are at least fifty-one (51%) percent legitimately owned, operated and controlled by minorities and/or women, each such firm shall be required to be certified as to its Small/Minority/Women Business Enterprise (S/M/WBE) ownership status by the SDOP Office <u>prior</u> to the time of the proposal submission. A letter or certificate confirming the SDOP Office's approval of S/M/WBE certification status must be submitted on behalf of each S/M/WBE firm as part of the Qualifications Statement submission.

Important Points to Remember:

- (a) Please submit any S/M/WBE certification application with all supporting documentation well in advance of the time of proposal submission, as the application review and approval process typically takes up to sixty (60) days after receipt of the completed application.
- (b) Please contact the Supplier Diversity Outreach Program Office for the application or information on the certification requirements and process at (754) 321-0505 or email at sdopcertification@browardschools.com.

DEFINITIONS

The following terms shall have the listed meanings (the definitions shall not apply outside of this section where inconsistent with those listed elsewhere in the Solicitation Documents).

RFQ FY20-192 Page 11 of 54
Program Manager - Owner's Representative Services

Affirmative Procurement Initiative (API): refers to various SDOP tools and Solicitation Incentives that are used to encourage greater prime and subcontract participation by S/M/WBE firms, including bonding assistance, evaluation preferences, subcontracting goals, and joint venture incentives. (For full descriptions of these and other SDOP tools, see, Section E of SBBC Policy No. 3330 and the SDOP Standard Operating Procedures.)

Consultant: An individual, firm, partnership, corporation or combination thereof submitting a proposal for the work as specified herein.

Subconsultant: A person or entity who has a direct contract with the prime Submitting Firm to perform a portion of the Work.

Minority and/or Women Business Enterprises (M/WBE or M/WBEs): a firm that is certified in accordance with SBBC Policy No. 3330 as either a minority business enterprise or as a women business enterprise and which is at least fifty-one percent (51%) owned, managed, and controlled by one or more minority group members and/or women, and that is ready, willing, and able to sell goods or services that are purchased by the School District.

Small Business Enterprises (SBE or SBEs): a corporation, partnership, sole proprietorship, or other legal entity for the purpose of making a profit, which is certified in accordance with SBBC Policy No. 3330 as being independently owned and operated by individuals legally residing in, or that are citizens of, the United States or its territories, and which annual revenues and number of employees shall be no greater than fifty percent (50%) of the small business size standards for its industry(ies) as established by the U.S. Small Business Administration, and meets the significant business presence requirements as defined SBBC. Policy No. 3330.

Significant Business Presence: to qualify for participation as an S/M/WBE firm in the SDOP policy, an S/M/WBE must be headquartered or have a *significant business presence* for at least one year within the relevant marketplace, defined as: an established place of business in one or more of the three counties that make up the Southern Florida relevant marketplace, from which at least fifty percent (50%) of its full-time, part-time, and contract employees are domiciled and regularly based, and from which a substantial role in the S/M/WBE's performance of a commercially useful function is conducted. A location utilized solely as a post office box, mail drop or telephone message center, or any combination thereof, with no other substantial work function, shall not be construed to constitute a significant business presence.

- Protest of RFQ Conditions/Specifications: Any person desiring to protest the conditions/specifications in this RFQ, or any Addenda subsequently released thereto, shall file a notice of intent to protest within seventy-two (72) hours after the posting, in writing to the Director of Procurement and Warehousing Services. Saturdays, Sundays, state holidays or days during which the school district administration is closed shall be excluded in the computation of the of the 72-hour time period. The formal written protest shall be filed within ten (10) calendar days after the date the notice of protest was filed. The formal written protest must be received on or before 5:00 p.m. EST of the tenth (10th) calendar day at the office of the Director of Procurement and Warehousing Services. If the tenth (10th) day in which to file a formal written protest falls on a Saturday, Sunday, state holiday or a day during which the school district administration is closed, the formal written protest must be filed the next business day. Refer to Section 120.57(3)(b), Florida Statutes, as currently enacted or as amended from time to time, for the requirements for the written protest.
 - 3.12.1 Failure to file a notice of protest or to file a formal written protest within the time prescribed by Section 120.57(3), Florida Statutes, or a failure to post the bond or other security required by SBBC Policy 3320, Part VIII, within the time allowed for filing a bond, shall constitute a waiver of proceedings.
 - 3.12.2 Notices of protests, formal written protests, and the bonds required by School Board Policy 3320, shall be filed at the office of the Director of Procurement and Warehousing Services, 7720 West Oakland Park Boulevard, Suite 323, Sunrise, Florida 33351. **Fax filing will not be acceptable for the filing of bonds.**
- 3.13 Posting of RFQ Recommendations: RFQ Recommendations will be posted on www.demandstar.com as noted on the Calendar (Section 2.0) or as modified by an addendum or by posting in the Procurement and Warehousing Services Department and at www.demandstar.com (under the document section for this RFQ). The Recommendations will remain posted for at least

seventy-two (72) hours. It is the responsibility of each Submitting Firm to ascertain any revised date for the posting of RFQ Recommendations.

- Protest of Intended Decision: Any person desiring to protest the Recommended Decision shall file a notice of protest, in writing, within seventy-two (72) hours after the posting of the RFQ Recommendation and shall file a formal written protest within ten calendar days after the date the notice of protest was filed. The formal written protest must be received on or before 5:00 p.m. EST of the tenth (10) calendar day at the office of the Director of Procurement and Warehousing Services. Saturdays, Sundays, state holidays and days during which the school district administration is closed shall be excluded in the computation of the seventy-two (72) hours. If the tenth (10th) day in which to file a formal written protest falls on a Saturday, Sunday, state holiday, or a day during which the school district administration is closed, the formal written protest must be filed the next business day.
 - 3.14.1 Failure to file a notice of protest or to file a formal written protest within the time prescribed by Section 120.57(3), Florida Statutes, or a failure to post the bond or other security required by School Board Policy 3320, Part VIII, within the time allowed for filing a bond, shall constitute a waiver of proceedings.
 - 3.14.2 Notices of protests, formal written protests, and the bonds required by School Board Policy 3320, shall be filed at the office of the Director of Procurement and Warehousing Services, 7720 West Oakland Park Boulevard, Suite 323, Sunrise, Florida 33351. **Fax filing will not be acceptable for the filing of bonds.**
- 3.15 <u>Use of Other Contracts:</u> SBBC reserves the right to utilize any other SBBC contract, any State of Florida Contract, any contract awarded by any other city or county governmental agencies, other district school boards, other community college/state university system cooperative agreements, in lieu of any offer received or award made as a result of this RFQ if it is in SBBC's best interest to do so. SBBC also reserves the right to waive irregularities or technicalities in any Qualifications Statement received if such action is in the best interest of SBBC. However, such a waiver shall in no way modify the RFQ requirements or excuse the Submitting Firm from full compliance with the RFQ specifications and other contract requirements if the Submitting Firm is awarded a contract.
- 3.16 **Assignment:** Neither party to the Agreement shall sell, assign or sublet the same without the written consent of the other; nor shall a Submitting Firm assign any monies due or to become due to the Submitting Firm, or by reason of the Contract without SBBC's previous written consent and as approved by SBBC's Attorney.
- 3.17 **Cancellation:** In the event any of the provisions of this RFQ are violated by the awardee, the Director of Procurement and Warehousing Services shall give written notice to the awardee stating the violations and unless they are corrected within five (5) business days, a recommendation will be made to SBBC for cancellation.

3.18 SBBC Photo Identification Badge

Background Screening: Awardee agrees to comply with all of the applicable requirements of Sections 1012.32, 1012.465, 1012.467 and 1012.468, Florida Statutes, and that awardee and all its personnel who (1) are to be permitted access to school grounds when students are present, (2) will have direct contact with students, or (3) have access or control of school funds will successfully complete the background screening required by the referenced statutes and meet the standards established by the statutes. Personnel shall include employees, representatives, agents or sub-contractors performing duties under the contract to SBBC and who meet any or all of the three requirements identified above. Awardee will bear the cost of acquiring the background screening required under Section 1012.32, Florida Statutes, and any fee imposed by the Florida Department of Law Enforcement to maintain the fingerprints provided with respect to awardee and its personnel. The parties agree that the failure of awardee to perform any of the duties described in this section shall constitute a material breach of this RFQ entitling SBBC to terminate immediately with no further responsibilities or duties to perform under this Agreement. Awardee agrees to indemnify and hold harmless SBBC, its officers and employees of any liability in the form of physical or mental injury, death or property damage resulting from awardee's failure to comply with the requirements of this section or Sections 1012.32, 1012.465, 1012.467 or 1012.468, Florida Statutes.

- 3.18.1 SBBC issued identification badges must be worn at all times when the awardee's personnel or agents are on SBBC property and must be visible and easily readable.
- 3.18.2 Badge Vendor Information shall be provided to the Submitting Firm at the time of Award.

Program Manager - Owner's Representative Services Version Date: 04/14/2020

RFQ FY20-192

3.19 Withdrawal or Cancellation of RFQ:

- 3.19.1 **Withdrawal:** In the best interest of SBBC, SBBC reserves the right to withdraw this RFQ at any time prior to the time and date specified for the Qualifications Statement opening.
- 3.19.2 **Cancellation:** In the best interest of SBBC, SBBC reserves the right to cancel this RFQ at any time prior to the award of a contract(s).
- 3.20 **Severability:** In case of one or more of the provisions contained in this RFQ shall be for any reason be held to be invalid, illegal, unlawful, unenforceable or void in any respect, the invalidity, illegality, unenforceability, unlawful or void nature of that provision shall not affect any other provision and this provision shall be considered as if such invalid, unlawful, unenforceable, or void provision had never been included herein.
- 3.21 It is the sole responsibility of the Submitting Firm to assure it has received the entire Qualifications Statement and any and all Addenda.
- 3.22 It is the sole responsibility of the Submitting Firm to assure that its Qualifications Statement is time stamped in the Procurement and Warehousing Services Department on or before the due time on the date due. No submissions made after the Qualifications Statement opening, amending or supplementing the Qualifications Statement, shall be considered.
- 3.23 No verbal or written information which is obtained other than by information in this document or by Addenda to this RFQ shall be binding on SBBC.

3.24 Acceptance and Rejection of Qualifications Statements:

- 3.24.1 **Acceptance:** All responsive Qualifications Statements properly completed and submitted will be evaluated in accordance with Section 5.0.
- 3.24.2 **Rejection:** SBBC reserves the right to reject any and all Qualifications Statements. In addition, a Qualifications Statement may be rejected if it does not conform to the rules or the requirements contained in this RFQ. Examples for rejection include, but are not limited to, the following:
 - 3.24.2.1 The Qualifications Statement is time-stamped at the Procurement and Warehousing Services Department after the deadline specified in the RFQ.
 - 3.24.2.2 Each Submitting Firm found legally guilty of collusion amongst Submitting Firms shall be rejected and the participants in such collusion shall be barred from future procurement opportunities until such time as they are reinstated by SBBC.
 - 3.24.2.3 The Qualifications Statement shows non-compliance with applicable laws; or contains any unauthorized additions or deletions; is a conditional Qualifications Statement; is an incomplete Qualifications Statement; or contains irregularities of any kind.
 - 3.24.2.4 The Submitting Firm adds provisions reserving the right to accept or reject an award or to enter into a contract pursuant to an award or adds provisions contrary to those in the RFQ.
 - 3.24.2.5 Failure to strictly comply with the Supplier Diversity Outreach Program, **SHALL** result in a recommendation to reject the Qualifications Statement.
- 3.24.3 SBBC also reserves the right to waive irregularities or technicalities (refer to Section 3.29) in any Qualifications Statement received if such action is in the best interest of SBBC. However, such a waiver shall in no way modify the RFQ requirements or excuse the Submitting Firm from full compliance with the RFQ specifications and other contract requirements if the Submitting Firm is awarded the contract.

RFQ FY20-192 Page 14 of 54
Program Manager - Owner's Representative Services

Version Date: 04/14/2020

- Maintenance of Records: An awardee shall keep all records and supporting documentation which concern or reflect its Services hereunder. The records and documentation shall be retained by the awardee for a period of seven (7) years from the completion date of the project associated with the Authorization to Proceed; or such period of time as required by law. SBBC or its duly authorized agents or representatives shall have the right to inspect and copy all such records and documentation as often as they deem necessary during the period of this Agreement and during the period noted above in which the records are to be retained. Such activity shall be conducted at the reasonable convenience of the awardee.
- 3.26 <u>Liability:</u> Refer to Attachment H Sample Program Manager Owner's Representative Services Agreement.
- 3.27 **SBBC Information Security Guidelines:** It is the responsibility of the Submitting Firm to read and adhere to the SBBC Information Security Guidelines when using any device connected to the SBBC's network. Following the conclusion of the contract term, all of SBBC's confidential information must be removed from the awardee's equipment and all access privileges must be revoked. Final payment will be withheld until the awardee has confirmed, in writing, that all of SBBC's confidential information has been purged from any and all electronic technology devices that were used during this contract and were connected to the SBBC's network.
- 3.28 <u>Site Visit:</u> Site visits may be requested by a Submitting Firm or its subconsultants through an RFI (refer to Section 1.4) at any time prior to the due date of the qualifications statement. Upon receipt of request, district staff will review and coordinate with location personnel to select a date and time for a site visit (if available). Once confirmed, site visit date and time will be posted on DemandStar for all interested parties. Only one pre-evaluation site visit may be scheduled for each location within the procurement. Firms will be required to submit the names of personnel that will be attending the site visit. Participants will be allowed to photograph areas related to the RFQ scope of services. Site visit participants cannot photograph any students.
- 3.29 <u>Deductions:</u> The Submitting Firm must strictly comply with the submission requirements of this RFQ in its entirety. Firms that do not strictly comply with each section's requirements will receive a one (1) point deduction for EACH technicality or irregularity. All applicable deductions shall reduce the Submitting Firm's final score.
- 3.30 **GOB Projects:** The Submitting Firm shall not be an awardee with any open and active Construction projects with the SBBC (i.e.: awarded Construction Manager (CM), Trade Contractor, etc.) to preclude any Conflict of Interest. All construction projects must be complete prior to the due date of the Qualifications Statement.
- 3.31 **CERTIFICATION OF NON-SUSPENSION OR DEBARMENT:** The Submitting Firm, by submitting its Qualifications Statement, certifies that:
 - 3.31.1 The Submitting Firm, owners, or principals are aware of the Federal Excluded Parties List System and have not been excluded from receiving Federal Contracts, certain subcontracts and certain Federal Financial and non- financial aid and benefits pursuant to the provisions of 31 U.S.C. 6101, note, E.O. 12549, E.O. 12689, 48 CFR 9.404, in addition to each Submitting Firm's codification of the Common Rule for Non-procurement suspension and debarment.as accessed and described on the federal website System for Award Management (SAM).
 - 3.31.2 State of The Submitting Firm. of the Florida owners. or principals are aware Convicted/Suspended/Discriminatory/Complaints Vendor Lists and have not been excluded from receiving State of Florida contracts pursuant to the provisions of Section 287.133(3)(d), Florida Statutes, Rule 60A-1.006, Florida Administrative Code (F.A.C), sections 287.057(1), (2) and (3), Florida Statutes, and rule 60A-1.006(1), F.A.C, and Rule 60A-1.006(1), F.A.C, as accessed and described on the Florida Department of Management Services (https://www.dms.myflorida.com/business_operations/state_purchasing/state_agency_resources/vendor_registrati on and vendor lists).
 - 3.31.3 The Submitting Firm, owners, or principals, are eligible to participate in this solicitation and are not listed on either the Federal Excluded Parties List System or the State of Florida Department of Management Services Procurement Convicted / Suspended / Discriminatory / Complaints Vendor Lists; and

3.31.4	If awarded the Contract, the Submitting Firm, owners, or principals will immediately notify the SBBC in writing if any of its principals are placed on the Federal Excluded Parties List System or the State of Florida Department of Management Services Procurement Convicted / Suspended / Discriminatory / Complaints Vendor Lists.
	(Remainder of page intentionally left blank)

SECTION 4.0 - REQUIRED QUALIFICATIONS FORMAT AND RESPONSE INFORMATION

- 4.1 SBBC's Procurement and Warehousing Services Department staff shall determine whether each Submitting Firm has addressed and provided all RFQ submittal requirements. SBBC's Procurement and Warehousing Department shall identify the status of completion for each Submitting Firm's Qualifications Statement on a spreadsheet for the Evaluation Committee's use. Requirements shall be presented in two (2) parts. Part A is the submission of the Qualifications Statement. Part B is the presentation and interview with Submitting Firms.
- 4.2 **PART A Qualifications Statement Requirements:** It is required that Qualifications Statements be organized in the manner specified below and with all the information as identified.

4.2.1 <u>Title Page</u>

- Line 1 Include RFQ number and name;
- Line 2 The RFQ Due Date:
- Line 3 The name of the Submitting Firm (company/firm name);
- Line 4 Company/firm address;
- Line 5 Company/firm telephone number.

4.2.2 **Section A – General**

NOTE: Failure to strictly comply with this section may result in a recommendation to reject the Qualifications Statement.

Section A1 Table of Contents

Include a clear identification of the material by scoring sections with section number and by page number. The Qualifications Statement page numbers should correspond to the page numbers in the Table of Contents. The Sections of the Qualifications Statement should correspond with the Sections in 4.2 of the RFQ.

Section A2 Letter of Responsibility

Include the names of the persons who will be authorized to make decisions for the Submitting Firm for this Qualifications Statement, and for the Submitting Firm efforts that may result from this RFQ. Provide titles, work addresses, telephone numbers, and e-mail addresses for each authorized decision-maker. The <u>Letter of Responsibility is to be submitted on the Submitting Firm's letterhead and signed by an Officer of the Submitting Firm's company.</u> For joint ventures, the information in this section should be provided for the principal place of business designated with the State of Florida Department of State. (One (1) page maximum)

Section A3 Signed Addenda

Include signed and dated copies of **the first page of all addenda** to verify and acknowledge receipt. All pages to each addendum do not need to be submitted.

4.2.3 <u>Section B</u> – <u>Required Forms, Licenses, Certificates, History</u>

NOTE: Failure to strictly comply with this section may result in a recommendation to reject the Qualifications Statement.

Section B1 Required Response Form

Modifications or alterations to this form shall not be accepted and will cause the Qualifications Statement to be rejected and not reviewed. The Required Response Form, shall be the only acceptable form. The Required Response Form shall be completed in ink or typewritten. The signed "original" Required Response Form shall be submitted within the Qualifications Statement Package labeled as the "original." **Failure to include the**

RFQ FY20-192
Program Manager - Owner's Representative Services
Page 17 of 54

Version Date: 04/14/2020

Required Response Form and/or failure to sign the Required Response Form with in the Qualifications Statement labeled as the "original" SHALL result in the rejection of the Submitting Firm's Qualification Statement. Refer to Attachment B.

JOINT VENTURES

The Required Response Form for Joint Venture Qualifications Statements shall follow the requirements set forth in this section. In the event multiple business entities submit a joint Qualifications Statement as a joint venture in response to the RFQ, such joint venture shall comply in all respects with the requirements of Rule 61G6-5.011, Florida F.A.C. The joint venture shall designate a single Submitting Firm to be the "Prime Firm". The Prime Firm shall prepare and submit a list of the names and addresses of all business entities of the Joint Venture Qualifications Statement. The Prime Firm shall provide all insurance requirements, execute any and all Contract Documents, complete the Required Response Form shown herein, and have overall and complete authority to resolve any dispute which may arise from work associated with this contract.

Only a single contract with one Submitting Firm shall be acceptable. Prime Firm responsibilities shall include, but not be limited to, performing of overall contract administration, preside over other business entities participating or present at SBBC meetings, oversee preparation of reports and presentations, and file any notice of protest and final protest as described herein. The Prime Firm shall also prepare and present a consolidated invoice(s) for services performed.

The SBBC shall issue only one (1) check for each consolidated invoice to the Prime Firm for services performed. The Prime Firm shall remain responsible for performing services associated with response to this RFQ. Each participant in the joint venture must sign a statement of authority (as set out in Rule 61G4-15.002, F.A.C.) giving the Prime Firm full authority to conduct the contracting business of the participant. A copy of the executed joint venture agreement and statements of authority must be submitted together with its Qualifications Statement. If the joint venture is a limited partnership, the qualifying business organization or individual must be a general partner of the joint venture.

Section B2 Licenses and Registrations (Florida)

A Submitting Firm **shall** be registered in the SBBC eProcure Online Supplier Portal (www.browardschools.com/pws) and obtain a Compliance Approval status prior to the due date of its Qualifications Statement. Requirement will be validated by Procurement staff after the receipt of the Qualifications Statement.

Additionally, all appropriate licensing numbers shall be provided on a list and include expiration dates. This list is to include the type of license or registration license number and name of the license holder. Examples include, but not limited to:

- 1. State of Florida Department of State Division of Corporations filing (Sunbiz.org) for the firm
- 2. The Submitting Firm's Fictitious Name filing (d/b/a registration) (if applicable)
- 3. All appropriate licenses on a list and include expiration dates
- 4. Firms Construction Licenses (not required) and Registrations for the Submitting Firm and Subconsultants
- 5. General Contracting License(s) (not required)
- 6. S/M/WBE Certificate (as applicable) for Submitting Firm
- 7. Continue as appropriate

Section B3 Submitting Firm History

Provide a listing of current and former business entities that the Submitting Firm is operating under and has operated under in the past. List the number of years the Submitting Firm has operated under each name. Provide the total number of years the Submitting Firm has been in business. For joint ventures, each partner/entity should provide a listing of current and former business entities, the number of years it operated under each name, and the total number of years the Submitting Firm has been in business.

RFQ FY20-192 Page 18 of 54

The Submitting Firm must have been in continuous business for a minimum of ten (10) years. If the PMOR is a joint venture, at least one of the companies within the joint venture must have been in continuous business for a minimum of ten (10) years.

The letter must be on the Submitting Firm's company letterhead and signed by an officer of the Submitting Firm's company.

Section B4 Litigation

Provide a list of any legal or regulatory action within the last three (3) years in which the Submitting Firm is a named party. For all applicable legal or regulatory actions, identify the court or agency in or before which the action was instituted, the applicable case or file number, and the current status or disposition for such reported action. If the Submitting Firm is not a party to any legal or regulatory action, provide a statement to that effect. For joint venture or team Submitting Firms, submit the information requested above for each member of the joint venture or team.

As part of its response, the Submitting Firm shall provide SBBC with a list of all instances within the immediate past three (3) years where there has been a final adjudicated determination in a legal or administrative proceeding in the State of Florida that Submitting Firm discriminated against its subcontractors, vendors, suppliers, or commercial customers, and a description of the status or resolution of that complaint, including any remedial action taken.

Section B5 Conflict of Interest

Submitting Firms must disclose the names of any employees who are employed by the PMOR that are also employees of the SBBC. The identified persons may have obligations and restrictions applicable to them under Chapter 112, Florida Statutes. Failure to submit the Conflict of Interest Form (Attachment E) with the Qualifications Statement may deem the Submitting Firm's Qualifications Statement non-responsive.

Section B6 Financial Capacity

Submitting Firms shall submit a copy of their three (3) most current audited financial statements, within the past five (5) years. If the Submitting Firm is a joint venture or partnership all firms comprising the submitting entity shall submit the three (3) most current audited financial statement, within the past five (5) years. In addition, Submitting Firms shall provide a one (1) page summary that indicates the Working Capital Ratio (Current Assets / Current Liabilities) and Net Profit Margin (Net Income/Sales*100) for each year. The Working Capital Ratios shall average a minimum of 1.5 over the three (3) most current audited financial statements, within the past five (5) years. The Net Profit Margin shall be no less than 1 standard deviation below the mean of all Submitting Firms in each of the three (3) most current audited financial statements, within the past five (5) years.

NOTES:

Section E shall be verified, evaluated, and scored by District Staff (not QSEC).

Portions of Section C2 and Section D1 below shall be evaluated and scored partially by District Staff.

All other portions of Section C and Section D below shall be evaluated and scored by QSEC.

The Qualifications Statement may NOT contain a reference to fees or construction costs.

4.2.4 Section <u>C</u> – <u>Experience and Qualifications</u> (Fifty-five (55) maximum combined total points for C1, C2, and C3)

Section C1 Executive Summary / Approach / Current Workload (Twenty-five (25) Points Maximum)

Executive Summary – (Five (5) Points Maximum) Submit a brief abstract stating the PMOR's understanding of the nature and scope of the services to be provided and capability to comply with all terms and conditions of this RFQ (Four (4) page maximum).

RFQ FY20-192 Page 19 of 54

Approach – (Fifteen (15) Points Maximum) Discuss your approach for managing the SBBC Capital Program. Provide details on how you plan to manage costs and schedules and Submitting Firm's ability to deliver quality projects of various sizes in an effective and timely manner. Include an understanding of the risks, challenges, strategy, and proposed methodology that will be employed to successfully complete the Program. Describe PMOR team's knowledge of the SBBC Capital Program and its policies, State Requirements for Educational Facilities (SREF), local codes and ordinances, and cost containment strategies. Explain Submitting Firm's approach for managing design and construction firms that fail to perform and meet their contractual obligations. Explain the approach to recruiting and retaining program management staff and assigning staff to projects. State the office location from which the Submitting Firms' Senior Management will be conducted and its distance from the Office of Capital Programs, located at 2301 NW 26th Street, Fort Lauderdale, FL 33311. If the stated office location is outside of Broward, Miami-Dade, and Palm Beach Counties, PMOR staff and assigned lead(s) will have office space for program management staff assigned at the Office of Capital Programs or other designated location. (Five (5) page maximum).

Current Workload Overall – (Five (5) Points Maximum) Demonstrate Submitting Firm's ability to perform this work given Submitting Firm's current project workload. Provide a list of all current and active Program Management Contracts and Open-end/Continuing Contracts. Provide the start and end date, client names and contact information, dollar value associated with PMOR's workload, the name of key personnel assigned to each contract including but not limited to executive management and project management (both in office and in field). Provide a list of current projects for all clients for firms that are being managed by the PMOR's office that will be responsible for this contract. Include all projects currently under contract and all projects where Submitting Firm has been selected but is not yet under contract. Include project construction value, major scope items, delivery method, percent (%) completion, and anticipated completion date. Include the name of the Project Manager, Assistant Project Manager, and Superintendent assigned to each project. **Provide the above information for Submitting Firm's subconsultant(s)/team members**. For joint ventures, the information should be provided for all partners. (Five (5) page maximum).

Section C2 Firm Experience and Qualifications (Twenty (20) Points Maximum)

Submitting Firm shall possess a minimum of five (5) years of experience providing Program Management Services for construction programs for K-12 non-charter public schools over the last ten (10) years. Submitting Firm shall either provide a dated contract award letter from the awarding entity or provide a copy of the fully executed contract agreement as evidence of experience.

Submitting Firm shall provide a statement of qualifications for the Submitting Firm and its subconsultants / team members to perform the services requested under this RFQ. Submitting Firm shall describe the Submitting Firm's capability with like size scope in the following areas:

- 1. Project Management,
- 2. Quality Assurance,
- 3. Scheduling,
- Cost Estimating,
- Project controls and reporting (including experience with web-based project management software), and
 - State approach to document control including the responsibilities of the project managers and support staff,
 - i. Explain how responsibilities are differentiated between responsibilities and support staff,
 - b. Explain experience using web-based tools to facilitate document control (like e-Builder), and
 - c. List experience with e-builder software (if any)
- 6. Other technical capability for which the firm may be able to offer. Include detailed experience with:

RFQ FY20-192 Page 20 of 54

- a. BIM Modeling and how it has been utilized within programs of similar size and scope,
- b. Contract negotiations on behalf of the owner,
 - i. Detail the number of negotiations performed and the experience of team members of applicable team members performing negotiations on behalf of owner.
- c. Scope of services and procurement document development,
 - i. Detail the examples of procurement document development preparation. Submit three (3) examples of solicitations developed.
- d. Community and public outreach and presentations,
 - i. Provide at three (3) examples of outreach documentation that demonstrates your firms' experience.
- e. Preparing and reporting on program objectives,
- f. Creating robust packages of deliverables for oversight and community groups.

Submitting Firm shall list <u>ALL</u> the projects/programs that best demonstrate each firm's experience in successfully completing projects/programs similar to the projects included in the scope of this RFQ <u>utilizing</u> the staff being assigned to this contract. Programs must be active or have been completed in the last ten (10) years from the due date of this RFQ. Projects must be active or have been completed in the last five (5) years from the due date of this RFQ. Include the following information by firm for each project/program:

- Name and address of the project. For program, list name and city, county, and state.
- Project / program owner and owner's representative name, address, telephone, and email.
- Date project / program was completed or is anticipated to be completed.
- Size of project / program (gross square feet).
- Project/program construction costs.
- Role of the firm (Prime or Subconsultant).
- Scope of work for projects or scope of services for programs for which the Submitting Firm's staff or subconsultant's staff was responsible.
- Status of project/program.
- Name and roles of Staff provided for the project/program (i.e. Management staff, Project Managers, Estimators, Superintendents, etc.) and whether that personnel will be assigned to the contract resulting from this RFQ.
- A brief narrative that demonstrates the value-added Program Management Services in delivering high quality projects within the program on time and within budget.
- Any unique issues associated with the program and the creative initiative implement by the PMOR to resolve the issues.
- For programs listed, provide a reference letter from the owner, an architect involved in the program, and a general contractor/construction manager involved with the program. If unavailable, provide reasons for the unavailability.

(Fifteen (15) page maximum).

Section C3 Relevant Projects/Programs and References (Fifteen (15) Points Maximum)

Relevant Projects/Programs (Three and One Half (3.5) Points Maximum per program / Three (3) Points Maximum per project) – Provide a list and description of a total of three (3) relevant projects and/or programs in which the Submitting Firm provided owner's representative services, program management, design management, or construction management services for similar projects and scope to this solicitation. Relevant Program Management*** submissions must have been successfully completed within the past ten (10) years and include scope similar to the scope in this RFQ. The Submitting Firm may use project management references for public K-12 school renovation projects at least five (5) million dollars and similar in scope to the projects within this program that were completed in the past five (5) years. Submitting firm shall submit at least one (1) program management reference that meets the requirements in this section for consideration. Only

RFQ FY20-192 Page 21 of 54

projects/programs in the contiguous United States of America shall be considered for evaluation and scoring.

Indicate if Submitting Firm was a subconsultant on each program management reference and include the detailed scope of services provided as a subconsultant. For individual Project references, the Submitting Firm must have been the Prime Contractor or Construction Manager. Program references where the Submitting Firm was a subconsultant and individual project specific references submitted shall only be eligible to receive a maximum of 1.0 Point in Relevancy of Project and Scope Section b.

The Relevant Program & Project / Reference Form (RPP) advertised with this RFQ shall be utilized in submitting the project/program references. For each **Relevant Project**, include:

- Name and location of Relevant Project**;
- Summary of the Relevant Project's scope, including construction delivery method**;
- Detailed scope of work that was performed by the Submitting Firm;
- Original estimated Cost of Construction**;
- Final Construction Cost**;
- Planned versus actual construction schedule in days**;
- Planned versus actual completed date (Month/Year of Certificate of Occupancy)**;
- S/M/WBE Percentage Goals vs Actual percentages**; and
- Name, title, and contact information of client/Relevant Project Owner** ("Reference").

For each Relevant Program Management Contract, include:

- Name and location of Relevant Program**;
- Summary of the scope of the Program, including construction delivery method(s) utilized, number and size range of projects, and types of project scopes of work included in the program. Provide details related to whether projects were new construction or renovations active sites;
- Detailed scope of services that was performed by Submitting Firm (include whether the Submitting Firm was the prime or a subconsultant);
- Original estimated Cost of Construction for the program**;
- Final Construction Cost for the program**;
- Planned versus actual construction schedule in days**;
- Planned versus actual completed date (Month/Year of Certificate of Occupancy)**;
- S/M/WBE Percentage Goals vs Actual percentages**; and
- Name, title, and contact information of client/Relevant Project Owner** ("Reference").

** Indicates items that are requested in the RPP Form. This information does not need to be replicated once included in the form. If an item on the RPP Form is not applicable, indicate with "N/A".

References (One and One Half (1.5) Points Maximum per project) - Provide References for the three (3) Relevant Projects utilizing the RPP Form (**Attachment D**). Relevant Program or Project References will be contacted by School District staff. School District staff shall attempt to personally contact References no more than three (3) times. If a Reference is unable to be personally contacted or otherwise fails to respond to District staff, then such Reference shall be scored as zero ("0") points. References for incomplete projects who respond will be scored as appropriate.

Relevant Projects and References for programs completed more than ten (10) years ago or projects completed more than five (5) years shall be scored as zero (0) points. If a project is incomplete, the project may receive a reference score based on the criteria stated in the reference section, but the project will not receive a relevancy score. A complete project is a project that has received a Certificate of Occupancy (CO).

RFQ FY20-192 Page 22 of 54

^{***} References must be provided for programs managed by staff in the proposed office location.

Neither the Project Relevancy/Reference Form nor the CO (if provided) count towards page limits. (Four (4) page maximum)

Respondents are encouraged to build upon the references provided with additional information to provide case studies of work performed for clients. Submitting Firms shall include:

- How document control and project management software was utilized;
- How BIM modeling was utilized;
- Examples of outreach to the public and community throughout program including meetings held, literature provided, types of presentation utilized;
- Examples of strategies used in working with difficult and/or non-performing designers and contractors. Examples should be provided for both designers and contractors and should include remedies and resolution with the applicable firms;
- Information on the frequency and types of reporting provided; and
- Detail information on the interaction with Cost Controls and Facilities Maintenance department.

4.2.5 Section D – Team Composition: (Twenty-Five (25) maximum combined total points for sections D1 and D2)

Section D1 Team Structure (Ten (10) Points Maximum)

Submitting Firm shall describe the structure of its team, including all subconsultants. It shall include the relationship of team members, past experience and the number of years working together as prime and subconsultants on a team, and the availability of team members to successfully complete the scope of this RFQ. Indicate which services the prime firm offers with in-house staff, and indicate which services the firm may opt to utilize consultants. (Three (3) pages maximum).

Section D2 Key Personnel (Fifteen (15) Points Maximum)

Provide the names of the PMOR's Management Staff, including Senior Project Managers/Team Leads, intended for use in this Program. Identify the Company Officer to be in charge of the Program. Identify any other key staff member and team members that will be assigned to this project. List the office location / main office from which each staff works. Provide a list of positions and number of each position that the PMOR feels will be required to successfully fulfil the scope of services required in this solicitation. Include a responsibility matrix that goes above and beyond an organizational chart and resumes for key personnel and all personnel identified for this solicitation. Resumes shall include, but not be limited to, name, title, other job assignments, years with current firm, and years with previous firms, project experience with each firm (include project types, size in square footage and construction costs, role on project), people/team management experience (if applicable), education, and credentials including any licenses, registrations, and certifications held by the proposed personnel.

Key Personnel* indicated in the PMOR's Qualifications Statement are expected to actually perform services in this program. The use of Key Personnel other than those listed in the Submitting Firm's Qualifications Statement without the prior written approval by the Executive Director, Capital Programs, shall be a material breach of the Program Manager – Owner's Representative Services Agreement.

4.2.6 Section E – Supplier Diversity Outreach Program (Fifteen (15) Points Maximum)

Submitting Firm shall review the Supplier Diversity Outreach Program (SDOP) Guidelines (Attachment A) attached to this RFQ. Each Submitting Firm shall submit its intended or planned subconsultant(s) for this project as applicable and in the format as required per this RFQ.

Industry-Specific Remedial Affirmative Procurement Initiatives

^{*}All staffing will need to be approved by the owner.

The Goal Setting Committee (GSC) has considered the following in the course of reaching a determination regarding which, if any, SBE or M/WBE industry-specific remedial programs shall be applied to this RFQ and resulting contract:

- a. Whether the most recent data on M/WBE utilization in the absence of race-conscious APIs indicate that construction contracts of this type have exhibited significant disparities in the utilization of M/WBE subcontractors and/or M/WBE prime contractors;
- b. Whether race-and/or gender-neutral remedies alone are likely to fully eliminate any such disparities in the utilization of M/WBE subcontractors and/or M/WBE prime contractors based upon past contract award and payment data;
- c. Whether a particular API is the least burdensome available remedy to non-M/WBE Respondents that is narrowly-tailored and that can effectively eliminate the disparities in the utilization of M/WBEs in construction contracts; and
- d. The experiences of other jurisdictions within the District's relevant marketplace for construction services.

Accordingly, the GSC has applied the terms and conditions of the following Industry-Specific Remedial Affirmative Procurement Initiatives to this RFQ and the resulting contract:

SDOP Goals

As a condition of being awarded the contract for this RFQ, the Submitting Firm agrees and commits to using its best efforts to achieve or fully satisfy all the requirements for each Affirmative Procurement Initiative that has been applied by the GSC to this contract. Such compliance is understood to be a material term of the PMOR contract, and shall include, but not be limited to, the consultant's responsibility to ensure compliance with the M/WBE Subcontracting Goals Commitment.

For this RFQ and the resultant contract, the following terms and conditions shall apply:

Minority and/or Women-Owned Business Enterprise (M/WBE) Subcontracting Goals Program

SBBC has identified a **twenty-nine percent (29%) participation goal** for Submitting Firms that subcontract with eligible SBBC-Certified M/WBE firms for at least twenty-nine percent (29%) of the total contract value. See the definition of M/WBE in Section B.2. of Policy 3330. In the absence of a waiver granted by SDOP, failure of a Bidder to commit as required in the solicitation to satisfying the M/WBE subcontracting goal shall render its Qualifications Statement non-responsive to the MWBE requirements.

The full scoring matrix for assignment of M/WBE Evaluation Preference points for PMOR M/WBE team participation is located in Section 5.2.3 of this RFQ.

INSTRUCTIONS TO SUBMITTING FIRMS

In accordance with SBBC Policy No. 3330, Section E.2.b., the GSC has established an MWBE Subcontracting Goals Program in the scoring of proposals submitted in response to this RFQ. Bidder Firms responding to this solicitation must submit a completed Statement of Intent (Document 00470) and Participation Schedule (Document 00475) listing subconsultants intended to utilize on this contract. (See Attachments A.1 and A.2)

Submission of Subcontractor Utilization Plan

Identify all M/WBE (if any) firms which will be utilized as Subcontractors in this Procurement. Sub-contractor also refers to Subconsultants and team members that are utilized during the course of this procurement. The Submitting Firm must provide SBBC M/WBE certificates for each firm to be utilized as a prime or subcontractor. Each M/WBE subcontractor and prime shall be listed in the Subcontractor Utilization Plan (as defined below) and submitted together with the proposal. The listing of M/WBEs in the Subcontractor Utilization Plan shall constitute the Submitting Firm's

RFQ FY20-192 Page 24 of 54

representation to SBBC that the certified firms are technically and financially qualified and available to perform the assigned work. Failure to provide complete and accurate information shall result in the proposal being deemed non-responsive to the M/WBE Requirements.

The Subcontractor Utilization Plan, consisting of the following documentation, must be attached to the Qualifications Statement, if the document applies:

- 1. **Attachment A.1**, S/M/WBE Subcontractor Participation Schedule (Document 00475), <u>signed by the Submitting</u> Firm.
- 2. **Attachment A.2**, Statement of Intent to Perform as an S/MWBE Subcontractor Form for each M/WBE firm (Document 00470), signed by each Subconsultant/Subcontractor that has been selected for the project.
- 3. **Attachment A.4,** S/M/WBE Good Faith Effort (Document 00480), must be submitted by the Submitting Firm if the assigned subcontracting goals is not met. (*This Document is ONLY required IF subcontracting Goals CANNOT be met*)

Should there be any discrepancy between percentages indicated in Attachments A.1 and A.2, the percentages indicated in Attachment A.2 shall be used to determine participation. If Attachment A.2 is not signed by the Subcontractor, participation attributed to the listed vendor <u>will not be included even if they are listed on A.1</u>. If no percentage is listed, the Submitting Firm will not receive points. If a percentage range instead of fixed percentage is provided for a particular Subcontractor on the Statement of Intent to Perform as a S/M/WBE Subcontractor Form, the highest percentage in the range will be used to calculate participation, thereby effectively representing the Submitting Firm's commitment to the contract goal.

Good Faith Effort

If the Submitting Firm, after a good faith effort, cannot meet the MWBE participation requirements from eligible SBBC-certified Minority/Women Business Enterprise (M/WBE) subcontractors, the Submitting Firm shall submit a fully completed and executed M/WBE Good Faith Effort Form (Attachment A.4) with the RFQ. See the Document 00467, Supplier Diversity Outreach Program (SDOP) Guidelines (Attachment A) for more information.

Please review Section E.2.d. of Policy 3330 at http://www.browardschools.com/SDOP, or call the Supplier Diversity Outreach Program Office at 754-321-0505.

4.3 **PART B – Presentation Requirements:** The Presentation shall utilize Microsoft PowerPoint. The presentation may elaborate on information contained in the Qualifications Statement, but it shall not contain information that is not presented in its Qualifications Statement. No other hand-outs, display boards, models or any other props are to be utilized or even brought into the presentation. The presentation shall not contain a reference to fees or construction costs.

4.3.1 Presentation Submission Requirements

- Submitting Firm shall provide twelve (12) printed hard-copies the day of the presentation
- Submitting Firm shall provide a flash drive to the Purchasing Agent with the final Microsoft PowerPoint presentation on the day of the presentation.
- <u>Submitting Firm shall provide a near final draft of Microsoft PowerPoint presentation to the Purchasing Agent via</u> email by **2:00 PM, two (2) days** prior to the presentation for review.
- 4.3.2 **Section AA Scoring of Approach** (Twenty-nine (29) Points Maximum)

Approach to Project Management (Six (6) Points Maximum) Submitting Firm shall discuss qualifications and approach for delivering the projects under this RFQ, documenting the services to be provided, and your firm's M/WBE commitment. The PMOR shall describe how it will effectively manage and execute the projects.

RFQ FY20-192 Page 25 of 54

Approach to Cost Controls (Eight (8) Points Maximum) Submitting Firm shall describe recommendations that may directly impact the costs and schedule.

<u>Ability to Meet Schedule</u> (Eight (8) Points Maximum) Submitting Firm shall <u>present measures that will be taken to insure the firm will maintain established construction schedules.</u>

<u>Availability of Resources</u> (Seven (7) Points Maximum) Submitting Firm shall detail resources available to perform services related to this RFQ, explain experience with BIM Modeling and e-Builder at how it will be utilized to further the program and project management objectives. The Submitting Firm shall discuss program-specific constraints including how the constraints will be navigated.

4.3.3 <u>Section BB</u> – <u>Scoring of Key Personnel</u> (Eleven (11) Points Maximum)

Submitting Firm shall discuss key personnel, including both the PMOR's staff and the subconsultants' staff. Submitting Firm shall explain key roles and the experience of the personnel filling those roles.

(Remainder of page intentionally left blank)

RFQ FY20-192 Page 26 of 54
Program Manager - Owner's Representative Services

Version Date: 04/14/2020

SECTION 5.0 - EVALUATION OF QUALIFICATIONS

5.1 **Evaluation of Firms – This procurement will be performed in a two-step process.** Part A will be the submittal of a Statement of Qualifications which will be evaluated by the District and QSEC. Following evaluation of Part A, Submitting Firms shall make a presentation to QSEC based on the firm's approach, experience, and team composition. Scores from Part A and Part B of the QSEC meeting will be combined to determine the ranking of Submitting Firms.

PART A - Evaluation of Qualifications - Qualified Qualifications Statements shall be evaluated by the following Categories.

SECTION	POSSIBLE POINTS	EVALUATOR
Section A – General Section B – Required Forms	Pass/Fail Pass/Fail	District Staff District Staff
Section C – Experience and Qualifications		
Section C1Executive Summary	0 to 5	QSEC
 Approach Current Work Load Overall 	0 to 15 0 to 5	QSEC QSEC
Section C2 – Submitting Firm Experience and Qualifications Section C2 – Submitting Firm Experience and Qualifications	0 to 18	QSEC
(SBBC Prime Program Management experience Section C3 – Relevant Projects / References	e) 0 or 2 0 to 15	District Staff QSEC
Section D – Team Composition		
Section D1 – Team Structure Section D1 – Team Structure (Team experience working toge	0 to 10 ther) 0 or 2	QSEC District Staff
Section D2 – Key Personnel	0 to 13	QSEC
Section E – Supplier Diversity & Outreach Program Section E – M/WBE Subcontracting Goal	0 to 15	District Staff
Section E - MANDE Subcontracting Goal	0 10 13	District Start

TOTAL POSSIBLE POINTS PART A - 100

<u>PART B - Evaluation of Interviews</u> – Submitting Firms will make a presentation to QSEC, which shall be evaluated by the following Categories.

SECTION	POSSIBLE POINTS	EVALUATOR			
Section AA – Approach to the project and demonstrated ability to meet the schedule					
 Approach to project management 	0 to 6	QSEC			
 Approach to cost controls 	0 to 8	QSEC			
Ability to meet schedule	0 to 8	QSEC			
 Availability of resources 	0 to 7	QSEC			
Section BB – Key Personnel	0 to 11	QSEC			

TOTAL POSSIBLE POINTS PART B - 40

TOTAL POSSIBLE COMBINED POINTS - 140

- 5.2 <u>Initial Screening</u>: School District staff will review factors for each Qualification Statement on the Qualification Statements Evaluation Checklist Form.
 - Responsiveness: Submitting Firms' Qualifications Statements shall be evaluated by District staff to determine whether each Submitting Firm is responsive and responsible. Submitting Firms determined by District staff to be non-responsive or non-responsible shall be notified of such determination in advance of the applicable QSEC meeting. Staff shall present its recommendation to find applicable Submitting Firm(s) non-responsive or non-responsible to QSEC for consideration at the QSEC meeting. Staff shall present its recommendation of waivers and technicalities found in Submitting Firms' Qualifications Statements (when applicable) to QSEC for consideration at the beginning of QSEC meeting. Should the recommendation of waivers and technicalities for a Submitting Firm(s) not be accepted by QSEC, the applicable Submitting Firm(s) will be deemed non-responsive and discussions with the Submitting Firm(s) will not take place. QSEC will recommend to SBBC, that SBBC reject the applicable Submitting Firm(s) Qualifications Statement.
 - 5.2.2 <u>Initial Scoring:</u> Section E Supplier Diversity Outreach Program will be scored by District staff, not by QSEC, and such scores shall be provided to QSEC on the day of the meeting after QSEC members have completed their individual scoring.

Section C2 – Firm Experience and Qualifications, shall be <u>partially</u> scored by District staff. Submitting Firms with prior Program Manager experience with SBBC as a Prime (not as a subconsultant) shall receive two (2) bonus points which shall be added to the scores provided by each QSEC member.

Section D1 – Team Structure, shall be <u>partially</u> scored by District staff. Submitting Firms with prior experience working together with its identified team for at least two (2) consecutive years shall receive two (2) bonus points which shall be added to the scores provided by each QSEC member.

5.2.3 Scoring of M/WBE Participation – Section E

The SBBC shall award a maximum of fifteen (15) points for M/WBE Participation as listed in the Evaluation Point Tables below. At the time the Qualifications Statement is submitted, the Submitting Firm shall identify all SBBC-certified M/WBE firms (if any) which will be utilized by completing Attachment A.1, S/M/WBE Subcontractor Participation Schedule (Document 00475) and Attachment A.2, Statement of Intent to Perform as an S/M/WBE Subcontractor (Document 00470).

M/WBE Participation Commitment Percentage of Total Contract Value	Assigned # of M/WBE Preference Points
0% – 29%*	0
30% – 34%	5
35% – 39%	9
40% – 44%	12
45% – 49%	15

^{*29%} is the minimum stated goal. No points are awarded for committing to the minimum goal.

NOTE: Points for Category E shall be provided by the Economic Development and Diversity (EDDC) Staff for use by QSEC members.

5.3 QSEC Evaluation Scoring and Qualification Selection Evaluation Committee RFQ Scoring Rubric

The *Qualification Selection Evaluation Committee RFQ Scoring Rubric ("the Rubric")* provides guidance to the QSEC members during evaluations of Program Manager – Owner's Representative (PMOR) Services Qualifications Statements. In each category shown below, recommendations are provided to include a breakdown for Excellent, Good, Fair, or Poor. QSEC members will be asked to associate specific points to the response of each specific Submitting Firm. The breakdown below should be used to assist QSEC members in assigning points to the quality of each submittal and each presentation.

- Part A Evaluation Scoring: QSEC members shall review the RFQ responses in order to rank the Submitting Firms. QSEC members shall be provided with the *Rubric* below to provide guidance during the evaluation of qualification statements. Each QSEC member shall complete a Final Evaluation Form Part A and score independently. Each QSEC Members shall ensure that no two (2) Submitting Firms' scores shall be the same on their individual Final Evaluation Form. The highest and lowest scores given by a QSEC Member for each Submitting Firm shall be omitted and the remaining scores shall be averaged for each Submitting Firm. Part A Evaluation Scoring consists of one hundred (100) points maximum, of which eighty-one (81) points maximum are scored by QSEC members.
 - 5.3.1.1 **Section C Experience and Qualifications** (Sixty-three (63) Points Maximum, Sixty-one (61) Points Maximum scored by QSEC)

Section C1 Executive Summary (Five (5) Points Maximum):

Excellent – 5 points Good – 3 to 4 points Fair – 2 points Poor – 1 point Not Included – 0 points

Executive Summary is within page count limitations (see section C1 of the RFQ). Single Site Procurements

Firm shall demonstrate:

- A strong understanding the scope of services for this RFQ.
- The capability to deliver scope outlined in this RFQ.
- The ability to comply with terms and conditions.

Section C1 Approach (Fifteen (15) Points Maximum):

Excellent – 14 to 15 points Good – 10 to 13 points Fair – 5 to 9 points Poor – 1 to 4 points Not Included – 0 points

Approach is within page count limitations (see section C1 of the RFQ).

- Firm should demonstrate:

 The willingness and ability to:
 - manage project budgets and schedules.
 - deliver quality projects of various sizes.
 - manage multiple contractors and resolve issues with contractors or designers.
- The ability to recruit and retain program management staff and assign staff to roles within the program.
 - Firm should discuss the number of schools assigned to project management and senior staff.
- Knowledge of the size of the scope covered by this RFQ.
 - Firm should state that similar approaches would be used for different sites or present variances in the approach based on the site.

RFQ FY20-192 Page 29 of 54
Program Manager - Owner's Representative Services

- Firm should state risks, challenges, strategies (including cost containment), and proposed methodology that will be employed in the Program.
- Firm should clearly show how it obtains independent cost estimates. Firms should explain its experience with independent Program Controls teams.
- Firm should discuss its knowledge of SBBC Policy, SREF, local codes and ordinances.
- Its approach to construction project management, permitting/inspections, construction schedule management. It should detail examples of successful approaches.
- The specific risks to the program and how their approach would mitigate those risks.
- The office location from which this Senior Management overseeing this work will be located and the distance from the Office of Capital Programs (as required by section C1 of the RFQ).

Section C1 Current Workload Overall (Five (5) Points Maximum):

Excellent – 5 points Good – 3 to 4 points Fair – 2 points Poor – 1 point Not Included – 0 points

Current Workload Overall is within page count limitations (see section C1 of the RFQ).

- Firm should provide list of current, active projects and projects that Submitting Firm has been selected for, but is not yet under contract, including any Program Management and Open-end/Continuing Contracts. Information should include:
 - Contract Start and anticipated end dates
 - Client names
 - Dollar value associated with Firm's workload
 - Key personnel assigned to contract (Executive Management, Project Manager, Assistant Project Managers, and Superintendent)
 - Percent completion for project or program
- Firm should demonstrate workload capacity through the firm's:
 - number of staff
 - current assignments of staff
 - stage of project completion
 - qualifications and roles of staff members presented for this project
- The above information should be provided for the Submitting Firm and any subconsultants presented.

Section C2 Firm Experience and Qualifications (Twenty (20) Points Maximum, QSEC Scores Eighteen (18) Points Maximum) (Two (2) points are scored by District Staff) (See Section 5.2.2 Section C2)

Excellent – 16 to 18 points Good – 10 to 15 points Fair – 5 to 9 points Poor – 1 to 4 points Not Included – 0 points

Firm Experience and Qualifications should be within page count limitations (see section C2 of the RFQ).

- Firm shall demonstrate experience with:
 - Delivery methods utilized in this program (i.e. CMAR, Design Bid Build, Continuing and Open-end Contracts)
 - Design reviews for Southeast Florida building codes.
 - o Extensive Experience is preferred in Broward County Public Schools, Broward, Miami-Dade, and Palm Beach counties.
 - **Primary Experience**
 - SBBC projects

Version Date: 04/14/2020

Page 30 of 54

- Has the Submitting Firm and its subconsultants had experience working as a Program Manager for SBBC together?
- Secondary: Has the Submitting Firm and its subconsultants had experience working on projects with the District in a variety of capacities (CM at Risk or Bids) and how many projects have they worked on simultaneously?
- o Other Florida Public K-12 Projects (includes charter schools)
 - Has the Submitting Firm and its subconsultants had experience working as a Program Manager for Miami-Dade and/or Palm Beach Schools?
 - For Other Florida K-12 programs?
 - Did the Submitting Firm and its subconsultants have experience working on multiple projects simultaneously with a singular program, and/or
- Other Florida Education Projects (i.e. colleges, universities, private K-12 schools);
- Secondary Experience
 - o Other similar sized and scoped educational programs,
 - o Government programs with similar scopes to this RFQ, and/or
 - Other programs with similar sizes and scopes.

QSEC member should review the number of years of firm's experience within applicable categories; review how recent the experience is with the applicable categories, with the most weight given to Recent Experience (last 5 years) within primary experience categories with SBBC Programs (1st), Miami Dade/Palm Beach County Programs (2nd), Other Programs (3rd), SBBC Projects (4th) and Other Educational Projects in Florida (5th). Other projects with similar size and scopes are secondary.

QSEC Members should focus on proven success, not just experience. Many PMORs have undertaken work at large organizations but failed to deliver on-time and/or on-budget results. QSEC members should review experience with team members as it relates to 1. Design support in south Florida, 2. Cost estimating, and 3. E-builder. Firms with stronger experience should receive more weight.

Section C3 Relevant Program and/or Projects / References (Fifteen (15) Points Maximum):

The 2nd page of the Program/Project Relevancy / Reference Form will be utilized by committee members to compile individual project scores.

Reference scores (Four and a Half (4.5) Points Maximum, One and a Half (1.5) Points Maximum per reference):

References will be verified by staff and provided to the committee for review and final scoring. The committee will review references based on the accuracy of the information submitted, whether the reference indicated it would do business with the firm again, and the overall performance based on timely completion of the project/program; cost effectiveness (was project within budget); quality of the communications/coordination with the program/project team members; responsiveness to the owner; and quality of the project.

Relevancy scores (Ten and a Half (10.5) Points Maximum, Three and a Half 3.5 Points Maximum per program/project):

- a. Program References get 0.5 Points. Project References get 0.0 Points.
- b. Relevancy of scope of the program or project is similar in size and scope to BCPS SMART Program. (1.75 Maximum Points)
- Program or Project is for a public educational facility in the State of Florida (1.0 Point Maximum)
- d. Program or Project is for an educational facility in the United State of America (0.25 Points Maximum)

5.3.1.2 **Section D – Team Composition**

Section D1 Team Structure (Ten (10) Points Maximum, QSEC scores Eight (8) Points Maximum) (Two (2) points are scored by District Staff (See Section 5.2.2 Section D1):

Excellent – 8 points Good – 5 to 7 points Fair – 3 to 4 points Poor – 1 to 2 points Not Included – 0 points

The Team Structure should be within page count limitations (see Section D1 of the RFQ).

Firm shall provide:

- Its team structure is identified, and it included all subconsultants
- Description of team members
- Past experience and number of years working together
- Availability of team members to complete the scope of this RFQ
- Services offered by in-house staff
- Services which Submitting Firm will opt to utilize consultant

Section D2 Key Personnel (Fifteen (15) Points Maximum):

Excellent – 14 to 15 points Good – 10 to 13 points Fair – 5 to 9 points Poor – 1 to 4 points Not Included – 0 points

Firm shall (see Section D2 of the RFQ):

- List staff to be included in this project, including, but not limited to, the following specific roles:
 - Officer in charge of SBBC Projects
 - Kev Personnel, including, but not limited to:
 - Management Staff
 - o Senior Project Managers / Team Leads
 - Project Managers (if identified)
 - Assistant Project Managers (if identified)
 - Project Superintendent (if identified)
- Provide:
 - A responsibility matrix that goes beyond organizational chart for the Program.
 - Resumes for all identified key staff including:
 - Name and current title,
 - Other iob assignments.
 - Years with current firm,
 - Years with previous firms,
 - Project experience with each firm (include project types, size in square footage and construction costs, role on project),
 - People/team management experience (if applicable),
 - Education, and
 - Credentials including any licenses, registrations, and certifications
 - Information above for subconsultants' personnel.

Committee member should review firm's staff members and subconsultants staff intended to be used on this project with the following:

- SBBC program management and project management experience
- Miami-Dade and Palm Beach County Public Schools program management and project management experience

RFQ FY20-192
Program Manager - Owner's Representative Services

Version Date: 04/14/2020

- Other Florida Public K-12 program management and project management experience
- Other Florida Educational program management and project management experience
- Other similar sized and scoped educational programs
- Government program management with similar scopes to this RFQ
- Other projects with similar sizes and scopes
- Number of years working on programs/types of projects indicated
- Number and types of projects indicated.

QSEC Members should focus on proven success, not just experience. Many PMORs have undertaken work at large organizations but failed to deliver on-time and/or on-budget results. QSEC members should review experience with team members as it relates to 1. Design support in south Florida, 2. Cost estimating, and 3. E-builder. Firms with stronger experience should receive more weight.

5.3.1.3 Section E – Supplier Diversity and Outreach Program (Fifteen (15) Points Maximum):

- Scored by district staff
- 5.3.2 Part B Evaluation Scoring: Submitting Firms shall give a 20-minute presentation to QSEC members which will be followed with a Question and Answer period. Each QSEC member shall complete a Part B Evaluation Form and score independently. Each QSEC member shall rank the Submitting Firms using the scoring rubric provided. Each QSEC Members shall ensure that no two (2) Submitting Firms' scores shall be the same on their individual Final Evaluation Form. The highest and lowest of the committee's scores for each Submitting Firm shall be dropped, and an average shall be taken of the remaining scores per Submitting Firm. Part B Evaluation of Presentations consists of forty (40) points maximum.
 - 5.3.2.1 **Section AA Scoring of Approach** (Twenty-nine (29) Points Maximum)

Approach to Project Management (Six (6) Points Maximum) (See section 4.3.2 of the RFQ):

Excellent – 6 points Good – 4 to 5 points Fair – 3 points Poor – 1 to 2 points Not Included – 0 points

Firm shall:

- Demonstrate the ability to manage the scope of this RFQ.
- Demonstrate the ability to deliver quality projects that address any site-specific constraints including the approach to work around the constraints.
- Demonstrate minimum qualifications for project staff.
- Discuss use of software to document project.

QSEC Members should focus on proven capabilities. Weight should be given to detailed examples.

Approach to Cost Controls (Eight (8) Points Maximum) (See section 4.3.2 of the RFQ):

Excellent – 7 to 8 points Good – 5 to 6 points Fair – 3 to 4 points Poor – 1 to 2 points Not Included – 0 points

Firm should:

- Demonstrate the ability to manage projects to reduce change orders and stay within the projects' budgets.
- Discuss use of Management Information Systems like e-Builder to accomplish this.

Ability to Meet Schedule (Eight (8) Points Maximum) (See section 4.3.2 of the RFQ):

Excellent – 7 to 8 points Good – 5 to 6 points Fair – 3 to 4 points Poor – 1 to 2 points Not Included – 0 points

Firm should:

- Clearly demonstrate the ability to deliver a program and the project or projects within the program within an established construction schedule.
- State measures that will be utilized to maintain established construction schedule.
- Provide examples of experience with similar schedules on similar projects / program.

Availability of Resources (Seven (7) Points Maximum) (See section 4.3.2 of the RFQ):

Excellent – 7 points Good – 5 to 6 points Fair – 3 to 4 points Poor – 1 to 2 points Not Included – 0 points

Firm should:

- Explain experience with BIM Modeling Demonstrate the ability to deliver quality projects
- Discuss any program-specific constraints including the approach to work around the constraints in detail.

5.3.2.2 **Section BB – Key Personnel** (Eleven (11) maximum points) (See Section 4.3.3 of the RFQ):

Excellent – 10 to 11 points Good – 7 to 9 points Fair – 4 to 6 points Poor – 1 to 3 points Not Included – 0 points

Firm shall:

- List staff to be included in this project, including, but not limited to, the following specific roles:
 - Officer in charge of SBBC Projects
 - Key Personnel, including, but not limited to:
 - Management Staff
 - Senior Project Managers / Team Leads
 - Project Managers (if identified)
 - Assistant Project Managers (if identified)
 - Project Superintendents (if identified)
- Have all members listed as key personnel on its team (Prime and Subconsultants) physically present for the Part B Evaluation.
- Detail why identified personnel are the most qualified for the Program Management of Scope in this RFQ.
- Discuss its responsibility matrix and how that is utilized to open communication and deliver a successful program.
- Demonstrate the experience and ability of staff on the team to meet deliver a successful program.

Committee member should review firm's staff members and subconsultants staff intended to be used on this project with the following:

SBBC program management and project management experience

- Miami-Dade and Palm Beach County Public Schools program management and project management experience
- Other Florida Public K-12 program management and project management experience
- Other Florida Educational program management and project management experience
- Other similar sized and scoped educational programs
- Government program management with similar scopes to this RFQ
- Other projects with similar sizes and scopes
- Number of years working on programs/types of projects indicated
- Number and types of projects indicated.

QSEC Members should focus on proven success, not just experience. Many PMORs have undertaken work at large organizations but failed to deliver on-time and/or on-budget results. QSEC members should review experience with team members as it relates to 1. Design support in south Florida, 2. Cost estimating, and 3. E-builder. Firms with stronger experience should receive more weight.

5.3.3 <u>Combined Part A and Part B Evaluation Scoring</u>: At the conclusion of Part B Evaluation Scoring, QSEC's final averaged scores shall be combined from both Parts along with District scores from Part A.

5.4 **Submitting Firm Ranking and Tiebreakers**

5.4.1 **Rankings**

After the completion of Part B, responsive and responsible Submitting Firms shall be ranked based on the total of: a) the points received from the initial screening completed by conducted by District staff, and b) the QSEC's averaged final evaluation Part A and Part B. The Submitting Firms receiving the most combined points shall be considered to be the **most qualified** Submitting Firm ("Firm 1"). All remaining Submitting Firms shall be ranked in descending order based on total combined points received (Firm 2, Firm 3, Firm 4, and etc.).

5.4.2 **Tie Breakers**

- 5.4.2.1 The first tie-breaker between identically scored Submitting Firms shall be the total number of times the applicable Submitting Firms are ranked as the **most qualified ("Firm 1")** by individual QSEC committee members.
- 5.4.2.2 The second tie-breaker between identically scored Submitting Firms shall be the total number of times the applicable Submitting Firms are ranked as the **second most qualified** ("Firm 2") by individual QSEC committee members.
- 5.4.2.3 The third tie-breaker between identically scored Submitting Firms shall be the total number of times the applicable Submitting Firms are ranked as the **third most qualified** (**"Firm 3"**) by individual QSEC committee members.
- 5.4.2.4 The fourth tie-breaker shall be a coin flip, to be called in the air, by the <u>tied</u> Firm that held its discussions with QSEC first.

5.5 Recommendations to Negotiate and Award

After scoring has been completed, QSEC shall:

- 5.5.1 Recommend to the Superintendent that either one (1) or two (2) of the highest ranked firms be selected to provide Program Manager Owner's Representative Services.
- 5.5.2 After considering QSEC's recommendations, the Superintendent shall determine whether to pursue agreements with either one (1) or two (2) of the highest ranked firms.

RFQ FY20-192 Page 35 of 54
Program Manager - Owner's Representative Services

5.5.2.1 One (1) Firm - If the Superintendent elects to pursue an agreement with only one (1) firm, he shall direct District staff to negotiate terms for Program Manager – Owner's Representative Services with Firm 1 at compensation which staff determines is fair, competitive, and reasonable.

Should staff be unable to negotiate satisfactory terms with Firm 1, negotiations with Firm 1 shall be formally terminated**. Staff shall then undertake negotiations with Firm 2. Should staff be unable to negotiate a satisfactory contract with Firm 2, staff shall formally terminate** negotiations with Firm 2, and then undertake negotiations with Firm 3.

This process shall continue with the remaining, most qualified firms until a satisfactory contract is negotiated. If no negotiations are successful, the procurement shall automatically terminate.

5.5.2.2 <u>Two (2) Firms</u> - If the Superintendent elects to pursue an agreement with two (2) firms, he shall direct District staff to negotiate terms for two (2) separate agreements for Program Manager – Owner's Representative Services with Firm 1 and Firm 2 at compensation which staff determines is fair, competitive, and reasonable.

Should staff be unable to negotiate satisfactory terms with Firm 1, Firm 2 or both firms, negotiations with the applicable firm(s) shall be formally terminated**. Staff shall then undertake negotiations with Firm 3, and should staff be unable to negotiate a satisfactory contract with Firm 3, staff shall formally terminate** negotiations with Firm 3, and then undertake negotiations with Firm 4.

This process shall continue until satisfactory contracts are negotiated with two (2) firms. If negotiations are unsuccessful with two (2) firms, the Superintendent may elect to continue with a recommendation to award to just one (1) firm or recommend a termination of the procurement. If no negotiations are successful, the procurement shall automatically terminate.

**NOTE: If negotiations with a firm are formally terminated, that firm shall be removed from further consideration for award of a Program Manager – Owner's Representative Services under this RFQ.

- After successful negotiations with the applicable Submitting Firm(s), the applicable Agreement(s) shall be submitted to the Board for approval and award.
- 5.7 **Award:** The Agreement(s) resulting from these negotiations shall be governed by the laws of the State of Florida, and shall have venue established in the Circuit Court of the 17th Judicial Court in and for Broward County, Florida, or the United States District Court of the Southern District of Florida.

END OF EVALUATION OF QUALIFICATIONS

6.0 SCOPE OF SERVICES

Section 6.0 begins with an overview of the purpose and objectives of the Broward County Public Schools SMART Program and overall Capital Program, as well as the scope and current status of the District's Capital Program. This overview is followed by three categories of District expectations of its Program Manager – Owner's Representatives (PMORs). These expectations center around the District's objectives in retaining owner's representatives, the PMOR's responsibility for various Capital Program relationships, and most importantly, the District's expectations related to the PMOR's approach to program management and its project management methodology. Respondents must specifically address each of the following sections in their Qualifications Statement of services and are encouraged to expand upon the sections below and/or provide additional or innovative value-added services based upon their own unique expertise and knowledge. The PMOR is expected to utilize the e-Builder program/project management system for all project-related activities. As such, respondents should include experience using web-based tools to facilitate document control. Respondents should also provide evidence of successful experience in delivering programs on time and within budget.

6.1 Purpose and Objectives of the Broward County Public Schools Capital Improvement Program

6.1.1 General Purpose and Objectives of Broward County Public Schools Office of Capital Programs (OCP) and Physical Plant Operations (PPO)

Broward County Public Schools (District) has as its mission a commitment to educating all students to reach their highest potential. The strategic goals of the District are High-Quality Instruction, Safe & Supportive Environment, and Effective Communication. The Office of Capital Programs (OCP) indirectly contributes to the first goal through its effective delivery of much-needed capital improvements at schools. The OCP is committed to providing safe schools that create a supportive learning environment to ensure that it maximizes the limited resources of the District to meet the needs of every school. In delivering services and much needed capital improvements to schools, the OCP seeks to engage each school community in a manner that meets the third goal. Some of the District's values, which directly relate to services provided by the OCP include positive stakeholder involvement, holding everyone to the highest ethical standards to achieve excellence, high-quality customer service, and valuing the diversity of our community.

The Office of Capital Programs (OCP), along with Physical Plant Operations (PPO), must ensure that facilities meet the operational needs of 241 schools serving nearly 270,000 pre-kindergarten to twelfth grade students and approximately 175,000 adult students. In meeting the needs of schools, OCP and PPO must manage the maintenance, repairs, and improvements of the portfolio of facilities in an effective and efficient manner as trustees of public assets.

The Office of Capital Program's (OCP's) primary objective is the continued management of the 5-Year Capital Plan, while the Physical Plant Operations Department's (PPO's) primary objective is to move from a demand maintenance cycle to life cycle management of assets as part of its current and long-term strategic plan.

6.1.2 Purpose and Objectives of the Broward County Public Schools SMART Program

In order to clearly identify and prioritize the needs of schools, in 2014 the District completed a district-wide comprehensive Facilities Needs Assessment to obtain accurate physical plant structural and building system deficiencies listings, as well as classroom infrastructure and relational supporting educational space deficiencies, as measured against established District standards. The objective of the needs assessment process was to provide an objective, data-driven overview of the District as a whole, as well as a strategic planning tool for prioritizing the most critical needs facing our schools for safety and security; repairs and renovations; and technology and technology infrastructure.

These needs were organized into the SMART (Safety, Music & Arts, Athletics, Renovation, and Technology) program, and an \$800 Million General Obligation Bond was passed by voters on November 4, 2014 to fund the program. A corresponding 5-Year Capital Plan was developed. The 5-year Capital Plan from February 24, 2015 included both the \$800 million dollars in bond-funded projects, as well as another \$184 million in new and previously funded projects, for a total Capital Program of \$984 million.

The District decided to engage the assistance of a team of construction industry professionals as the most flexible, cost-effective means of forging its projects into a coherent program. To accomplish this objective, the District engaged

RFQ FY20-192
Program Manager - Owner's Representative Services
Page 37 of 54

one Program Management firm to act as its Owner's Representative (PMOR), and another firm to provide Cost and Program Controls (CPCM) services.

6.1.3 Current Status of the Broward County Public Schools Capital Improvement Program

The Capital Program ("Program" or "Capital Program") shown in the 2019-20 Adopted District Educational Facilities Plan (ADEFP) consists of SMART program projects funded by the 2014 General Obligation Bond, additional DEFP projects that were added to the SMART program, and previously-funded projects not included in the SMART program.

Adjustments have been made to SMART program project budgets as construction contracts have been awarded. The current Program budget is planned at over \$1.3 Billion. The program has faced several challenges since inception. An increased awareness of the importance of understanding and adhering to district design standards and local building code, especially those related to roofing, is expected to have a positive effect on roofing project timelines moving forward. If the planned process improvements continue, and if market conditions remain stable or improve, the Program is to be executed by the end of the 2025-26 school year.

The Program is comprised primarily of a variety of renovation projects with about 65% of scope (\$879 million) focused on building exterior envelope. Roof recovery and replacement projects represent the majority of building envelope renovations. The Program also consists of \$347 million in HVAC (25%), and \$95 million in life safety systems (10%) projects. Less than eight percent, or \$101 million, of the district's capital program consists of new construction projects.

The design work for approximately 71% of projects (\$894 million) has been completed. Most of the projects left in the design phase are year 4 and 5 projects, although more complex projects from earlier program years represent 24% (\$115 million) of the design work to be completed. Construction has been completed for about 7% (\$44 million) of the projects in the Program. Additional information related to the scope of the individual projects may be found in the Bond Oversight Committee (BOC) quarterly reports (click here) and 2019-20 ADEFP (click here).

A comprehensive listing of all active projects by life cycle status is included in this RFQ as Attachment I. Additional projects may be identified during the term of the contract and any extensions. As the District's goal is retain two Owner's Representatives to manage its Capital Program, each PMOR will be assigned a portion of the program to manage, while a third firm will be engaged under a separate solicitation to provide Cost and Program Controls Management for the overall program.

The remainder of Section 6.0 is organized around the following categories of District expectations: 1) objectives for retaining Owner's Representatives; 2) PMOR's responsibility for various Capital Program relationships; and 3) PMOR's approach to program management and project methodology. District expectations of the PMOR are followed by criteria upon which the program and PMOR performance will be evaluated, and a summary of PMOR deliverables.

6.1.4 Broward County Public School's Objectives in Retaining Owner's Representatives to Manage its Capital Program

In order to be effective in meeting the facility and operational needs of the Schools in a prioritized, transparent, and resource conscious manner, the District must continue to manage the capital projects of its Schools as a coherent program. There is a need for coordination and synergy at each point in the process of planning, designing, and building capital projects for this District. Moreover, there is a need to achieve this coordination through means that are both flexible and cost-effective. The District has determined that engaging the assistance of a team of construction industry professionals organized in two Program Manager - Owner's Representative (PMOR) agents is the most flexible, cost-effective means to achieve the coordination and synergy necessary to meet the demands of the Schools.

A PMOR offers flexibility through its capacity to engage professionals with specific expertise at appropriate points in the planning and implementation of projects. This is also cost-effective in that staffing is scalable according to the volume of work in the Program; thereby limiting the District's financial liability during periods when volume does not require an extensive staff. Scalable staffing is of interest to the District as the capital needs, and available funding, varies from year to year.

RFQ FY20-192 Page 38 of 54
Program Manager - Owner's Representative Services

The PMORs will also enhance the District's efficiency by keeping the District current with the latest developments in management systems and practices. For flexible staffing to be effective, there must be a core set of systems in place that enables both the owner's representatives and the District's professionals to contribute their knowledge and work product to, and resource information from, a central, coordinated repository of data.

As such, the District has selected e-Builder as the required project management software to serve as the central repository of data for the Capital Program. The PMOR shall utilize e-Builder program/project management system for all project-related activities. By implementing, maintaining, and upgrading e-Builder, the District's management information systems appropriate for the District's projects, the owner's representatives will facilitate the efficient and effective use of information throughout the District's capital projects. Through benchmarking the best practices of project management on a local and national scale, the PMOR will assist the District's senior management in establishing metrics that will identify areas where the Capital Improvement Program is functioning efficiently, and areas that require improvement.

The role of the PMOR will be to manage all assigned capital projects from their current status (which may include projects at 100% of the design phase) through commissioning and closeout. The PMOR, in its management of the District's capital projects acts as the District's representative and actively seeks to identify issues and resolve issues. Where issues cannot be resolved by the PMOR, the PMOR's responsibility as the manager of projects is to provide options, and/or recovery plans, to the Office of Capital Programs ("OCP") for review and direction.

While the Program has an established sequencing, the PMOR will be responsible for continually reviewing and making recommendations to optimize the sequencing of the assigned portion of the program considering changing external and internal conditions and needs of the District.

As such, the PMOR should demonstrate capabilities in a wide range of renovations in facility portfolios of a similar nature to that of the District's. The PMOR should also demonstrate capabilities in managing renovation work within operating facilities. Demonstrated capabilities to provide high quality customer service to facility stakeholders will be critical to the success of the PMOR and the overall Program. Furthermore, the PMOR will be responsible for effectively communicating with the numerous stakeholders of school capital renovation projects.

The District expects the members of the PMOR to apply their professional expertise, experience, judgment, and information management systems in a manner that will facilitate the District's efforts to develop, maintain, and enhance the relationships necessary to meet the District's purpose and objectives.

6.2 PMOR Responsibility for Developing, Maintaining, and Enhancing Capital Program Relationships

The PMOR Lead (Program Manager/Director) is wholly responsible for the relationship between the PMOR and the District. As such, the District considers the PMOR Lead to be a crucial member of its Capital Program team. The PMOR Lead plays a particularly important role in mobilizing the PMOR in its initial application of the program approach and project methodology, and in managing the implementation of the PMOR's strategic plan. The District expects the PMOR Lead to identify and, as needed, work with the management of the District's Office of Capital Programs to initiate critical relationships within the District and between the District, the PMORs and schools. Moreover, the District expects the PMOR Lead to assist and enhance the efforts of the management of the Office of Capital Programs to instill the sense of collective identity and collegiality among all members of the Program necessary for the successful operation of the Program. As such, the District expects the PMOR to take primary responsibility for coordinating the relationships between the team members responsible for planning, design, construction, controls, and administration in a manner that is consistent with best practices in the pertinent industry, and which facilitates prompt, complete and smooth hand-offs from one function to the next, thereby minimizing gaps and/or delays in the progress of the Program.

The relationships for which the PMOR Lead will be responsible for developing, maintaining, and enhancing include, but are not limited to, the following areas: 1) the PMOR staff assigned to the District; 2) District Senior Management; 3) Schools; 4) Project Design Professionals and Contractors; 5) the Cost and Program Controls Manager (CPCM) and other agents for the District; 6) the District's departments related to Planning through Closeout; Procurement; Economic Development and Diversity Compliance; Physical Plant Operations (PPO); and the District's Building Department and other Regulatory agencies. A description of the relationship between the PMOR and each area follows.

RFQ FY20-192 Page 39 of 54

6.2.1 Relationships within the PMOR

The District expects the PMOR Lead to direct the day-to-day operations of the PMOR and to ensure that each individual within the PMOR: 1) understands his or her role and responsibilities; 2) performs the tasks required by that role and those responsibilities in a manner consistent with the purpose and objectives of the District; and 3) administers the operation of the PMOR in an engaged, open, and professional manner. If it comes to the attention of OCP management that any member of the PMOR team is not performing per the three prior expectations, the PMOR Lead will be notified of the need for immediate corrective action, and/or the need for a change in the PMOR team staffing. Similarly, the PMOR Lead will notify OCP management should the Lead become aware of a PMOR team member performance problem as defined above. Proposed changes to the staffing of the PMOR team during the term of the contract and any extensions are to be presented to the Executive Director, Capital Programs and appropriate OCP management for review and approval prior to staff changes or new assignments. This review process includes proposed changes to the PM Lead and/or Principal-in-Charge. The Superintendent and School Board must also approve proposed changes to the PM Lead and/or Principal-in-Charge.

6.2.2 Relationships with District Senior Management

In a support role to the District's Senior Leadership Team, the PMOR staff will have a direct, working relationship with several senior members of the District, but will report on a day-to-day basis to the Office of Capital Programs (OCP) leadership and staff. The District's Senior Leadership Team will be relying upon the PMOR staff to: 1) implement project management processes as determined by OCP that coordinate and integrate the District's projects into a program; 2) work to improve and coordinate the District's relationships with regulatory agencies, municipal agencies, Schools and the public where applicable; 3) anticipate and identify issues; 4) provide prompt and comprehensive notice of issues as they arise; 5) work with OCP staff to identify those issues which require the input of District management, and those issues which are resolvable by the PM; 6) furnish accurate and timely information regarding the progress of projects and emerging issues; 7) offer options for responding to issues and needs as they arise, and do so in a manner that assesses the risks and opportunities afforded by those options; 8) implement decisions made by District management; and 9) report promptly and thoroughly on the consequences of implementation.

The District expects the PMOR staff to perform services related to its relationship with the District's Senior Leadership Team in a manner that brings the entire weight of the PM's professional experience, expertise and judgment to bear. Moreover, the District expects the PMOR to perform these functions in a manner that supports and facilitates the District's management operations as a collaborative team, and in particular as a team in which each of the members understands its role, and respects and appreciates the roles and contributions of other team members.

The PMOR shall provide a scalable staffing structure which is responsive to the variability and seasonality expected in a capital program. This structure, while providing the appropriate and necessary amount of responsiveness to the District's Capital Program, shall do so in a manner which consistently meets the expectations of the relationship just described. Although there is a need for adaptation of staffing, the PMOR shall have a consistent Lead for the term of the contract. The requirement of a consistent PMOR Lead is discussed in more detail under section 6.3.4.

6.2.3 Relationships with Schools

The OCP oversees the planning, design, and construction services for all facilities within the District's portfolio. The District expects the PMOR to be essential to its efforts to deliver timely, cost-effective and productive services to schools. The nature of the PMOR's relationship with schools, instructional departments, and other stakeholder groups may vary during the project's lifecycle. However, with clear and accurate project details provided to school administrators and other school-related personnel at the onset of a project, the PMOR will be able to successfully manage the requirements and expectations of the project.

The District expects the PMOR to work closely with curriculum and instructional staff during the initiating phase of a project to elicit the details of the school's project and programmatic requirements. During this stage, the PMOR must

RFQ FY20-192 Page 40 of 54
Program Manager - Owner's Representative Services

demonstrate the ability to balance stakeholders' varying interests and ensure that the project team communicates and interacts with stakeholders in a professional and cooperative manner to obtain positive outcomes.

The PMOR shall also demonstrate an acknowledgement that the relationship with schools is based on ensuring school administrators have a clear understanding of the roles and responsibilities of team members from project inception to closeout and full turnover to PPO. Although the relationship will change over the course of the project and interactions and interface will vary depending on the phase of the project, it is important that clear lines of communication remain intact, and school administrators and team members respect the limits of each other's responsibilities and authority. Hence, it is important that the PMOR establish a sign-off process during each phase of the project to ensure expectations are aligned as the project evolves.

The PMOR must also establish a process that clearly defines milestones and activities during every phase of a project, and particularly during the construction phase where maintenance of operations is key to school administrators. The PMOR must be prepared to work with school staff in outlining logistical solutions to keeping school functions in progress without any disruption to the teaching or student learning experience, thus preserving the integrity of the educational process. It is essential therefore that the PMOR establish a key goal of maintaining a zero-negative impact to student instruction during all phases of the project's lifecycle.

6.2.4 Relationships with Project Design Professionals and Contractors

The PMOR will have direct day-to-day responsibility for the oversight and management of the District's design professionals and contractors awarded projects. In establishing this Program, the District must be a client of choice for design professionals and contractors in the South Florida construction market. The District expects the PMOR to facilitate achieving this goal through: 1) the implementation of clear and thorough procedures for each phase of project delivery; 2) consistent and fair implementation of the procedures; 3) supporting District staff efforts in developing, managing and maintaining qualified pools of vendors; 4) timely review of invoicing for all vendors; 5) developing working relationships with the design and construction providers in the South Florida market that are predicated upon respect and trust, that support the free flow of ideas and innovation, and that consistently demonstrate to design professionals and contractors that their ideas and concerns will not only be heard but addressed in a mutually beneficial manner; 6) facilitating problem-solving and decision-making among design and construction entities in an impartial and goal-driven manner; and 7) fostering a collaborative environment between design professionals, contractors, and key stakeholder departments, such as the Building Department, Office of the Chief Fire Official, and PPO to address issues and concerns early in the project development process, and throughout the multiple phases of a project.

The PMOR is also expected to leverage its relationship with the Building Department to improve the permitting and inspection processes. This requirement is particularly crucial to the District's efforts to apply lessons learned to the roofing sub-permit process. The District expects the PMOR to work with the Building Department to create a mutually beneficial process that results in fewer revisions to permit packages, and fewer re-inspections. The PMOR's ability to create, maintain, and enhance the relationship between itself, the Building Department, and Design Professionals and Contractors is critical to the PMOR's ability to deliver the Program on time and on budget.

6.2.5 Relationships with Cost and Program Controls Manager (CPCM) and Other District Consultants

Each PMOR is expected to actively support the program controls and procedures as set forth by District staff, or by its assigned agent. While each PMOR will be assigned discrete and independent projects that comprise the Program, the District expects each PMOR to focus on delivering their assignments with fidelity to the District's plan of a single cohesive program delivering critically needs improvements to all schools in Broward County. As such, a PMOR is expected to develop working relationships with District agents that are predicated upon respect and trust, that support the free flow of ideas and innovation, and that consistently address issues in a manner that ensures the District's best interest are primary to all actions.

The District will be relying upon the PMOR to: 1) implement Program and project management processes that aid in coordinating Owner's Representative and Cost and Program Controls function; 2) anticipate and identify issues related to the different functions 3) work with OCP staff, the Cost and Program Controls Manager (CPCM), or other assigned agents thereof, to identify those issues which require the input of District management, and those issues which are

RFQ FY20-192 Page 41 of 54

Program Manager - Owner's Representative Services

resolvable by the PM; 4) provide prompt and comprehensive notice of issues as they arise to the appropriate District Staff or its assigned agent; 5) furnish accurate and timely information regarding the progress of projects to the Cost and Program Controls manager; 6) offer options for responding to issues and needs as they arise, and do so in a manner that assesses the risks and opportunities afforded by those options; 7) implement decisions made by District management; 8) report promptly and thoroughly on the consequences of implementation; and 9) incorporate lessons learned from the work of the Program, regardless of which agent of the District has gleaned such lesson so that the continued planning, design, and construction work of the District is clearly viewed as an effective continuous improvement process.

The PMOR, while working closely with the CPCM, will be responsible for certain project-based estimating tasks, while the CPCM will be responsible for more of the Program-based estimating functions. The PMOR staff will handle estimating tasks, including but not limited to, contractor's Schedule of Value review and approval, Earned Value analysis for contractor's monthly pay applications, evaluation and verification of unit prices for bidding, and during construction, monitoring, reviewing, and analyzing change orders under \$25,000. The PMOR will also be responsible for the review and analysis of Construction Services Minor Projects (CSMP) bids, and consultant fee proposals. The CPCM will be responsible for the Program level estimates and forecasts, project budget estimates, and change orders exceeding \$25,000.

6.2.6 Relationships with Planning through Closeout

OCP staff will be relying upon the PMOR staff to: 1) implement program and project management processes that coordinate and integrate the District's projects into a program; 2) anticipate and identify issues; 3) work with OCP staff to identify those issues which require the input of District management, and those issues which are resolvable by the PM; 4) provide prompt and comprehensive notice of issues as they arise; 5) furnish accurate and timely information regarding the progress of planning and design activities, along with any emerging issues; 6) offer options for responding to issues and needs as they arise, and do so in a manner that assesses the risks and opportunities afforded by those options; 7) implement decisions made by District management; 8) report promptly and thoroughly on the results of implementation; 9) provide constructability reviews during the design phase of each project; 10) ensure the timely and effective closeout and commissioning of all projects; 11) communicate and incorporate lessons learned throughout the Program; and 12) engage in professional activities within the planning, design, construction, and construction management communities such that the PMOR will be able to keep the District informed of current trends, developments and best practices within those communities.

6.2.6.1 Relationships with Planning and Design

The PMOR staff reports directly to the PMOR Lead, with principal day-to-day responsibility for the planning and design functions assigned to the PM. OCP staff will facilitate planning efforts, and the PMOR preconstruction staff will both serve as a resource in support and monitor conceptual design to ensure compliance with Program standards and continuity in project design. The District expects the PMOR staff to distinguish between project-driven and programmatic changes to a design, to track each set of changes appropriately, and to ensure that the distinction between project and programmatic changes is maintained throughout the District's planning and design processes (i.e. ensure that programmatic changes are appropriately addressed within each project, while project-specific changes do not become program-wide practices). The PMOR staff is expected to evaluate the available procurement methods on a per project basis and make a formal recommendation to OCP, which provides support that demonstrates the best balance of cost, timeliness, and quality. The District currently utilizes five methods of procurement: 1) Design-Bid-Build; 2) Construction Manager at Risk; 3) Continuing Services; 4) Design-Build; and 5) A hybrid method including any of the previous methods along with in-house construction staff.

The PMOR staff will support OCP in the selection of design professionals of record, where applicable to the selected procurement approach, and work closely with OCP staff to manage each project's transition from planning to design. Once a design professional for a project is selected from the qualified pool of designers, the PMOR staff will assume the lead in managing the project, and OCP staff will serve as a resource to the design team for each project. When a design-build project delivery method is approved to procure the project,

RFQ FY20-192 Page 42 of 54
Program Manager - Owner's Representative Services

the PMOR shall develop the necessary design criteria package as part of its management of design development.

6.2.6.2 Relationships with Construction and Closeout

The PMOR, through its Project Management staff, will be responsible for the planning, scheduling, and managing of the construction phase from award through project closeout. OCP staff will provide guidance to the PMOR Project Managers, however, the PMOR Project Managers will be solely responsible for managing the Contractor's day to day construction activities, ensuring compliance with the District's Specifications, Standards and Policies, coordinating the construction activities with on-site staff as well as District staff, reviewing, approving and processing of all required construction documentation, and overseeing the project punchlist and closeout activities. The PMOR staff is expected to work closely with the Contractor and Project Consultant to ensure the contractor maintains the project budget and schedule, minimizes disruptions to the school, and provides the Owner with sound, quality construction through industry standard QA/QC programs. At Substantial Completion, the PMOR staff shall work closely with the Project Consultant to create the project's Substantial Completion punchlist, and work with the Contractor to ensure that all items on the punchlist are properly resolved. At Final Completion, the PMOR staff will review all project closeout documents and warranties, schedule all required training, and submit all required closeout documentation.

6.2.7 Relationships with Procurement

The PMOR is responsible for developing Support Documents for Requests for Qualifications, Requests for and Proposals, and Bid Support Documents according to regulatory and District requirements. Once the solicitation package is complete, the PMOR turns it the package over to procurement. The PMOR is responsible for participating in Pre-Bid & Pre-submittal Conferences, answering RFIs, facilitating communication with the design teams, and assisting procurement in any and all ways, as directed, to have successful solicitations. The PMOR shall communicate with and engage procurement regarding any contractor, subcontractor, etc. related project issues. The PMOR may be required to provide staff augmentation for the procurement process, which shall be approved by the Director of Procurement & Warehousing Services. While procurement has primary responsibility for the management and oversight of personnel involved in the procurement process, the PMOR is expected to include any staff augmentation in the R.A.C.I. chart prepared for the OCP.

6.2.8 Relationships with the Economic Development and Diversity Compliance Department (MWBE Compliance)

The PMOR is responsible for supporting EDDC with the compliance process and the execution of Policy 3330. The PMOR may be required to provide staff augmentation to provide assistance with Supplier Diversity Outreach Program compliance, contractor outreach, and other assistance in support of Policy 3330. While EDDC has primary responsibility for the management and oversight of personnel involved in the facilitation of Policy 3330 in support of the SMART Program, the PMOR is expected to include any staff augmentation in the R.A.C.I. chart prepared for the OCP.

6.2.9 Relationships with Physical Plant Operations (PPO)

An important responsibility of the PMOR is to ensure that the District's PPO staff, both school and District-based, are fully trained in the use and maintenance of all building systems and equipment. Once Substantial Completion has been achieved, the PMOR will be responsible for coordinating and managing all warranty items during the 12-month warranty period, including but not limited to, scheduling all owner required training sessions, enforcement of all specification requirements for the operation, maintenance training, testing, and start-up of equipment. The PMOR shall be responsible for distinguishing between routine maintenance of the building structure and its components as opposed to defects requiring corrective action by the General Contractor.

All training shall be documented, and video-recorded for future reference and use in training new employees. The PMOR will also be responsible for transitioning all warranty responsibilities to the District's maintenance department after the first year.

RFQ FY20-192 Page 43 of 54
Program Manager - Owner's Representative Services

The PMOR shall also be responsible for ensuring that the District's established commissioning process that commences at the initial phases of a project and continues through the end of the warranty period is managed effectively, and that the proper functional testing of equipment and systems occurs to verify that they are working and performing, as specified. This process shall be documented by the PMOR and any problems that arise be resolved as quickly as possible.

The final hand-off to PPO occurs at the completion of 12-month warranty period and surveillance of the project and its components, at which time PPO will be responsible for the operations and maintenance, including administration of all material and equipment warranties.

6.2.10 Relationships with the District's Building Department and other Regulatory Agencies

Each PMOR shall fully comprehend the role of internal and external regulatory bodies and Authorities Having Jurisdiction. Each PMOR must demonstrate an understanding of the objectives of the District's Building Department in providing professional plan review and inspection services for minor and major projects within the District; the responsibilities of reviewing and approving construction documents for compliance with various building codes and conducting inspections to insure that work is being done in accordance with approved plans, and such a manner that the safety, health and welfare of individuals utilizing the constructed facility are preserved and protected.

It is crucial to the success of the individual projects and the overall program that the PMOR recognizes the importance of compliance and effectively managing the processes that govern regulatory requirements, while balancing the goal of executing a project. By establishing a framework that addresses the competing priorities of project management and compliance to standards, codes, and regulations, the PMOR must exhibit a clear appreciation of the effort required to achieve both sets of goals. Incorporating this understanding into the initial phases of a project through the Project Charter process, the challenges of compliance can be properly documented and dealt with in mitigating risks. This effort must be carried through the ensuing phases of a project, such as in bidding and award, and construction phases. Hence, the PMOR must indicate knowledge of the permitting process during design, as well as the inspection process during construction, with particular attention to the specific requirements for approval of shop drawings and sub-permits for roofing.

It is essential that the PMOR recognizes and documents the necessity of compliance with Authorities Having Jurisdiction as a key element of project execution and factors the impact and level of priority during its preparation of the Risk Register. Through early acknowledgement of such requirements and their effect on project and program progress, the PMOR will be able to showcase its efforts in ensuring adherence to internal and external codes and regulations, and how that directly correlates to the success of the program.

6.3 District Expectations of the PMOR's Approach to Program Management

6.3.1 Overview

The District expects the Program Manager – Owner's Representative (PMOR) to deliver projects that maximize taxpayer dollars in a timely and effective manner, while fully complying with District and other regulatory agency requirements. In addition to successfully implementing the relationships critical to the success of the District's Capital Program, the strategic approach and methodology employed by the PMOR to successfully address the inherent risks, challenges, and day-to-day functions associated with launching and managing a program of this size and complexity are key to achieving this goal.

Broward County Public Schools expects the PMOR to use its professional expertise and knowledge to engage in a comprehensive approach to the management of the District's Capital Program. While the District expects the PMOR to perform all the tasks necessary to meet its standard of care and to do so in a manner that is consistent with that standard of care and the objectives and expectations of the District, the PMOR is expected to take a system approach to contribute to the optimization of the overall Program, not simply particular tasks or components thereof—in short, the District expects true Program Management from each PMOR firm. Program level activities the District expects the PMOR to provide included but are not limited to day-to-day management and administration of the program and PMOR staff and sub consultants, staff augmentation, holistic management of the planning, design, pre-construction, and

RFQ FY20-192 Page 44 of 54

construction, procurement management and support, contract management, contract negotiations, program-wide reporting, interaction with the School Board and District senior leadership, the Bond Oversight Committee and other Stakeholder groups providing capital program oversight, community outreach, warranty management and asset preservation.

The District has decided to keep the Cost and Program Control function separate from its Program Manager-Owner's Representative function. While this separation of duties creates a system of checks and balances, it also creates the need for functional relationships, collaboration, and communication to enable both types of Program Managers to successfully fulfill their responsibilities. Because the Cost and Program Controls Manager will be responsible overall program costs and controls, the PMOR will also be responsible for ensuring the execution of improvements and changes made at the Program level across individual projects. Respondents must indicate their understanding of the working relationship between the PMOR and CPCM teams and include a plan for how program-level Cost and Program Controls improvements and changes will be executed.

As part of its project management methodology, and in keeping with a strategy for risk management, the PMOR is expected to utilize the Enterprise Project Management software, known as e-Builder, for the full term of the Agreement, as well as any optional extensions, as may be granted by mutual consent. The e-Builder software is the exclusive Program/Project Management system to be used throughout the entire lifecycle of each project to track all key elements and activities.

The PMOR's success in implementing its strategic approach to program management and its project methodology will be evaluated quarterly by OCP management throughout the term of the contract and any extensions.

6.3.2 Mobilization Plan

Although the PMOR will be assuming responsibility for a set of projects that have been previously managed as part of a program, to ensure as seamless a transition as possible, the District expects the PMOR's mobilization efforts to immediately focus on both evaluating projects and developing an approach for its share of the program. Respondents shall propose a detailed mobilization plan. The mobilization plan must clearly demonstrate that the respondents recognize that time is of the essence, and provide a timeline with key activities at 30, 60, and 90 days.

6.3.3 Program Management Plan

At the program level, the District expects the PMOR launch team to analyze the overall program and begin developing the Management Plan that will form the foundation for the assigned portion of the Program. The District expects the PMOR 's strategic approach to include an emphasis on the program and project level activities most critical to successfully managing the risks, challenges, and daily activities most relevant to the remaining phases of the Program. As such, the District is particularly interested in how the PMOR will mitigate risk related to current local construction market conditions, construction schedules and budgets, and permitting and inspections. Expected activities include, but are not limited to the following: 1) Meeting with OFC management and the CPCM to review and validate the program master schedule and budget; 2) Reviewing previously established standard operating procedures (SOPs); 3) Meeting with OCP management and the Building Department to review key issues and lessons learned related to District design specifications, permitting, and inspection, as well as MWBE goals and pre-qualification of sub-consultants; and 4) evaluating other risk factors.

At the project level, the District expects the PMOR to identify key issues that might impact the ability of the PMOR to move projects forward. Expected activities include but are not limited to: 1) Validating project scope, budget, and schedules; 2) Developing initial project management plans.

6.3.4 Leadership

The District expects the PMOR to provide a set of comprehensive program management services and deliverables that encompass both program and project level activities, and which are performed consistently and effectively throughout the term of the contract and any extensions.

RFQ FY20-192 Page 45 of 54

The District believes leadership is key to the PMOR's ability to fulfill this requirement. As such, the PMOR shall have consistent leadership (PMOR lead/Program Manager/Director) for the term of the contract and any extensions. Proposed changes to the PMOR Lead and/or Principal-in-Charge shall be brought before the School Board for approval.

The PMOR lead will be responsible for day-to-day program management and communication as well as program-level activities such as interaction with and reporting to the School Board, District Senior Leadership, the Bond Oversight Committee, community stakeholder groups, and key district personnel; staffing, contract management, and program level reporting.

6.3.5 Staffing

In addition to the scalable staff required to manage the Program, the PMOR may be required to provide scalable staff augmentation for the Procurement and Warehousing Services Department, the Economic Development and Diversity Compliance Department, and The Office of Capital Programs.

The PMOR is required to maintain, and provide to management, on a quarterly basis, a Responsible, Accountable, Consult, Inform Matrix (R.A.C.I.) that includes both its staff and the staff of its subconsultants.

6.3.6 District Design and Material Standards and Design Criteria

The PMOR is required to actively participate and contribute to the continuous improvement and updating of the District's Design and Material Standards and Design Criteria. The PMOR's contribution shall be for the benefit of the owner and take into consideration, code requirements and restrictions, life-cycle costs and value, and be aligned with industry standards and best practices. The PMOR's input shall be in collaboration with the District's Standards Committee which is comprised of representatives from OCP, Building Department, PPO, and other District stakeholders, as necessary. In developing its mobilization plan, the PMOR shall meet with the Building Department to gain a working knowledge of the inherent differences between the District's Master Permit and Sub Permit requirements and processes related to roofing.

6.3.7 Procurement Management

The PMOR will serve as a key team member in ensuring the Program is continuously and consistently engaging the contractor and design professional community within South Florida, to maximize participation and competition. The PMOR's management of the projects shall also include a timely, consistent, and transparent vendor evaluation procedure that seeks to foster continuous improvement rather than simply assess penalties. Results of vendor evaluations will be used not only to inform PMOR performance, but to assist in the vendor selection process.

The PMOR shall also provide to OCP a recommended procurement strategy from the five (5) delivery options available to the District for each project within the context of the project and overall program objectives. The District's five delivery options are the following: 1) Design-Bid-Build; 2) Construction Manager at Risk; 3) Continuing Services; 4) Design-Build; and 5) a hybrid method including any of the previous methods along with in-house construction staff.

The District expects the PMOR to explicitly define a process by which these recommendations are to be made that shall minimally include a risk assessment of the procurement/ project options along with an optimization of cost, time, and quality. A procurement delivery method matrix shall be developed and maintained at the project level and the process and matrix summarized at the Program level.

The PMOR will work with the procurement department to avoid unnecessary project delays or bid protests by ensuring regulatory and District requirements are being followed regardless of delivery method. The PMOR will also work with the procurement department to ensure solicitation packages are complete and comprehensive in terms of 1) proper project scope; and 2) performance expectations and compliance requirements as applicable relative to safety, cost, schedule, quality, site logistics, close out documentation, submittals, engineering, manufacturing installation quality, coordination, delivery, start-up commissioning, and schedule objectives for each.

RFQ FY20-192 Page 46 of 54
Program Manager - Owner's Representative Services

As the Owner's Representative, the PMOR will serve as an integral part of the District's negotiation team during the negotiation phase of the solicitation process. For each solicitation, the PMOR shall assign a lead negotiator, as a member of the negotiating team. The lead negotiator must be familiar with market conditions, hourly rates, labor burden rates, M/WBE requirements, and other factors that are critical in determining a fair fee and arriving at a mutual agreement with the selected firm or company.

6.3.8 Reporting

The PMOR lead will be expected to work with the OCP and CPCM to prepare a series of program status reports on a monthly and/or quarterly basis. As part of this requirement, the PMOR shall, on a monthly basis, provide an executive summary of the performance of the program, while collaborating with the Cost and Program Controls team to ensure no duplication of efforts. Specific reporting requirements associated with the executive summary are detailed in 6.5.1 of this RFQ.

Stakeholders for whom the PMOR will be required to prepare and present reports include the School Board, Superintendent, District senior leadership, and community stakeholder groups. A significant amount of reporting is required to be prepared quarterly for the District's Bond Oversight Committee. The PMOR lead will be expected to coordinate the timely delivery of transparent, accurate, and user-friendly reports that highlight key areas of progress or concern. The latest Bond Oversight Committee Report (BOC) may be found here (Click here).

6.3.9 Community Outreach and Public Involvement

As part of the PMOR's approach to communication, the District expects the PMOR to provide services that help facilitate the District's communication goals for the Program related to community outreach and public involvement. These services may include identifying and assessing stakeholders; planning, implementing, and facilitating public meetings; designing and producing communication materials; and developing and implementing educational programs. The District will be relying upon the PMOR to deliver these services using a variety of informational tools including, but not limited to, electronic/social media, print media, virtual and physical meetings, videos, and websites—customized to different target audiences.

6.3.10 Operational Integration

All projects executed as part of the Capital Program shall do so with a full understanding of the complete life cycle of systems impacted by each project. The PMOR shall, therefore, ensure that ongoing operations and maintenance, along with life cycle costs, are integrated during the definition stage of each project and delivered as part of each project's closeout. The PM's professional expertise, combined with a robust set of master specifications and guidelines, shall be complemented by engaging and collaborating with PPO to ensure a seamless transition from the closeout of the project to ongoing operations.

6.3.11 Knowledge Management

In addition to the PMOR bringing current industry best practices to bear in all its services, the PMOR is to participate in the District's system to collect and codify processes and practices utilized in the implementation of the Capital Program. The system deployed by the District, or its agent, shall ensure that the experience gained in executing the program becomes common knowledge of the organization through a clear process of capturing, retaining, and sharing information, regardless of changes in staff, process or technology. The PMOR should anticipate that the knowledge management system deployed will go beyond a latent system of collecting information and be robust enough in its workflows for lessons learned to be imposed on the ongoing program as part of a continuous improvement model.

6.4 District Expectations of PMOR's Project Management Methodology

6.4.1 Overview

The PMOR is expected to integrate into the project methodology a process of gatekeeping, between each phase, to assess, report and mitigate project and program risk, by defining and executing a comprehensive approach to risk

RFQ FY20-192 Page 47 of 54

management that includes a qualitative risk analysis through a Risk Register, and quantitative analyses to track program performance. These analyses shall be conducted in conjunction with the Cost and Program Controls Manager. With the potential for constraints, both financial and workforce, the District expects the PMOR to establish a formal risk management approach that identifies negative risk occurrences impacting scope, schedule, budget, and quality of the deliverables through development of a Risk Management Plan. The Risk Management Plan must therefore include, but not be limited to, defining the probability and impact of risks, and the methodology of managing same, including, responsible parties, at both project and program levels.

The PMOR must demonstrate an understanding of the importance of defining and structuring key activities in a manner that will reduce risk at each phase of a project, and implementing a plan based on program priorities, sequencing, and other constraints. The plan should incorporate industry best practices, such as effective design, quality, cost, time, and construction management, in order to deliver best-in-class projects to the owner, and should also address a comprehensive insurance and bonding compliance program will be required for opportunities to assign and track risk, to third parties, where feasible. In addition, the PMOR must be prepared to conduct regular Risk Register updates and Risk Workshops involving all stakeholders to provide insight into changes in qualitative and quantitative risks. The following subsections detail the District's expectations of the PMOR related to key project management activities.

6.4.2 Project Charter

As part of the definition and design phases of each project, the PMOR will be required to engage both OCP and the School, where applicable, to develop a clear set of objectives that address all stakeholder requirements within the resource and time constraints. The charter will serve as a baseline upon which to measure the overall effectiveness of the PM's service, the quality benchmark for the project, and establish clear expectations for all stakeholders.

Prior to the commencement of a project, the PMOR is expected to engage in the development of a Project Charter that clearly outlines the high-level information that is necessary for a well-defined project start and project boundaries, and clearly defines key elements of the project, including, but not limited to, purpose, measurable project objectives, project description, risks, milestone schedule, budget, stakeholders, and approval requirements. While this process is typically conducted at the project level, the data gathered should be analyzed in consultation with the District's Cost and Program Controls Manager to formulate a high-level assessment of the program.

It is therefore required that the PMOR implement a high-quality project charter process that accompanies every project and is sequenced at the optimum milestone in the project lifecycle, and that the process allows for appropriate stakeholder validation. The Project Charter shall also be used as a reference tool during the progress of a project to keep stakeholder expectations aligned from beginning to end.

6.4.3 Design Management

The PMOR shall provide management of the design process from project definition through completion of preconstruction. The PMOR team shall be fully capable of developing design requirements for projects, as assigned. Design requirements shall provide a clear scope of work upon which design professionals can propose and develop projects. The PMOR shall ensure that the design professional or other design entity produces a design that considers obtaining the best value for the established construction budget, while meeting the standards, criteria, and Educational Specifications established by the District, and complying with the Florida Building Code, State Requirements for Educational Facilities, and all other applicable codes and regulations for design and construction within the south Florida area. The PMOR shall ensure those responsible for design reviews are familiar with the same.

As part of effective design management, the PMOR must develop and implement a structured process by which the project management team diligently manages the progress of the design phase, from scope validation through obtaining a Letter of recommendation for Permit from the District's Building Department. The entire process, and required deliverables are well defined within each consultant's Professional Services Agreement, with which the PMOR and individual PM must be knowledgeable.

It is imperative that the PMOR recognizes that for the success of a project and the overall program that the design phase, including the required submittals by the design professional, reviews by the Building Department and other regulatory agencies, and constructability reviews by OCP/others be carefully monitored and actively managed to be

RFQ FY20-192 Page 48 of 54

able to progress projects in accordance with approved schedules and projected budgets. The PM must also ensure that the submittals are of a high quality and include all required documents, such as complete drawings, project manuals and specifications, in accordance with the phase of the submittal.

In addition, it is essential that the PM hold regular project meetings with the design professionals and construction manager, if applicable, to ensure that issues are addressed and any changes to design documents be implemented promptly. The meetings shall be conducted in a collaborative manner to identify and address project concerns or conflicts that may arise from code and constructability reviews, and to further mitigate the occurrence of foreseeable changes during bidding, and/or construction.

Although the District has made a commitment to executing the approved scope of work for all projects rather than deleting or reducing scope, the PM shall establish a clear understanding of the cost drivers and other risks of the project prior to the development of construction documents. The PMOR shall implement the District's robust process to control changes during the design to avoid unnecessary increases in project budgets. These constraints, as noted above in Section 6.4.3.1 will help mitigate the occurrence of foreseeable changes during construction.

While the District does not currently require design documents to be developed using Building Information Modeling (BIM), the PMOR is expected to make recommendations to incorporate BIM to further enhance the design process. The PMOR shall develop and execute on a plan to expand and improve the use of BIM which can be used to resolve design issues and potential conflicts before they reach the field. By incorporating BIM tools into the design process, the project team can quickly resolve issues and conflicts between design and constructability without having to participate in on-site meetings. Therefore, with the advantages BIM can offer, the project and overall program can benefit through enhanced efficiency and reduced risk.

6.4.4 Project Controls

The District is relying upon the PMOR to provide accurate, current information regarding the status and progress of assigned projects that comply with the procedures and control systems of the District. The importance of control systems with respect to design and construction is well understood. Their significance for program and project planning is often less clear. It is of primary concern to the District, therefore, that the PMOR works closely with OCP, or its agent, to ensure that planning processes are appropriately informed by the District's experience, in addition to the industry expertise of the PM. It is equally important to the District that the PMOR understands that the budgets and schedules produced in the planning process become the parameters of the control systems for the balance of each project's life. The District expects the PMOR to effectively participate in the controls systems to facilitate the thorough, effective movement of each project through each step of the program, and to do so in a manner that causes all stakeholders, whether administrative District employees, school based staff, contractors or consultants, to buy in and commit to the use of the controls systems and the program they represent. The following items in this section are not exhaustive and respondents are encouraged to build upon the remainder of this section with their expertise and knowledge.

6.4.4.1 Document Control

E-Builder shall be the central repository for all project information such as contractual agreements, correspondence, design submittals, Requests for Information and associated responses, invoices and pay applications, change orders, and project documentation. In addition, the PMOR will be able to execute many of the day-to-day processes and procedures through embedded workflows that support core functions such as Schedule and Resource Management, Cost Control, Construction Administration, Change Management, Funding Management, and Claims Mitigation.

6.4.4.2 Schedule Controls

The PMOR is responsible for developing and maintaining project schedules. The schedule will show each of the District's projects assigned to the PMOR and provide milestones for each major phase of a project, as well as the steps between milestones--including at a minimum: planning, design, procurement, construction, and closeout. Each schedule must show the current status of each project, the phases that have been completed, and the target dates for the phases yet to be commenced. The PMOR will ensure that the schedule

RFQ FY20-192
Program Manager - Owner's Representative Services
Page 49 of 54

proposed minimizes any impact on the instructional operations of each school, through a collaborative process of engaging all school-based stakeholders at the initiation of the planning process. The PMOR will provide a report which states the assumptions, risks and reasons for the project milestones presented in project schedules as part of establishing an approved schedule.

The District understands that building a schedule is a critical element in the planning of a project. Accordingly, the District expects each initial project schedule, to include the planning and refining of the pertinent objectives and milestones among the items to be tracked. The District, or its agent, will work collaboratively with the PMOR to integrate and revise project level schedules into a single program schedule, which minimizes risk, maximizes resources and still meets projects-based objectives. Although some planning or assessment efforts may not move forward as a funded project, it is still imperative to schedule and track these preproject/pre-construction efforts with planning schedules. A program level schedule will be established prior to establishing project level schedules to ensure that project schedules are properly coordinated into the master program plan.

The PMOR will, accordingly, prepare project schedules, which commence with a fully approved project charter (approved by all project stakeholders and the Facilities & Construction Management Team) and track activities and milestones through the completion and turnover of the project to the school and the PPO staff. The PMOR will provide timely project updates indicating the progress, or lack thereof, of each project on a regular basis determined by OCP. The District expects the PMOR to provide OCP with the promptest notice possible of any slippage or delays on a project. The notice will, to the extent possible, explain the delay. If the reasons require further inquiry by the PM, the notice will indicate as much and will provide a timeframe for the provision of the explanation. The PMOR will also provide OCP with an action plan for addressing the slippage and recovery plan within a reasonable period subsequent to the determination of the cause of the delay.

6.4.4.3 Budgeting and Cost Control Process

The PMOR shall be responsible for the timely update of expenditures and forecasted costs based upon the current progress and or challenges of each project. While the PMOR tracks costs and maintains budgets at a project level, the CPCM will be aggregating all projects to produce a monthly Program Cash Flow model to assist Capital Budget with forecasting expenditures. The District may request that a planning or project definition service be provided for a prospective project and may provide a limited amount of funding for this purpose. Whether the District provides a planning budget or an overall budget amount for a project, the PMOR will work with the OCP to:

- 1. allocate the amount provided between the various components of the project
- 2. provide timely updates on expenditures and forecasts during planning, design, and construction; and
- 3. implement measures that will provide prompt notice to the District with respect to circumstances that might require changes to a project's budget.

The PMOR is expected to utilize each project charter as a means for establishing clear expectations and controls on the sources and uses of the project budget. The final project cost, as compared to the project charter budget, is a key metric upon which the District will inform the evaluation of the effectiveness of the PM's project management services.

The relationship between the project's construction budget and expenditures during construction is of concern to the District. The District expects the PMOR to track payments to the general contractor in conjuncture with the progress of the Work and to provide prompt notice to OCP or its agent when a payment request exceeds progress.

6.4.4.4 Change Management

The District sees the change management process as a critical and project long effort to control costs and quality. The PMOR is expected to deploy the District's program-wide change management process. The District recognizes that changes and discovered conditions are a reality in renovations of the District's portfolio

RFQ FY20-192 Page 50 of 54

but also recognizes that appropriate due diligence prior to beginning design, and especially before construction, are the most effective means to mitigate budget overruns in the Capital Program while also ensuring that completed projects meet their Charter's standard of quality. The District expects the PMOR to closely monitor and report on the impact of changes to the project's scope and schedule concerning the project's budget. This includes, beyond designer or contractor changes, the additional costs incurred by the District and PMOR for the management of the project, as in the event of negligent or dilatory performance by the general contractor, such costs may be charged against that general contractor.

6.4.4.5 Construction Oversight

The PMOR's project managers (PM) will be responsible for the day-to-day management of the District's projects during their respective construction phases. The project management team shall be comprised of personnel who are experienced in construction management of complex projects that may occur while existing operations continue; can make decisions promptly to ensure schedule compliance; can effectively manage cost of work and project budgets; and will adhere to proper record-keeping and maintenance of project documentation and files. The project manager and team members in each of their roles must be capable of ensuring that consultants and contractors are held to the terms and conditions of their agreements with the owner, and that they comply with the requirements of the review and inspection processes in an effort to execute the work efficiently.

It is essential that the PMOR establish a structure whereby the project management team works collaboratively with the Authorities Having Jurisdiction (AHJ) to be able avoid unnecessary delays to project schedules, and to be able to meet major milestones of Substantial Completion and Final Completion. As part of building quality into each construction project from the start, the District is interested in construction management methodologies and commissioning strategies, which eliminate protracted project closeouts and reliance on acceptance by the AHJ and punch list process alone.

The District expects the PMOR to provide a project construction schedule developed and maintained by the General Contractor (GC) or Construction Manager (CM). The District also expects a schedule to be furnished by the PMOR for each project linking the milestones of the GC's or CM's schedule. Additionally, the PM is expected to attend all meetings--including but not limited to, jobsite meetings, negotiations, change order review meetings, and Owner/PMOR meetings. Also, the PM is required to maintain logs/documentation of, including, but not limited to, project budget costs, project meeting minutes, shop drawing submittals, Building Department inspections, contractor pay applications and payments, Direct Owner Purchases (DOP), requests for information, responses to requests for information, field orders, change orders, quality deficiency reports, safety incident reports, claims, EEO compliance, M/WBE compliance, and testing. As such, the PMOR shall both track compliance and implement project assignments which ensure all requirements are met.

The PMOR shall oversee and manage the District's Direct Owner Purchase (DOP) program, at the assigned project level, by complying with the existing SOP, and with emphasis on enforcing the project contract requirements. The PMOR shall also work with local vendors, suppliers, and contractors to encourage participation, and actively review and process the DOP documentation on specific projects.

The PMOR shall consider safety in all phases of the project to ensure the health and safety of the students, staff and the public, at all times. During the construction phase, the PMOR shall work with the contractor to develop a comprehensive job site safety plan, and perform a thorough review of all job site safety plans to ensure compliance with the Districts' standards and specifications, OSHA requirements, and building code safety requirements. PM and team members shall make periodic visits to all active job sites to ensure contractor compliance with the approved job site safety plan.

6.5 Performance Evaluation

6.5.1 Program Evaluation

The PMOR shall, on a monthly basis, provide an executive summary of the performance of the program, while collaborating with the Cost and Program Controls team to ensure no duplication of efforts. The District has identified Standard Operating Procedures (SOPs) and categories of metrics, which respondents shall support, evaluate against benchmarks, and build upon based upon their professional expertise/industry best practices.

The monthly executive summary shall include the program and project performance measures listed in 6.7.9 of this RFQ. Project-level performance measures will be summarized at the program level, with emphasis placed on specific projects significantly contributing to overall performance outcomes.

6.5.2 PMOR Evaluation

The PMOR's performance shall be evaluated on a quarterly basis using a combination of objective metrics and evaluation criteria. OCP management will assess the PMOR's performance using instrumentation designed to evaluate performance in two major categories--Deliverables and Attributes. The evaluation shall be for the purpose of reviewing and scoring leadership and management competencies, relationships with stakeholders, and compliance with the technical requirements of the agreement. The specific criteria to be assessed are indicated on the sample evaluation form included as Attachment J.

The program and project-level metrics used to evaluate program performance (see 6.5.1 of this RFQ) and listed in 6.7.9 of this RFQ, along with other metrics specifically called out in this document, will be the basis for the objective contractual metrics to be used in evaluating the PMOR's performance.

Upon completion of the evaluation by the Executive Director, Capital Programs and appropriate OCP management, the results shall be discussed with the PMOR Lead and management team in a timely manner and any changes or corrective actions be implemented promptly.

6.6 Performance-Based Fees

The District is relying on the PMOR's ability to deliver projects on time and within budget using an experienced team's proven approach and methodology. To help ensure cost-effective delivery of projects throughout the term of the contract and during any extensions, a portion of the PMOR's fees will be performance-based.

The portion of the PMOR fees placed at risk is to be based on a cost-plus fee (profit) structure. The portion of the PMOR fees placed at risk would therefore be the agreed upon percentage of program costs the District agrees to pay the PMOR in addition to negotiated PMOR labor rates. Beginning with the first quarterly evaluation described in 6.5.2, the PMOR's fees would be subject to being discounted at an agreed upon rate.

The portion of the fee placed at risk (discount rate) would be determined during negotiations and contingent upon the PMOR's performance of the objective contractual metrics identified in 6.5.2, PMOR Evaluation. The relative weight of each objective metric used to measure performance will be determined and agreed upon during negotiations.

6.7 Scope of Services: Related Deliverables

The following is a general summary of PMOR Deliverables and is not intended to be an all-inclusive list of the requirements for the Agreement, between the Owner and the PMOR.

6.7.1 Schedule Controls per (Program & Project) - Monthly

- 1. Planning, Design, Procurement, Construction & Close-out
 - A. Develop Baseline Schedule (Project/Program)
 - B. Maintain Monthly Schedule Updates (Project/Program)
 - C. Variance analysis (Schedule / Budget) slippage

RFQ FY20-192 Page 52 of 54

6.7.2 Project Charter per Project – At Project Inception

- 1. Project Charter (Stakeholder & Sponsor approvals) Per Project
 - A. Assumptions
 - B. Risks & reasons for Project milestones

6.7.3 Budgeting & Cost Control Process per Project

- 1. Project Funds Allocation (PFA Project Budget set at project inception by CPCM; PMOR works closely with CPCM to make changes at all project milestones)
- 2. Evaluation of Pay Reg. (Consultants & Contractor) (Monthly)
- 3. Prompt Notice of Budget over-run (immediately upon identification)
- 4. Project Charter as Baseline (at project inception)
- 5. **Charter** Metric for Program Evaluation (at project inception)
- 6. Earned Value Project Management (Monthly)

6.7.4 Change Management per Program/Project –Monthly

 Report & Monitor impact of Changes (Integration Management) (Quality, Scope, Schedule & Budget)

6.7.5 Knowledge Management/Continuous Improvement at Program & Project Level - Quarterly

- 1. Collect and codify processes and practices as a function of continuous improvement efforts; review quarterly and incorporate current industry best practices and lessons learned into SOPs
 - A. Industry Best practices
 - B. Lessons learned (Log)

6.7.6 Program/Project Management Methodology Plan – Initiation Phase

- 1. Methodology Plan
 - A. Conception /Planning Management Plan
 - B. Definition / Design Management Plan
 - C. Procurement Management Plan
 - D. Execution / Construction Management Plan
 - E. Risk Management Plan
 - F. Close-out Management Plan

6.7.7 Staffing / Responsibility Matrix – Initiation Phase and Quarterly

- 1. Responsible, Accountable, Consult, Inform (R. A. C. I.) Matrix
 - A. During the initiation phase, provide a 12-month staffing plan that correlates with the projected status of individual projects and the overall program over the 12-month period.
 - B. On a quarterly basis, provide a 12-month staffing plan that evaluates each team according to the projected status of individual projects and the overall program over the 12-month period.

6.7.8 Procurement Management – Initiation Phase and Major Milestones

- 1. Vendor evaluation procedure
 - A. Develop procedure at initiation phase; ensure evaluations are completed at major milestones
- 2. Procurement delivery method matrix
 - A. Develop matrix at initiation phase; review and modify as necessary

6.7.9 Performance Evaluation at Program & Project Level – Monthly

These project and program measures of performance will both be included in the monthly executive summary of program performance completed by the PMOR. Project-level performance measures will be summarized at the program level, with emphasis placed on specific projects significantly contributing to overall performance outcomes.

RFQ FY20-192 Page 53 of 54

- 1. Monthly Executive Summary of Program performance
 - A. Schedule adherence
 - B. Budget vs Cost
 - C. RFI Rates (Logs)
 - D. Change Order Rates
 - E. Vendor performance at Milestones
 - F. Stakeholder satisfaction
 - G. Project Quality Adherence
- 2. Monthly Executive Summary of Project Specific Performance
 - A. Contractor's Progress Payments/ Monthly
 - B. Responses to RFI's
 - C. Field & Change Order root causes
 - D. Quality deficiency & Building Dept. inspection reports
 - E. Safety incidents
 - F. Claims
 - G. (EDDC) compliance
 - H. M/WBE compliance
 - I. Material Testing Results

END OF RFQ

Page 54 of 54

THE FOLLOWING SECTION 6.0 - SCOPE OF SERVICES IS FOR REFERENCE PURPOSES ONLY. IT SHOWS THE CHANGES MADE SINCE THE PREVIOUSLY UPLOADED VERSION.

6.0 SCOPE OF SERVICES

Section 6.0 begins with an overview of the purpose and objectives of the Broward County Public Schools SMART Program and overall Capital Program, as well as the scope and current status of the District's Capital Program. This overview is followed by three categories of District expectations of its Program Manager – Owner's Representatives (PMORs). These expectations center around the District's objectives in retaining owner's representatives, the PMOR's responsibility for various Capital Program relationships, and most importantly, the District's expectations related to the PMOR's approach to program management and its project management methodology. Respondents must specifically address each of the following sections in their Qualifications Statement of services and are encouraged to expand upon the sections below and/or provide additional or innovative value-added services based upon their own unique expertise and knowledge. The PMOR is expected to utilize the e-Builder program/project management system for all project-related activities. As such, respondents should include experience using web-based tools to facilitate document control. Respondents should also provide evidence of successful experience in delivering programs on time and within budget.

6.1 Purpose and Objectives of the Broward County Public Schools Capital Improvement Program

6.1.1 General Purpose and Objectives of Broward County Public Schools Office of Capital Programs (OCP) and Physical Plant Operations (PPO)

Broward County Public Schools (District) has as its mission a commitment to educating all students to reach their highest potential. The strategic goals of the District are High-Quality Instruction, Safe & Supportive Environment, and Effective Communication. The Office of Capital Programs (OCP) indirectly contributes to the first goal through its effective delivery of much-needed capital improvements at schools. The OCP is committed to providing safe schools that create a supportive learning environment to ensure that it maximizes the limited resources of the District to meet the needs of every school. In delivering services and much needed capital improvements to schools, the OCP seeks to engage each school community in a manner that meets the third goal. Some of the District's values, which directly relate to services provided by the OCP include positive stakeholder involvement, holding everyone to the highest ethical standards to achieve excellence, high-quality customer service, and valuing the diversity of our community.

The Office of Capital Programs (OCP), along with Physical Plant Operations (PPO), must ensure that facilities meet the operational needs of 241 schools serving nearly 270,000 pre-kindergarten to twelfth grade students and approximately 175,000 adult students. In meeting the needs of schools, OCP and PPO must manage the maintenance, repairs, and improvements of the portfolio of facilities in an effective and efficient manner as trustees of public assets.

The Office of Capital Program's (OCP's) primary objective is the continued management of the 5-Year Capital Plan, while the Physical Plant Operations Department's (PPO's) primary objective is to move from a demand maintenance cycle to life cycle management of assets as part of its current and long-term strategic plan.

6.1.2 Purpose and Objectives of the Broward County Public Schools SMART Program

In order to clearly identify and prioritize the needs of schools, in 2014 the District completed a district-wide comprehensive Facilities Needs Assessment to obtain accurate physical plant structural and building system deficiencies listings, as well as classroom infrastructure and relational supporting educational space deficiencies, as measured against established District standards. The objective of the needs assessment process was to provide an objective, data-driven overview of the District as a whole, as well as a strategic planning tool for prioritizing the most critical needs facing our schools for safety and security; repairs and renovations; and technology and technology infrastructure.

These needs were organized into the SMART (Safety, Music & Arts, Athletics, Renovation, and Technology) program, and an \$800 Million General Obligation Bond was passed by voters on November 4, 2014 to fund the program. A corresponding 5-Year Capital Plan was developed. The 5-year Capital Plan from February 24, 2015 included both the \$800 million dollars in bond-funded projects, as well as another \$184 million in new and previously funded projects, for a total Capital Program of \$984 million.

The District decided to engage the assistance of a team of construction industry professionals as the most flexible, cost-effective means of forging its projects into a coherent program. To accomplish this objective, the District engaged

Page 37 of 54

one Program Management firm to act as its Owner's Representative (PMOR), and another firm to provide Cost and Program Controls (CPCM) services.

6.1.3 Current Status of the Broward County Public Schools Capital Improvement Program

The Capital Program ("Program" or "Capital Program") shown in the 2019-20 Adopted District Educational Facilities Plan (ADEFP) consists of SMART program projects funded by the 2014 General Obligation Bond, additional DEFP projects that were added to the SMART program, and previously-funded projects not included in the SMART program.

Adjustments have been made to SMART program project budgets as construction contracts have been awarded. The current Program budget is planned at over \$1.3 Billion. The program has faced several challenges since inception. An increased awareness of the importance of understanding and adhering to district design standards and local building code, especially those related to roofing, is expected to have a positive effect on roofing project timelines moving forward. If the planned process improvements continue, and if market conditions remain stable or improve, the Program is to be executed by the end of the 2025-26 school year.

The Program is comprised primarily of a variety of renovation projects with about 65% of scope (\$879 million) focused on building exterior envelope. Roof recovery and replacement projects represent the majority of building envelope renovations. The Program also consists of \$347 million in HVAC (25%), and \$95 million in life safety systems (10%) projects. Less than eight percent, or \$101 million, of the district's capital program consists of new construction projects.

The design work for approximately 71% of projects (\$894 million) has been completed. Most of the projects left in the design phase are year 4 and 5 projects, although more complex projects from earlier program years represent 24% (\$115 million) of the design work to be completed. Construction has been completed for about 7% (\$44 million) of the projects in the Program. Additional information related to the scope of the individual projects may be found in the Bond Oversight Committee (BOC) quarterly reports (click here) and 2019-20 ADEFP (click here).

A comprehensive listing of all active projects by life cycle status is included in this RFQ as Attachment I. Additional projects may be identified during the term of the contract and any extensions. As the District's goal is retain two Owner's Representatives to manage its Capital Program, each PMOR will be assigned a portion of the program to manage, while a third firm will be engaged under a separate solicitation to provide Cost and Program Controls Management for the overall program.

The remainder of Section 6.0 is organized around the following categories of District expectations: 1) objectives for retaining Owner's Representatives; 2) PMOR's responsibility for various Capital Program relationships; and 3) PMOR's approach to program management and project methodology. District expectations of the PMOR are followed by criteria upon which the program and PMOR performance will be evaluated, and a summary of PMOR deliverables.

6.1.4 Broward County Public School's Objectives in Retaining Owner's Representatives to Manage its Capital Program

In order to be effective in meeting the facility and operational needs of the Schools in a prioritized, transparent, and resource conscious manner, the District must continue to manage the capital projects of its Schools as a coherent program. There is a need for coordination and synergy at each point in the process of planning, designing, and building capital projects for this District. Moreover, there is a need to achieve this coordination through means that are both flexible and cost-effective. The District has determined that engaging the assistance of a team of construction industry professionals organized in two Program Manager - Owner's Representative (PMOR) agents is the most flexible, cost-effective means to achieve the coordination and synergy necessary to meet the demands of the Schools.

A PMOR offers flexibility through its capacity to engage professionals with specific expertise at appropriate points in the planning and implementation of projects. This is also cost-effective in that staffing is scalable according to the volume of work in the Program; thereby limiting the District's financial liability during periods when volume does not require an extensive staff. Scalable staffing is of interest to the District as the capital needs, and available funding, varies from year to year.

The PMORs will also enhance the District's efficiency by keeping the District current with the latest developments in management systems and practices. For flexible staffing to be effective, there must be a core set of systems in place that enables both the owner's representatives and the District's professionals to contribute their knowledge and work product to, and resource information from, a central, coordinated repository of data.

As such, the District has selected e-Builder as the required project management software to serve as the central repository of data for the Capital Program. The PMOR shall utilize e-Builder program/project management system for all project-related activities. By implementing, maintaining, and upgrading e-Builder, the District's management information systems appropriate for the District's projects, the owner's representatives will facilitate the efficient and effective use of information throughout the District's capital projects. Through benchmarking the best practices of project management on a local and national scale, the PMOR will assist the District's senior management in establishing metrics that will identify areas where the Capital Improvement Program is functioning efficiently, and areas that require improvement.

The role of the PMOR will be to manage all assigned capital projects from their current status (which may include projects at 100% of the design phase) through commissioning and closeout. The PMOR, in its management of the District's capital projects acts as the District's representative and actively seeks to identify issues and resolve issues. Where issues cannot be resolved by the PMOR, the PMOR's responsibility as the manager of projects is to provide options, and/or recovery plans, to the Office of Capital Programs ("OCP") for review and direction.

While the Program has an established sequencing, the PMOR will be responsible for continually reviewing and making recommendations to optimize the sequencing of the assigned portion of the program considering changing external and internal conditions and needs of the District.

As such, the PMOR should demonstrate capabilities in a wide range of renovations in facility portfolios of a similar nature to that of the District's. The PMOR should also demonstrate capabilities in managing renovation work within operating facilities. Demonstrated capabilities to provide high quality customer service to facility stakeholders will be critical to the success of the PMOR and the overall Program. Furthermore, the PMOR will be responsible for effectively communicating with the numerous stakeholders of school capital renovation projects.

The District expects the members of the PMOR to apply their professional expertise, experience, judgment, and information management systems in a manner that will facilitate the District's efforts to develop, maintain, and enhance the relationships necessary to meet the District's purpose and objectives.

6.2 PMOR Responsibility for Developing, Maintaining, and Enhancing Capital Program Relationships

The PMOR Lead (Program Manager/Director) is wholly responsible for the relationship between the PMOR and the District. As such, the District considers the PMOR Lead to be a crucial member of its Capital Program team. The PMOR Lead plays a particularly important role in mobilizing the PMOR in its initial application of the program approach and project methodology, and in managing the implementation of the PMOR's strategic plan. The District expects the PMOR Lead to identify and, as needed, work with the management of the District's Office of Capital Programs to initiate critical relationships within the District and between the District, the PMORs and schools. Moreover, the District expects the PMOR Lead to assist and enhance the efforts of the management of the Office of Capital Programs to instill the sense of collective identity and collegiality among all members of the Program necessary for the successful operation of the Program. As such, the District expects the PMOR to take primary responsibility for coordinating the relationships between the team members responsible for planning, design, construction, controls, and administration in a manner that is consistent with best practices in the pertinent industry, and which facilitates prompt, complete and smooth hand-offs from one function to the next, thereby minimizing gaps and/or delays in the progress of the Program.

The relationships for which the PMOR Lead will be responsible for developing, maintaining, and enhancing include, but are not limited to, the following areas: 1) the PMOR staff assigned to the District; 2) District Senior Management; 3) Schools; 4) Project Design Professionals and Contractors; 5) the Cost and Program Controls Manager (CPCM) and other agents for the District; 6) the District's departments related to Planning through Closeout; Procurement; Economic Development and Diversity Compliance; Physical Plant Operations (PPO); and the District's Building Department and other Regulatory agencies. A description of the relationship between the PMOR and each area follows.

6.2.1 Relationships within the PMOR

The District expects the PMOR Lead to direct the day-to-day operations of the PMOR and to ensure that each individual within the PMOR: 1) understands his or her role and responsibilities; 2) performs the tasks required by that role and those responsibilities in a manner consistent with the purpose and objectives of the District; and 3) administers the operation of the PMOR in an engaged, open, and professional manner. If it comes to the attention of OCP management that any member of the PMOR team is not performing per the three prior expectations, the PMOR Lead will be notified of the need for immediate corrective action, and/or the need for a change in the PMOR team staffing. Similarly, the PMOR Lead will notify OCP management should the Lead become aware of a PMOR team member performance problem as defined above. Proposed changes to the staffing of the PMOR team during the term of the contract and any extensions are to be presented to the Executive Director, Capital Programs and appropriate OCP management for review and approval prior to staff changes or new assignments. This review process includes proposed changes to the PM Lead and/or Principal-in-Charge. The Superintendent and School Board must also approve proposed changes to the PM Lead and/or Principal-in-Charge.

6.2.2 Relationships with District Senior Management

In a support role to the District's Senior Leadership Team, the PMOR staff will have a direct, working relationship with several senior members of the District, but will report on a day-to-day basis to the Office of Capital Programs (OCP) leadership and staff. The District's Senior Leadership Team will be relying upon the PMOR staff to: 1) implement project management processes as determined by OCP that coordinate and integrate the District's projects into a program; 2) work to improve and coordinate the District's relationships with regulatory agencies, municipal agencies, Schools and the public where applicable; 3) anticipate and identify issues; 4) provide prompt and comprehensive notice of issues as they arise; 5) work with OCP staff to identify those issues which require the input of District management, and those issues which are resolvable by the PM; 6) furnish accurate and timely information regarding the progress of projects and emerging issues; 7) offer options for responding to issues and needs as they arise, and do so in a manner that assesses the risks and opportunities afforded by those options; 8) implement decisions made by District management; and 9) report promptly and thoroughly on the consequences of implementation.

The District expects the PMOR staff to perform services related to its relationship with the District's Senior Leadership Team in a manner that brings the entire weight of the PM's professional experience, expertise and judgment to bear. Moreover, the District expects the PMOR to perform these functions in a manner that supports and facilitates the District's management operations as a collaborative team, and in particular as a team in which each of the members understands its role, and respects and appreciates the roles and contributions of other team members.

The PMOR shall provide a scalable staffing structure which is responsive to the variability and seasonality expected in a capital program. This structure, while providing the appropriate and necessary amount of responsiveness to the District's Capital Program, shall do so in a manner which consistently meets the expectations of the relationship just described. Although there is a need for adaptation of staffing, the PMOR shall have a consistent Lead for the term of the contract. The requirement of a consistent PMOR Lead is discussed in more detail under section 6.3.4.

6.2.3 Relationships with Schools

The OCP oversees the planning, design, and construction services for all facilities within the District's portfolio. The District expects the PMOR to be essential to its efforts to deliver timely, cost-effective and productive services to schools. The nature of the PMOR's relationship with schools, instructional departments, and other stakeholder groups may vary during the project's lifecycle. However, with clear and accurate project details provided to school administrators and other school-related personnel at the onset of a project, the PMOR will be able to successfully manage the requirements and expectations of the project.

The District expects the PMOR to work closely with curriculum and instructional staff during the initiating phase of a project to elicit the details of the school's project and programmatic requirements. During this stage, the PMOR must

Deleted: The relationships for which the PMOR Lead will be responsible for developing, maintaining, and enhancing include, but are not limited to, the following: 1) the PMOR staff assigned to the District: 2) District Senior Management; 3) Schools: 4) Project Design Professionals and Contractors; 5) the Cost and Program Controls Manager (CPCM) and other agents for the District: 2nd 6) the Districts Planning through Closeout areas. A description of the relationship between the PMOR and

demonstrate the ability to balance stakeholders' varying interests and ensure that the project team communicates and interacts with stakeholders in a professional and cooperative manner to obtain positive outcomes.

The PMOR shall also demonstrate an acknowledgement that the relationship with schools is based on ensuring school administrators have a clear understanding of the roles and responsibilities of team members from project inception to closeout and full turnover to PPO. Although the relationship will change over the course of the project and interactions and interface will vary depending on the phase of the project, it is important that clear lines of communication remain intact, and school administrators and team members respect the limits of each other's responsibilities and authority. Hence, it is important that the PMOR establish a sign-off process during each phase of the project to ensure expectations are aligned as the project evolves.

The PMOR must also establish a process that clearly defines milestones and activities during every phase of a project, and particularly during the construction phase where maintenance of operations is key to school administrators. The PMOR must be prepared to work with school staff in outlining logistical solutions to keeping school functions in progress without any disruption to the teaching or student learning experience, thus preserving the integrity of the educational process. It is essential therefore that the PMOR establish a key goal of maintaining a zero-negative impact to student instruction during all phases of the project's lifecycle.

6.2.4 Relationships with Project Design Professionals and Contractors

The PMOR will have direct day-to-day responsibility for the oversight and management of the District's design professionals and contractors awarded projects. In establishing this Program, the District must be a client of choice for design professionals and contractors in the South Florida construction market. The District expects the PMOR to facilitate achieving this goal through: 1) the implementation of clear and thorough procedures for each phase of project delivery; 2) consistent and fair implementation of the procedures; 3) supporting District staff efforts in developing, managing and maintaining qualified pools of vendors; 4) timely review of invoicing for all vendors; 5) developing working relationships with the design and construction providers in the South Florida market that are predicated upon respect and trust, that support the free flow of ideas and innovation, and that consistently demonstrate to design professionals and contractors that their ideas and concerns will not only be heard but addressed in a mutually beneficial manner; 6) facilitating problem-solving and decision-making among design and construction entities in an impartial and goal-driven manner; and 7) fostering a collaborative environment between design professionals, contractors, and key stakeholder departments, such as the Building Department, Office of the Chief Fire Official, and PPO to address issues and concerns early in the project development process, and throughout the multiple phases of a project.

The PMOR is also expected to leverage its relationship with the Building Department to improve the permitting and inspection processes. This requirement is particularly crucial to the District's efforts to apply lessons learned to the roofing sub-permit process. The District expects the PMOR to work with the Building Department to create a mutually beneficial process that results in fewer revisions to permit packages, and fewer re-inspections. The PMOR's ability to create, maintain, and enhance the relationship between itself, the Building Department, and Design Professionals and Contractors is critical to the PMOR's ability to deliver the Program on time and on budget.

6.2.5 Relationships with Cost and Program Controls Manager (CPCM) and Other District Consultants

Each PMOR is expected to actively support the program controls and procedures as set forth by District staff, or by its assigned agent. While each PMOR will be assigned discrete and independent projects that comprise the Program, the District expects each PMOR to focus on delivering their assignments with fidelity to the District's plan of a single cohesive program delivering critically needs improvements to all schools in Broward County. As such, a PMOR is expected to develop working relationships with District agents that are predicated upon respect and trust, that support the free flow of ideas and innovation, and that consistently address issues in a manner that ensures the District's best interest are primary to all actions.

The District will be relying upon the PMOR to: 1) implement Program and project management processes that aid in coordinating Owner's Representative and Cost and Program Controls function; 2) anticipate and identify issues related to the different functions 3) work with OCP staff, the Cost and Program Controls Manager (CPCM), or other assigned agents thereof, to identify those issues which require the input of District management, and those issues which are

resolvable by the PM; 4) provide prompt and comprehensive notice of issues as they arise to the appropriate District Staff or its assigned agent; 5) furnish accurate and timely information regarding the progress of projects to the Cost and Program Controls manager; 6) offer options for responding to issues and needs as they arise, and do so in a manner that assesses the risks and opportunities afforded by those options; 7) implement decisions made by District management; 8) report promptly and thoroughly on the consequences of implementation; and 9) incorporate lessons learned from the work of the Program, regardless of which agent of the District has gleaned such lesson so that the continued planning, design, and construction work of the District is clearly viewed as an effective continuous improvement process.

The PMOR, while working closely with the CPCM, will be responsible for certain project-based estimating tasks, while the CPCM will be responsible for more of the Program-based estimating functions. The PMOR staff will handle estimating tasks, including but not limited to, contractor's Schedule of Value review and approval, Earned Value analysis for contractor's monthly pay applications, evaluation and verification of unit prices for bidding, and during construction, monitoring, reviewing, and analyzing change orders under \$25,000. The PMOR will also be responsible for the review and analysis of Construction Services Minor Projects (CSMP) bids, and consultant fee proposals. The CPCM will be responsible for the Program level estimates and forecasts, project budget estimates, and change orders exceeding \$25,000.

6.2.6 Relationships with Planning through Closeout

OCP staff will be relying upon the PMOR staff to: 1) implement program and project management processes that coordinate and integrate the District's projects into a program; 2) anticipate and identify issues; 3) work with OCP staff to identify those issues which require the input of District management, and those issues which are resolvable by the PM; 4) provide prompt and comprehensive notice of issues as they arise; 5) furnish accurate and timely information regarding the progress of planning and design activities, along with any emerging issues; 6) offer options for responding to issues and needs as they arise, and do so in a manner that assesses the risks and opportunities afforded by those options; 7) implement decisions made by District management; 8) report promptly and thoroughly on the results of implementation; 9) provide constructability reviews during the design phase of each project; 10) ensure the timely and effective closeout and commissioning of all projects; 11) communicate and incorporate lessons learned throughout the Program; and 12) engage in professional activities within the planning, design, construction, and construction management communities such that the PMOR will be able to keep the District informed of current trends, developments and best practices within those communities.

6.2.6.1 Relationships with Planning and Design

The PMOR staff reports directly to the PMOR Lead, with principal day-to-day responsibility for the planning and design functions assigned to the PM. OCP staff will facilitate planning efforts, and the PMOR preconstruction staff will both serve as a resource in support and monitor conceptual design to ensure compliance with Program standards and continuity in project design. The District expects the PMOR staff to distinguish between project-driven and programmatic changes to a design, to track each set of changes appropriately, and to ensure that the distinction between project and programmatic changes is maintained throughout the District's planning and design processes (i.e. ensure that programmatic changes are appropriately addressed within each project, while project-specific changes do not become program-wide prostrictes). The PMOR staff is expected to evaluate the available procurement methods on a per project basis and make a formal recommendation to OCP, which provides support that demonstrates the best balance of cost, timeliness, and quality. The District currently utilizes five methods of procurement: 1) Design-Bild-Build; 2) Construction Manager at Risk; 3) Continuing Services; 4) Design-Build; and 5) A hybrid method including any of the previous methods along with in-house construction staff.

The PMOR staff will support OCP in the selection of design professionals of record, where applicable to the selected procurement approach, and work closely with OCP staff to manage each project's transition from planning to design. Once a design professional for a project is selected from the qualified pool of designers, the PMOR staff will assume the lead in managing the project, and OCP staff will serve as a resource to the design team for each project. When a design-build project delivery method is approved to procure the project,

the PMOR shall develop the necessary design criteria package as part of its management of design development.

6.2.6.2 Relationships with Construction and Closeout

The PMOR, through its Project Management staff, will be responsible for the planning, scheduling, and managing of the construction phase from award through project closeout. OCP staff will provide guidance to the PMOR Project Managers, however, the PMOR Project Managers will be solely responsible for managing the Contractor's day to day construction activities, ensuring compliance with the District's Specifications, Standards and Policies, coordinating the construction activities with on-site staff as well as District staff, reviewing, approving and processing of all required construction documentation, and overseeing the project punchlist and closeout activities. The PMOR staff is expected to work closely with the Contractor and Project Consultant to ensure the contractor maintains the project budget and schedule, minimizes disruptions to the school, and provides the Owner with sound, quality construction through industry standard QA/QC programs. At Substantial Completion, the PMOR staff shall work closely with the Project Consultant to create the project's Substantial Completion punchlist, and work with the Contractor to ensure that all items on the punchlist are properly resolved. At Final Completion, the PMOR staff will review all project closeout documents and warranties, schedule all required training, and submit all required closeout documentation.

6.2.7 Relationships with Procurement

The PMOR is responsible for developing Support Documents for Requests for Qualifications, Requests for and Proposals, and Bid Support Documents according to regulatory and District requirements. Once the solicitation package is complete, the PMOR turns it the package over to procurement. The PMOR is responsible for participating in Pre-Bid & Pre-submittal Conferences, answering RFIs, facilitating communication with the design teams, and assisting procurement in any and all ways, as directed, to have successful solicitations. The PMOR shall communicate with and engage procurement regarding any contractor, subcontractor, etc. related project issues. The PMOR may be required to provide staff augmentation for the procurement process, which shall be approved by the Director of Procurement & Warehousing Services. While procurement has primary responsibility for the management and oversight of personnel involved in the procurement process, the PMOR is expected to include any staff augmentation in the R.A.C.I. chart prepared for the OCP.

6.2.8 Relationships with the Economic Development and Diversity Compliance Department (MWBE Compliance)

The PMOR is responsible for supporting EDDC with the compliance process and the execution of Policy 3330. The PMOR may be required to provide staff augmentation to provide assistance with Supplier Diversity Outreach Program compliance, contractor outreach, and other assistance in support of Policy 3330. While EDDC has primary responsibility for the management and oversight of personnel involved in the facilitation of Policy 3330 in support of the SMART Program, the PMOR is expected to include any staff augmentation in the R.A.C.I. chart prepared for the OCP.

6.2.9 Relationships with Physical Plant Operations (PPO)

An important responsibility of the PMOR is to ensure that the District's PPO staff, both school and District-based, are fully trained in the use and maintenance of all building systems and equipment. Once Substantial Completion has been achieved, the PMOR will be responsible for coordinating and managing all warranty items during the 12-month warranty period, including but not limited to, scheduling all owner required training sessions, enforcement of all specification requirements for the operation, maintenance training, testing, and start-up of equipment. The PMOR shall be responsible for distinguishing between routine maintenance of the building structure and its components as opposed to defects requiring corrective action by the General Contractor.

All training shall be documented, and video-recorded for future reference and use in training new employees. The PMOR will also be responsible for transitioning all warranty responsibilities to the District's maintenance department after the first year.

The PMOR shall also be responsible for ensuring that the District's established commissioning process that commences at the initial phases of a project and continues through the end of the warranty period is managed effectively, and that the proper functional testing of equipment and systems occurs to verify that they are working and performing, as specified. This process shall be documented by the PMOR and any problems that arise be resolved as quickly as possible.

The final hand-off to PPO occurs at the completion of 12-month warranty period and surveillance of the project and its components, at which time PPO will be responsible for the operations and maintenance, including administration of all material and equipment warranties.

6.2.10 Relationships with the District's Building Department and other Regulatory Agencies

Each PMOR shall fully comprehend the role of internal and external regulatory bodies and Authorities Having Jurisdiction. Each PMOR must demonstrate an understanding of the objectives of the District's Building Department in providing professional plan review and inspection services for minor and major projects within the District; the responsibilities of reviewing and approving construction documents for compliance with various building codes and conducting inspections to insure that work is being done in accordance with approved plans, and such a manner that the safety, health and welfare of individuals utilizing the constructed facility are preserved and protected.

It is crucial to the success of the individual projects and the overall program that the PMOR recognizes the importance of compliance and effectively managing the processes that govern regulatory requirements, while balancing the goal of executing a project. By establishing a framework that addresses the competing priorities of project management and compliance to standards, codes, and regulations, the PMOR must exhibit a clear appreciation of the effort required to achieve both sets of goals. Incorporating this understanding into the initial phases of a project through the Project Charter process, the challenges of compliance can be properly documented and dealt with in mitigating risks. This effort must be carried through the ensuing phases of a project, such as in bidding and award, and construction phases. Hence, the PMOR must indicate knowledge of the permitting process during design, as well as the inspection process during construction, with particular attention to the specific requirements for approval of shop drawings and sub-permits for roofing.

It is essential that the PMOR recognizes and documents the necessity of compliance with Authorities Having Jurisdiction as a key element of project execution and factors the impact and level of priority during its preparation of the Risk Register. Through early acknowledgement of such requirements and their effect on project and program progress, the PMOR will be able to showcase its efforts in ensuring adherence to internal and external codes and regulations, and how that directly correlates to the success of the program.

6.3 District Expectations of the PMOR's Approach to Program Management

6.3.1 Overview

The District expects the Program Manager – Owner's Representative (PMOR) to deliver projects that maximize taxpayer dollars in a timely and effective manner, while fully complying with District and other regulatory agency requirements. In addition to successfully implementing the relationships critical to the success of the District's Capital Program, the strategic approach and methodology employed by the PMOR to successfully address the inherent risks, challenges, and day-to-day functions associated with launching and managing a program of this size and complexity are key to achieving this goal.

Broward County Public Schools expects the PMOR to use its professional expertise and knowledge to engage in a comprehensive approach to the management of the District's Capital Program. While the District expects the PMOR to perform all the tasks necessary to meet its standard of care and to do so in a manner that is consistent with that standard of care and the objectives and expectations of the District, the PMOR is expected to take a system approach to contribute to the optimization of the overall Program, not simply particular tasks or components thereof—in short, the District expects true Program Management from each PMOR firm. Program level activities the District expects the PMOR to provide included but are not limited to day-to-day management and administration of the program and PMOR staff and sub consultants, staff augmentation, holistic management of the planning, design, pre-construction, and

construction, procurement management and support, contract management, contract negotiations, program-wide reporting, interaction with the School Board and District senior leadership, the Bond Oversight Committee and other Stakeholder groups providing capital program oversight, community outreach, warranty management and asset preservation.

The District has decided to keep the Cost and Program Control function separate from its Program Manager-Owner's Representative function. While this separation of duties creates a system of checks and balances, it also creates the need for functional relationships, collaboration, and communication to enable both types of Program Managers to successfully fulfill their responsibilities. Because the Cost and Program Controls Manager will be responsible overall program costs and controls, the PMOR will also be responsible for ensuring the execution of improvements and changes made at the Program level across individual projects. Respondents must indicate their understanding of the working relationship between the PMOR and CPCM teams and include a plan for how program-level Cost and Program Controls improvements and changes will be executed.

As part of its project management methodology, and in keeping with a strategy for risk management, the PMOR is expected to utilize the Enterprise Project Management software, known as e-Builder, for the full term of the Agreement, as well as any optional extensions, as may be granted by mutual consent. The e-Builder software is the exclusive Program/Project Management system to be used throughout the entire lifecycle of each project to track all key elements and activities.

The PMOR's success in implementing its strategic approach to program management and its project methodology will be evaluated quarterly by OCP management throughout the term of the contract and any extensions.

6.3.2 Mobilization Plan

Although the PMOR will be assuming responsibility for a set of projects that have been previously managed as part of a program, to ensure as seamless a transition as possible, the District expects the PMOR's mobilization efforts to immediately focus on both evaluating projects and developing an approach for its share of the program. Respondents shall propose a detailed mobilization plan. The mobilization plan must clearly demonstrate that the respondents recognize that time is of the essence, and provide a timeline with key activities at 30, 60, and 90 days.

6.3.3 Program Management Plan

At the program level, the District expects the PMOR launch team to analyze the overall program and begin developing the Management Plan that will form the foundation for the assigned portion of the Program. The District expects the PMOR 's strategic approach to include an emphasis on the program and project level activities most critical to successfully managing the risks, challenges, and daily activities most relevant to the remaining phases of the Program. As such, the District is particularly interested in how the PMOR will mitigate risk related to current local construction market conditions, construction schedules and budgets, and permitting and inspections. Expected activities include, but are not limited to the following: 1) Meeting with OFC management and the CPCM to review and validate the program master schedule and budget; 2) Reviewing previously established standard operating procedures (SOPs); 3) Meeting with OCP management and the Building Department to review key issues and lessons learned related to District design specifications, permitting, and inspection, as well as MWBE goals and pre-qualification of sub-consultants; and 4) evaluating other risk factors.

At the project level, the District expects the PMOR to identify key issues that might impact the ability of the PMOR to move projects forward. Expected activities <u>include but are not limited to: 1) Validating project scope</u>, <u>budget</u>, <u>and schedules</u>; 2) Developing initial project management plans.

6.3.4 Leadership

The District expects the PMOR to provide a set of comprehensive program management services and deliverables that encompass both program and project level activities, and which are performed consistently and effectively throughout the term of the contract and any extensions.

Deleted: include, but

The District believes leadership is key to the PMOR's ability to fulfill this requirement. As such, the PMOR shall have consistent leadership (PMOR lead/Program Manager/Director) for the term of the contract and any extensions. Proposed changes to the PMOR Lead and/or Principal-in-Charge shall be brought before the School Board for approval.

The PMOR lead will be responsible for day-to-day program management and communication as well as program-level activities such as interaction with and reporting to the School Board, District Senior Leadership, the Bond Oversight Committee, community stakeholder groups, and key district personnel; staffing, contract management, and program level reporting.

6.3.5 Staffing

In addition to the scalable staff required to manage the Program, the PMOR may be required to provide scalable staff augmentation for the Procurement and Warehousing Services Department, the Economic Development and Diversity Compliance Department, and The Office of Capital Programs.

The PMOR is required to maintain, and provide to management, on a quarterly basis, a Responsible, Accountable, Consult. Inform Matrix (R.A.C.I.) that includes both its staff and the staff of its subconsultants.

6.3.6 District Design and Material Standards and Design Criteria

The PMOR is required to actively participate and contribute to the continuous improvement and updating of the District's Design and Material Standards and Design Criteria. The PMOR's contribution shall be for the benefit of the owner and take into consideration, code requirements and restrictions, life-cycle costs and value, and be aligned with industry standards and best practices. The PMOR's input shall be in collaboration with the District's Standards Committee which is comprised of representatives from OCP, Building Department, PPO, and other District stakeholders, as necessary. In developing its mobilization plan, the PMOR shall meet with the Building Department to gain a working knowledge of the inherent differences between the District's Master Permit and Sub Permit requirements and processes related to roofing.

6.3.7 Procurement Management

The PMOR will serve as a key team member in ensuring the Program is continuously and consistently engaging the contractor and design professional community within South Florida, to maximize participation and competition. The PMOR's management of the projects shall also include a timely, consistent, and transparent vendor evaluation procedure that seeks to foster continuous improvement rather than simply assess penalties. Results of vendor evaluations will be used not only to inform PMOR performance, but to assist in the vendor selection process.

The PMOR shall also provide to OCP a recommended procurement strategy from the five (5) delivery options available to the District for each project within the context of the project and overall program objectives. The District's five delivery options are the following: 1) Design-Bid-Build; 2) Construction Manager at Risk; 3) Continuing Services; 4) Design-Build; and 5) a hybrid method including any of the previous methods along with in-house construction staff.

The District expects the PMOR to explicitly define a process by which these recommendations are to be made that shall minimally include a risk assessment of the procurement/ project options along with an optimization of cost, time, and quality. A procurement delivery method matrix shall be developed and maintained at the project level and the process and matrix summarized at the Program level.

The PMOR will work with the procurement department to avoid unnecessary project delays or bid protests by ensuring regulatory and District requirements are being followed regardless of delivery method. The PMOR will also work with the procurement department to ensure solicitation packages are complete and comprehensive in terms of 1) proper project scope; and 2) performance expectations and compliance requirements as applicable relative to safety, cost, schedule, quality, site logistics, close out documentation, submittals, engineering, manufacturing installation quality, coordination, delivery, start-up commissioning, and schedule objectives for each.

As the Owner's Representative, the PMOR will serve as an integral part of the District's negotiation team during the negotiation phase of the solicitation process. For each solicitation, the PMOR shall assign a lead negotiator, as a member of the negotiating team. The lead negotiator must be familiar with market conditions, hourly rates, labor burden rates, M/WBE requirements, and other factors that are critical in determining a fair fee and arriving at a mutual agreement with the selected firm or company.

6.3.8 Reporting

The PMOR lead will be expected to work with the OCP and CPCM to prepare a series of program status reports on a monthly and/or quarterly basis. As part of this requirement, the PMOR shall, on a monthly basis, provide an executive summary of the performance of the program, while collaborating with the Cost and Program Controls team to ensure no duplication of efforts. Specific reporting requirements associated with the executive summary are detailed in 6.5.1 of this RFQ.

Stakeholders for whom the PMOR will be required to prepare and present reports include the School Board, Superintendent, District senior leadership, and community stakeholder groups. A significant amount of reporting is required to be prepared quarterly for the District's Bond Oversight Committee. The PMOR lead will be expected to coordinate the timely delivery of transparent, accurate, and user-friendly reports that highlight key areas of progress or concern. The latest Bond Oversight Committee Report (BOC) may be found here (Click here).

6.3.9 Community Outreach and Public Involvement

As part of the PMOR's approach to communication, the District expects the PMOR to provide services that help facilitate the District's communication goals for the Program related to community outreach and public involvement. These services may include identifying and assessing stakeholders: planning, implementing, and facilitating public meetings; designing and producing communication materials; and developing and implementing educational programs. The District will be relying upon the PMOR to deliver these services using a variety of informational tools including, but not limited to, electronic/social media, print media, virtual and physical meetings, videos, and websites—customized to different target audiences.

6.3.10 Operational Integration

All projects executed as part of the Capital Program shall do so with a full understanding of the complete life cycle of systems impacted by each project. The PMOR shall, therefore, ensure that ongoing operations and maintenance, along with life cycle costs, are integrated during the definition stage of each project and delivered as part of each project's closeout. The PM's professional expertise, combined with a robust set of master specifications and guidelines, shall be complemented by engaging and collaborating with PPO to ensure a seamless transition from the closeout of the project to ongoing operations.

6.3.11 Knowledge Management

In addition to the PMOR bringing current industry best practices to bear in all its services, the PMOR is to participate in the District's system to collect and codify processes and practices utilized in the implementation of the Capital Program. The system deployed by the District, or its agent, shall ensure that the experience gained in executing the program becomes common knowledge of the organization through a clear process of capturing, retaining, and sharing information, regardless of changes in staff, process or technology. The PMOR should anticipate that the knowledge management system deployed will go beyond a latent system of collecting information and be robust enough in its workflows for lessons learned to be imposed on the ongoing program as part of a continuous improvement model.

6.4 District Expectations of PMOR's Project Management Methodology

6.4.1 Overview

The PMOR is expected to integrate into the project methodology a process of gatekeeping, between each phase, to assess, report and mitigate project and program risk, by defining and executing a comprehensive approach to risk

management that includes a qualitative risk analysis through a Risk Register, and quantitative analyses to track program performance. These analyses shall be conducted in conjunction with the Cost and Program Controls Manager. With the potential for constraints, both financial and workforce, the District expects the PMOR to establish a formal risk management approach that identifies negative risk occurrences impacting scope, schedule, budget, and quality of the deliverables through development of a Risk Management Plan. The Risk Management Plan must therefore include, but not be limited to, defining the probability and impact of risks, and the methodology of managing same, including, responsible parties, at both project and program levels.

The PMOR must demonstrate an understanding of the importance of defining and structuring key activities in a manner that will reduce risk at each phase of a project, and implementing a plan based on program priorities, sequencing, and other constraints. The plan should incorporate industry best practices, such as effective design, quality, cost, time, and construction management, in order to deliver best-in-class projects to the owner, and should also address a comprehensive insurance and bonding compliance program will be required for opportunities to assign and track risk, to third parties, where feasible. In addition, the PMOR must be prepared to conduct regular Risk Register updates and Risk Workshops involving all stakeholders to provide insight into changes in qualitative and quantitative risks. The following subsections detail the District's expectations of the PMOR related to key project management activities.

6.4.2 Project Charter

As part of the definition and design phases of each project, the PMOR will be required to engage both OCP and the School, where applicable, to develop a clear set of objectives that address all stakeholder requirements within the resource and time constraints. The charter will serve as a baseline upon which to measure the overall effectiveness of the PM's service, the quality benchmark for the project, and establish clear expectations for all stakeholders.

Prior to the commencement of a project, the PMOR is expected to engage in the development of a Project Charter that clearly outlines the high-level information that is necessary for a well-defined project start and project boundaries, and clearly defines key elements of the project, including, but not limited to, purpose, measurable project objectives, project description, risks, milestone schedule, budget, stakeholders, and approval requirements. While this process is typically conducted at the project level, the data gathered should be analyzed in consultation with the District's Cost and Program Controls Manager to formulate a high-level assessment of the program.

It is therefore required that the PMOR implement a high-quality project charter process that accompanies every project and is sequenced at the optimum milestone in the project lifecycle, and that the process allows for appropriate stakeholder validation. The Project Charter shall also be used as a reference tool during the progress of a project to keep stakeholder expectations aligned from beginning to end.

6.4.3 Design Management

The PMOR shall provide management of the design process from project definition through completion of preconstruction. The PMOR team shall be fully capable of developing design requirements for projects, as assigned. Design requirements shall provide a clear scope of work upon which design professionals can propose and develop projects. The PMOR shall ensure that the design professional or other design entity produces a design that considers obtaining the best value for the established construction budget, while meeting the standards, criteria, and Educational Specifications established by the District, and complying with the Florida Building Code, State Requirements for Educational Facilities, and all other applicable codes and regulations for design and construction within the south Florida area. The PMOR shall ensure those responsible for design reviews are familiar with the same.

As part of effective design management, the PMOR must develop and implement a structured process by which the project management team diligently manages the progress of the design phase, from scope validation through obtaining a Letter of recommendation for Permit from the District's Building Department. The entire process, and required deliverables are well defined within each consultant's Professional Services Agreement, with which the PMOR and individual PM must be knowledgeable.

It is imperative that the PMOR recognizes that for the success of a project and the overall program that the design phase, including the required submittals by the design professional, reviews by the Building Department and other regulatory agencies, and constructability reviews by OCP/others be carefully monitored and actively managed to be

able to progress projects in accordance with approved schedules and projected budgets. The PM must also ensure that the submittals are of a high quality and include all required documents, such as complete drawings, project manuals and specifications, in accordance with the phase of the submittal.

In addition, it is essential that the PM hold regular project meetings with the design professionals and construction manager, if applicable, to ensure that issues are addressed and any changes to design documents be implemented promptly. The meetings shall be conducted in a collaborative manner to identify and address project concerns or conflicts that may arise from code and constructability reviews, and to further mitigate the occurrence of foreseeable changes during bidding, and/or construction.

Although the District has made a commitment to executing the approved scope of work for all projects rather than deleting or reducing scope, the PM shall establish a clear understanding of the cost drivers and other risks of the project prior to the development of construction documents. The PMOR shall implement the District's robust process to control changes during the design to avoid unnecessary increases in project budgets. These constraints, as noted above in Section 6.4.3.1 will help mitigate the occurrence of foreseeable changes during construction.

While the District does not currently require design documents to be developed using Building Information Modeling (BIM), the PMOR is expected to make recommendations to incorporate BIM to further enhance the design process. The PMOR shall develop and execute on a plan to expand and improve the use of BIM which can be used to resolve design issues and potential conflicts before they reach the field. By incorporating BIM tools into the design process, the project team can quickly resolve issues and conflicts between design and constructability without having to participate in on-site meetings. Therefore, with the advantages BIM can offer, the project and overall program can benefit through enhanced efficiency and reduced risk.

6.4.4 Project Controls

The District is relying upon the PMOR to provide accurate, current information regarding the status and progress of assigned projects that comply with the procedures and control systems of the District. The importance of control systems with respect to design and construction is well understood. Their significance for program and project planning is often less clear. It is of primary concern to the District, therefore, that the PMOR works closely with OCP, or its agent, to ensure that planning processes are appropriately informed by the District's experience, in addition to the industry expertise of the PM. It is equally important to the District that the PMOR understands that the budgets and schedules produced in the planning process become the parameters of the control systems for the balance of each project's life. The District expects the PMOR to effectively participate in the controls systems to facilitate the thorough, effective movement of each project through each step of the program, and to do so in a manner that causes all stakeholders, whether administrative District employees, school based staff, contractors or consultants, to buy in and commit to the use of the controls systems and the program they represent. The following items in this section are not exhaustive and respondents are encouraged to build upon the remainder of this section with their expertise and knowledge.

6.4.4.1 Document Control

E-Builder shall be the central repository for all project information such as contractual agreements, correspondence, design submittals, Requests for Information and associated responses, invoices and pay applications, change orders, and project documentation. In addition, the PMOR will be able to execute many of the day-to-day processes and procedures through embedded workflows that support core functions such as Schedule and Resource Management, Cost Control, Construction Administration, Change Management, Funding Management, and Claims Mitigation.

6.4.4.2 Schedule Controls

The PMOR is responsible for developing and maintaining project schedules. The schedule will show each of the District's projects assigned to the PMOR and provide milestones for each major phase of a project, as well as the steps between milestones--including at a minimum: planning, design, procurement, construction, and closeout. Each schedule must show the current status of each project, the phases that have been completed, and the target dates for the phases yet to be commenced. The PMOR will ensure that the schedule

proposed minimizes any impact on the instructional operations of each school, through a collaborative process of engaging all school-based stakeholders at the initiation of the planning process. The PMOR will provide a report which states the assumptions, risks and reasons for the project milestones presented in project schedules as part of establishing an approved schedule.

The District understands that building a schedule is a critical element in the planning of a project. Accordingly, the District expects each initial project schedule, to include the planning and refining of the pertinent objectives and milestones among the items to be tracked. The District, or its agent, will work collaboratively with the PMOR to integrate and revise project level schedules into a single program schedule, which minimizes risk, maximizes resources and still meets projects-based objectives. Although some planning or assessment efforts may not move forward as a funded project, it is still imperative to schedule and track these preproject/pre-construction efforts with planning schedules. A program level schedule will be established prior to establishing project level schedules to ensure that project schedules are properly coordinated into the master program plan.

The PMOR will, accordingly, prepare project schedules, which commence with a fully approved project charter (approved by all project stakeholders and the Facilities & Construction Management Team) and track activities and milestones through the completion and turnover of the project to the school and the PPO staff. The PMOR will provide timely project updates indicating the progress, or lack thereof, of each project on a regular basis determined by OCP. The District expects the PMOR to provide OCP with the promptest notice possible of any slippage or delays on a project. The notice will, to the extent possible, explain the delay. If the reasons require further inquiry by the PM, the notice will indicate as much and will provide a timeframe for the provision of the explanation. The PMOR will also provide OCP with an action plan for addressing the slippage and recovery plan within a reasonable period subsequent to the determination of the cause of the delay.

6.4.4.3 Budgeting and Cost Control Process

The PMOR shall be responsible for the timely update of expenditures and forecasted costs based upon the current progress and or challenges of each project. While the PMOR tracks costs and maintains budgets at a project level, the CPCM will be aggregating all projects to produce a monthly Program Cash Flow model to assist Capital Budget with forecasting expenditures. The District may request that a planning or project definition service be provided for a prospective project and may provide a limited amount of funding for this purpose. Whether the District provides a planning budget or an overall budget amount for a project, the PMOR will work with the OCP to:

- 1. allocate the amount provided between the various components of the project
- 2. provide timely updates on expenditures and forecasts during planning, design, and construction; and
- implement measures that will provide prompt notice to the District with respect to circumstances that might require changes to a project's budget.

The PMOR is expected to utilize each project charter as a means for establishing clear expectations and controls on the sources and uses of the project budget. The final project cost, as compared to the project charter budget, is a key metric upon which the District will inform the evaluation of the effectiveness of the PM's project management services.

The relationship between the project's construction budget and expenditures during construction is of concern to the District. The District expects the PMOR to track payments to the general contractor in conjuncture with the progress of the Work and to provide prompt notice to OCP or its agent when a payment request exceeds progress.

6.4.4.4 Change Management

The District sees the change management process as a critical and project long effort to control costs and quality. The PMOR is expected to deploy the District's program-wide change management process. The District recognizes that changes and discovered conditions are a reality in renovations of the District's portfolio

but also recognizes that appropriate due diligence prior to beginning design, and especially before construction, are the most effective means to mitigate budget overruns in the Capital Program while also ensuring that completed projects meet their Charter's standard of quality. The District expects the PMOR to closely monitor and report on the impact of changes to the project's scope and schedule concerning the project's budget. This includes, beyond designer or contractor changes, the additional costs incurred by the District and PMOR for the management of the project, as in the event of negligent or dilatory performance by the general contractor, such costs may be charged against that general contractor.

6.4.4.5 Construction Oversight

The PMOR's project managers (PM) will be responsible for the day-to-day management of the District's projects during their respective construction phases. The project management team shall be comprised of personnel who are experienced in construction management of complex projects that may occur while existing operations continue; can make decisions promptly to ensure schedule compliance; can effectively manage cost of work and project budgets; and will adhere to proper record-keeping and maintenance of project documentation and files. The project manager and team members in each of their roles must be capable of ensuring that consultants and contractors are held to the terms and conditions of their agreements with the owner, and that they comply with the requirements of the review and inspection processes in an effort to execute the work efficiently.

It is essential that the PMOR establish a structure whereby the project management team works collaboratively with the Authorities Having Jurisdiction (AHJ) to be able avoid unnecessary delays to project schedules, and to be able to meet major milestones of Substantial Completion and Final Completion. As part of building quality into each construction project from the start, the District is interested in construction management methodologies and commissioning strategies, which eliminate protracted project closeouts and reliance on acceptance by the AHJ and punch list process alone.

The District expects the PMOR to provide a project construction schedule developed and maintained by the General Contractor (GC) or Construction Manager (CM). The District also expects a schedule to be furnished by the PMOR for each project linking the milestones of the GC's or CM's schedule. Additionally, the PM is expected to attend all meetings--including but not limited to, jobsite meetings, negotiations, change order review meetings, and Owner/PMOR meetings. Also, the PM is required to maintain logs/documentation of, including, but not limited to, project budget costs, project meeting minutes, shop drawing submittals, Building Department inspections, contractor pay applications and payments, Direct Owner Purchases (DOP), requests for information, responses to requests for information, field orders, change orders, quality deficiency reports, safety incident reports, claims, EEO compliance, M/WBE compliance, and testing. As such, the PMOR shall both track compliance and implement project assignments which ensure all requirements are met.

The PMOR shall oversee and manage the District's Direct Owner Purchase (DOP) program, at the assigned project level, by complying with the existing SOP, and with emphasis on enforcing the project contract requirements. The PMOR shall also work with local vendors, suppliers, and contractors to encourage participation, and actively review and process the DOP documentation on specific projects.

The PMOR shall consider safety in all phases of the project to ensure the health and safety of the students, staff and the public, at all times. During the construction phase, the PMOR shall work with the contractor to develop a comprehensive job site safety plan, and perform a thorough review of all job site safety plans to ensure compliance with the Districts' standards and specifications, OSHA requirements, and building code safety requirements. PM and team members shall make periodic visits to all active job sites to ensure contractor compliance with the approved job site safety plan.

6.5 Performance Evaluation

6.5.1 Program Evaluation

The PMOR shall, on a monthly basis, provide an executive summary of the performance of the program, while collaborating with the Cost and Program Controls team to ensure no duplication of efforts. The District has identified Standard Operating Procedures (SOPs) and categories of metrics, which respondents shall support, evaluate against benchmarks, and build upon based upon their professional expertise/industry best practices.

The monthly executive summary shall include the program and project performance measures listed in 6.7.9 of this RFQ. Project-level performance measures will be summarized at the program level, with emphasis placed on specific projects significantly contributing to overall performance outcomes.

6.5.2 PMOR Evaluation

The PMOR's performance shall be evaluated on a quarterly basis using a combination of objective metrics and evaluation criteria. OCP management will assess the PMOR's performance using instrumentation designed to evaluate performance in two major categories—Deliverables and Attributes. The evaluation shall be for the purpose of reviewing and scoring leadership and management competencies, relationships with stakeholders, and compliance with the technical requirements of the agreement. The specific criteria to be assessed are indicated on the sample evaluation form included as Attachment J.

The program and project-level metrics used to evaluate program performance (see 6.5.1 of this RFQ) and listed in 6.7.9 of this RFQ, along with other metrics specifically called out in this document, will be the basis for the objective contractual metrics to be used in evaluating the PMOR's performance.

Upon completion of the evaluation by the Executive Director, Capital Programs and appropriate OCP management, the results shall be discussed with the PMOR Lead and management team in a timely manner and any changes or corrective actions be implemented promptly.

6.6 Performance-Based Fees

The District is relying on the PMOR's ability to deliver projects on time and within budget using an experienced team's proven approach and methodology. To help ensure cost-effective delivery of projects throughout the term of the contract and during any extensions, a portion of the PMOR's fees will be performance-based.

The portion of the PMOR fees placed at risk is to be based on a cost-plus fee (profit) structure. The portion of the PMOR fees placed at risk would therefore be the agreed upon percentage of program costs the District agrees to pay the PMOR in addition to negotiated PMOR labor rates. Beginning with the first quarterly evaluation described in 6.5.2, the PMOR's fees would be subject to being discounted at an agreed upon rate.

The portion of the fee placed at risk (discount rate) would be determined during negotiations and contingent upon the PMOR's performance of the objective contractual metrics identified in 6.5.2, PMOR Evaluation. The relative weight of each objective metric used to measure performance will be determined and agreed upon during negotiations.

6.7 Scope of Services: Related Deliverables

The following is a general summary of PMOR Deliverables and is not intended to be an all-inclusive list of the requirements for the Agreement, between the Owner and the PMOR.

6.7.1 Schedule Controls per (Program & Project) - Monthly

- 1. Planning, Design, Procurement, Construction & Close-out
 - A. Develop Baseline Schedule (Project/Program)
 - B. Maintain Monthly Schedule Updates (Project/Program)
 - C. Variance analysis (Schedule / Budget) slippage

6.7.2 Project Charter per Project – At Project Inception

- 1. Project Charter (Stakeholder & Sponsor approvals) Per Project
 - A. Assumptions
 - B. Risks & reasons for Project milestones

6.7.3 Budgeting & Cost Control Process per Project

- Project Funds Allocation (PFA Project Budget set at project inception by CPCM; PMOR works closely with CPCM to make changes at all project milestones)
- 2. Evaluation of Pay Req. (Consultants & Contractor) (Monthly)
- 3. Prompt Notice of Budget over-run (immediately upon identification)
- 4. Project Charter as Baseline (at project inception)
- 5. **Charter** Metric for Program Evaluation (at project inception)
- 6. Earned Value Project Management (Monthly)

6.7.4 Change Management per Program/Project -Monthly

 Report & Monitor impact of Changes (Integration Management) (Quality, Scope, Schedule & Budget)

6.7.5 Knowledge Management/Continuous Improvement at Program & Project Level - Quarterly

- Collect and codify processes and practices as a function of continuous improvement efforts; review quarterly and incorporate current industry best practices and lessons learned into SOPs
 - A. Industry Best practices
 - B. Lessons learned (Log)

6.7.6 Program/Project Management Methodology Plan – Initiation Phase

- Methodology Plan
 - A. Conception /Planning Management Plan
 - B. Definition / Design Management Plan
 - C. Procurement Management Plan
 - D. Execution / Construction Management Plan
 - E. Risk Management Plan
 - F. Close-out Management Plan

6.7.7 Staffing / Responsibility Matrix - Initiation Phase and Quarterly

- 1. Responsible, Accountable, Consult, Inform (R. A. C. I.) Matrix
 - A. During the initiation phase, provide a 12-month staffing plan that correlates with the projected status of individual projects and the overall program over the 12-month period.
 - B. On a quarterly basis, provide a 12-month staffing plan that evaluates each team according to the projected status of individual projects and the overall program over the 12-month period.

6.7.8 Procurement Management – Initiation Phase and Major Milestones

- 1. Vendor evaluation procedure
 - A. Develop procedure at initiation phase; ensure evaluations are completed at major milestones
- 2. Procurement delivery method matrix
 - A. Develop matrix at initiation phase; review and modify as necessary

6.7.9 Performance Evaluation at Program & Project Level – Monthly

These project and program measures of performance will both be included in the monthly executive summary of program performance completed by the PMOR. Project-level performance measures will be summarized at the program level, with emphasis placed on specific projects significantly contributing to overall performance outcomes.

RFQ FY20-192 Program Manager - Owner's Representative Services Version Date: 04/14/2020

- Monthly Executive Summary of Program performance
 A. Schedule adherence

 - B. Budget vs Cost

 - C. RFI Rates (Logs)D. Change Order Rates
 - E. Vendor performance at Milestones
 - F. Stakeholder satisfaction
 - G. Project Quality Adherence
- Monthly Executive Summary of Project Specific Performance
 A. Contractor's Progress Payments/ Monthly

 - B. Responses to RFI's
 - C. Field & Change Order root causes
 - D. Quality deficiency & Building Dept. inspection reports
 - E. Safety incidentsF. Claims

 - G. (EDDC) compliance
 H. M/WBE compliance
 I. Material Testing Results

END OF RFQ

RFQ FY20-192 Program Manager - Owner's Representative Services Version Date: 04/14/2020

The School Board of Broward County, Florida Economic Development & Diversity Compliance 7720 W. Oakland Park Blvd., Suite 323 Sunrise, Florida 33351 (754) 321-0505

Document 00467: Supplier Diversity Outreach Program Guidelines

General:

The Economic Development & Diversity Compliance's Supplier Diversity Outreach Program is designed to help small, minority, and women business enterprises (S/M/WBE) participate in school district procurement and contract activities. The purpose of the program is to spur economic development and support S/M/WBEs to successfully expand in the tricounty marketplace.

SBBC has adopted School Board Policy 3330 - Supplier Diversity Outreach Program. The provisions of the Policy and the Standard Operating Procedures shall apply to all competitive solicitations for construction, professional services, commodities, and other contractual services, and any resulting contract documents including change orders, and amendments.

Failure to comply with the Supplier Diversity Outreach Program requirements found in the solicitation or to submit any of the information required herein shall result in the bidder being found nonresponsive to the E/S/M/WBE Program requirements.

Information:

School Board Policy 3330 - Supplier Diversity Outreach Program (SDOP) and the SDOP Standard Operating Procedures established pursuant to that Policy serve the school district's compelling interest to remedy the various ongoing effects of marketplace discrimination against S/M/WBEs that are ready, willing, and able to sell goods and services to SBBC. The SBBC encourages each awardee to make every reasonable effort to include S/M/WBE participation on any contract award under the Solicitation.

"Notice to Proceed" ("NTP") and "Authorization to Proceed" ("ATP") are used interchangeably for the purposes of this document and other SDOP specific forms and references.

"Subcontractor" and "Subconsultant" are used interchangeably for the purposes of this document and other SDOP forms and documents.

"Firm", "Contractor" and "Consultant" are used interchangeably for the purposes of this document and other SDOP specific forms and references.

Emerging/Small/Minority/Women Business Enterprise (S/M/WBE) Certifications:

At the time of submission of a response to a solicitation, the contractor shall identify all E/S/M/WBE firms to be utilized or provide a completed signed Bidder/Proposal Contractor Assurance Statement – Document 00471. All E/S/MWBE Prime who is self-performing must complete Document 00470 – S/M/WBE Statement of Intent listing percentage of Self Performance.

In response to Construction Bids, all E/S/M/WBE Prime that is self-performing, must ensure the percentage of work listed in Document – 470 S/M/WBE Statement on Intent match the Percentage of Work listed on Document 00425Certificate of Intent. A letter or certificate confirming the EDDC Department's approval of E/S/M/WBE certification status must be submitted on behalf of each S/M/WBE firm as part of the bid submission.

Sunrise, Florida 33351

(754) 321-0505

Document 00467: Supplier Diversity Outreach Program Guidelines

In response to an RFQ: Any participation by firms not certified by SBBC at the time of Qualifications Statement will not count in the RFQ evaluation process in the Contractor S/M/WBE Participation Category. However, firms that are certified by SBBC after the Qualifications Statement's tentative award will count towards the Contractor's S/M/WBE project goal attainment. The Contractor shall contact SDOP to provide the updated information.

For information on E/S/M/WBE Certification, or to obtain information on locating certified S/M/WBEs, contact SBBC's Supplier Diversity Outreach Programs at 754-321-0505 or http://www.browardschools.com/sdop.

A "Minority Business Enterprise (MBE)" is defined as any legal entity, except a joint venture, that is organized to engage in for-profit transactions, which is certified as being at least fifty-one percent (51%) owned, managed, and controlled by one (1) or more minority group members, and that is ready, willing, and able to sell goods or services that are purchased by SBBC. The enterprise annual revenues and number of employees shall be no greater than fifty percent (50%) of the small business size standards for its industry as established by the U.S. Small Business Administration and meet the significant business presence requirements in accordance with School Board Policy 3330.

A "Women Business Enterprises (WBE)" is defined as any legal entity, except a joint venture, that is organized to engage in for-profit transactions, at least fifty-one percent (51%) owned, managed, and controlled by one (1) or more non-minority, women individuals that are lawfully residing in, or are citizens of, the United States or its territories, that is ready, willing, and able to sell goods or services that are purchased by SBBC and that meets the significant business presence requirements in accordance with School Board Policy 3330. The enterprise annual revenues and number of employees shall be no greater than fifty percent (50%) of the small business size standards for its industry as established by the U.S. Small Business Administration; and meets the significant business presence requirements in accordance with School Board Policy 3330.

A "Minority/Women Business Enterprise (M/WBE)" is defined as a firm that is certified as either a minority Women business enterprise and which is at least fifty-one percent (51%) owned, managed, and controlled by one (1) or more minority group members and/or women, and that is ready, willing, and able to sell goods or services that are purchased by the school district.

An "Emerging M/WBE" is defined as a certified M/WBE firm whose annual revenues and number of employees are no greater than twenty-five percent (25%) of the small business size standards for its industry as established by the U.S. Small Business Administration and meets the significant business presence requirements in accordance with School Board Policy 3330.

A "Small Business Enterprise (SBE)" is defined as a corporation, partnership, sole proprietorship, or other legal entity for making a profit, which is independently owned and operated by individuals legally residing in, or that are citizens of, the United States or its territories. The enterprise annual revenues and number of employees shall be no greater than fifty percent (50%) of the small business size standards for its industry(ies) as established by the U.S. Small Business Administration and meets the significant business presence requirements in accordance with School Board Policy 3330.

An "Emerging SBE" is defined as a certified SBE corporation, partnership, sole proprietorship, or other legal entity for the purpose of making a profit, which is independently owned and operated by individuals legally residing in, or that are citizens of, the United States or its territories whose annual revenues and number of employees are no greater than twentyfive percent (25%) of the small business size standards for its industry as established by the U.S. Small Business Administration and meets the Significant Business Presence requirements in accordance with School Board Policy 3330.

Sunrise, Florida 33351

(754) 321-0505

Document 00467: Supplier Diversity Outreach Program Guidelines

Compliance:

The School District shall monitor performance and compliance in accordance with School Board Policy 3330. During the term of any subsequent contract, the awardee shall comply with the S/M/WBE Subcontractor Utilization Plan made in its response to the solicitation. Each contract/agreement containing certified firm commitment must include a requirement of periodic reporting of both certified and non-certified subcontractor participation. For construction projects, reporting is monthly through e-Builder. For Design and Pre-construction CMAR Awards, reporting should be provided at each phase being invoiced through e-Builder. Further, non-receipt of utilization reporting may result in breach of contract. The submission of reporting is further identified to be a condition for payment of the contractor.

SDOP will monitor and record non-receipt of required reports on a quarterly basis. The report will include the name of the prime, the project(s) involved, and the report(s) missing. Compliance for use of S/M/WBEs shall include tasks and proportionate dollar amounts throughout the term of the contract, including amendments and change orders. After contract execution, the awardee shall maintain the level of utilization as established in the contract's S/M/WBE Subcontractor Utilization Plan. Failure to comply with the SDOP requirements in the contract will be considered a material breach resulting in recommendation for sanctions, penalties and which could include debarment proceedings pursuant to School Board Policy.

Prime & Subcontractor Terms and Conditions

In keeping with the objective of Policy 3330 and Chapters 9 and 13 of the Standard Operating Procedures, SBBC requires the terms and conditions that are included in the Prime Contractor or Prime Consultant or Prime Vendor's (Prime) Agreement between the Prime and SBBC be extended in similar or equal capacity in the contracts between the Prime and all lower tier Subcontractors. Refer to Document 00472 - Prime/Subcontractor Contract Terms & Conditions. All Bidders must comply with the Prime/Subcontractor Terms & Conditions Document 00472.

The Subcontractors Utilization Plan shall consist of the following documentation, which must be attached to the Bid solicitation:

- 1. S/M/WBE Subcontractor Participation Schedule (Document 00475)
- 2. Statement of Intent to Perform as an S/M/WBE Subcontractor (Document 00470)
- 3. Bidder/Proposer Assurance Statement (Document 00471) (Only Applicable to CSMP and CMAR projects) **Note:** For projects utilizing a CMAR project delivery method, exact M/WBE or SBE participation during construction phase will not be known until proposals are received, subcontractors are selected, and the Guarantee Maximum Price (GMP) is developed. Please state the CM's commitment for M/WBE or SBE participation during the construction phase on the Bidder/Proposer Assurance Statement (Document 00471). The Construction Manager (CM) will be obligated to meet this commitment as a condition to gain SBBC approval of the GMP.

In response to an RFQ If percentages vary between the Participation of Schedule (Document 00475) and the M/WBE Statement of Intent (Document 00470), the percentage listed on the Statement of Intent (Document 00470) will be used to determine participation. If the M/WBE Statement of Intent (Document 00470) - is not signed by the Subcontractor, participation attributed to the listed vendor will not be included even if they are listed on Participation Schedule Form (Document 00475). If no percentage is listed, Submitting Firm will not receive points. If a percentage range is provided for an M/WBE Subcontractor on the M/WBE Statement of Intent Form (Document 00470), the higher determined percentage in the range will be used to calculate participation and therefore, effectively, represents the prime's commitment to the contract goal.

Sunrise, Florida 33351 (754) 321-0505

Document 00467: Supplier Diversity Outreach Program Guidelines

Nondiscrimination:

The respondent hereby certifies and agrees that the following information is correct:

In preparing its response on this project, the respondent has considered all proposals submitted from qualified, potential subcontractors and suppliers, and has not engaged in "discrimination" as defined in School Board Policy 3330 – Supplier Diversity Outreach Program, Section D.1; to wit: discrimination in the solicitation, selection, or commercial treatment of any subcontractor, vendor, supplier, or commercial customer on the basis of race, color, religion, ancestry or national origin, sex, age, marital status, sexual orientation or on the basis of disability or other unlawful forms of discrimination. Without limiting the foregoing, "discrimination" also includes retaliating against any person or other entity for reporting any incident of "discrimination." Without limiting any other provision of the solicitation for responses on this project, it is understood and agreed that, if this certification is false, such false certification will constitute grounds for SBBC to reject the response submitted by the respondent on this project, and terminate any contract awarded based on the response. As a condition of submitting a response to SBBC, the respondent agrees to comply with SBBC's Nondiscrimination Policy as described under its School Board Policy 3330 – Supplier Diversity Outreach Program, Section D.1.

Affirmative Procurement Initiatives (APIs) for Professional Services:

"Affirmative Procurement Initiatives (API)" – refers to various SDOP tools and solicitation incentives that are used to encourage greater prime and subcontract participation by S/M/WBE firms, including bonding assistance, evaluation preferences, subcontracting goals, and joint venture incentives. (For full descriptions of these and other SDOP tools, see, Section E of School Board Policy 3330 and the SDOP Standard Operating Procedures.)

Penalties and Sanctions:

Upon the receipt of a recommendation from the Superintendent of Schools regarding the failure of the prime or its authorized agents to comply with any portion of the SDOP policy and Section 13 of the Standard Operating Procedures, the SBBC may take actions regarding the non-complying party, including, but not limited to, suspension of contract, withholding payments due, rescission or termination of contract based upon the failure to comply with SDOP policy requirements, refusal to accept bids, responses, Qualification Statements and proposals, disqualification of a respondent, contractor, or other business entity from eligibility for providing goods or services to the SBBC for a period not to exceed two (2) years (upon SBBC's approval) and/or liquidated damages equal to the difference between the committed versus actual dollar value of S/M/WBE participation.

Any person who violates these provisions shall be subject to penalties and sanctions established by SBBC and to the maximum penalty provided by law.

Good Faith Efforts:

If the information submitted in the response demonstrates that a Contractor does not meet the S/M/WBE Subcontractor Goal, then the Contractor shall submit information supporting its good faith efforts that is sufficient to satisfy SDOP.

If a respondent is unable to comply with the API requirements imposed by SBBC under the terms of the SDOP policy, as required in the solicitation, such respondent shall complete and submit an S/M/WBE Good Faith Effort Waiver Request Form, at the time the solicitation is due. The Form shall include specified documentation which demonstrates a good faith effort to comply with the requirements as described under the selected API.

In making the required judgment, SDOP will consider some or all of the following good faith efforts. Other factors or types of efforts may be relevant in appropriate cases. In determining whether a Contractor has made good faith efforts, SDOP will evaluate the efforts that a Contractor has made and the quantity and intensity of these efforts.

Sunrise, Florida 33351 (754) 321-0505

Document 00467: Supplier Diversity Outreach Program Guidelines

SBBC has developed concise, detailed, and quantifiable good faith effort requirements for prime contractors to ensure they are making a legitimate attempt to meet the APIs. Meeting the goal, or making good faith efforts to meet the goal, is a condition of being issued a Notice to Proceed (NTP) or Authorization to Proceed (ATP).

The listing of S/M/WBE Subcontractors by a Contractor shall constitute a representation by the Contractor that such entity is qualified and available, and a commitment by the Contractor that, if it is issued a NTP or ATP, it will enter into a subcontract in the amount set forth in its submission, subject to the terms of these Guidelines.

Respondents must obtain a total of **seventy (70) or more points out of One Hundred (100)** to indicate that the Good Faith Efforts were sufficient and to be deemed compliant

The following actions outline evidence of the minimum good faith effort to meet the SDOP S/M/WBE subcontracting goal and describe the criteria to quantify good faith efforts.

Advertising (5 points)

Effort: Bidders must advertise opportunities for SBBC-certified S/M/WBEs in a minimum of three (3) separate print newspapers and post ten (10) digital media notices using various combinations of Twitter, Facebook, Trade Associations, and ethnic focused media outlets by no later than ten (10) business days once the bid is advertised. The Bidder is highly encouraged to effectively use the services of available small/minority/women community organizations; small/minority/women contractors' groups; local, State, and Federal small/minority/women business assistance offices; and other organizations as allowed on a case-by-case basis to provide assistance in the recruitment and placement of S/M/WBEs. Bidders are required to publish these opportunities in the general circulation media, minority-focused media, trade association publications, or trade-related publications, unless the SBBC waives this requirement due to time constraints. Pro forma mailings to S/M/WBEs requesting bids are not alone sufficient to satisfy good faith efforts.

Documentation: The advertisement shall include the name and location of the project, the location where plans and specifications can be viewed, the subcontractor proposal due date, and the items of work or specialties being solicited. (i.e. Newspaper Tear Sheet and copy of Social Media Post).

Outreach to Identify S/M/WBEs (15 points)

Effort: The bidder should solicit this interest as early in the solicitation process as practicable, but no later than ten (10) business days after the Pre-bid Conference to allow the S/M/WBEs to respond to the solicitation and submit a timely offer for the subcontract. The bidder should determine with certainty if the S/M/WBEs are interested by taking appropriate steps to follow up initial solicitations.

The bidder should attempt to reach out to S/M/WBEs by utilizing the SDOP approved online databases of certified firms. The scope of work required should be identified, and the contractor needs to seek out relevant companies to perform the specified work.

Documentation: The Contractor should retain dated documentation of any correspondence targeting S/M/WBE.

Sunrise, Florida 33351

(754) 321-0505

Document 00467: Supplier Diversity Outreach Program Guidelines

Pre-Bid Meeting Attendance (5 points)

Effort: Attendance at the pre-bid meeting is mandatory to comply with the good faith effort requirement in order to be awarded the 5 points. If a pre-bid meeting is not offered, five (5) points will be deducted from the seventy (70) point minimum score.

Documentation: The Contractor's name should appear on SBBC's pre-bid meeting sign-in sheet to document the company presence.

Timely Written Notification (20 points)

Effort: Selecting portions of the work to be performed by S/M/WBEs in order to increase the likelihood that the assigned goals will be achieved. This includes, where appropriate, breaking out contract work items into economically feasible units (for example, smaller tasks or quantities) to facilitate S/M/WBE participation, even when the prime bidder might otherwise prefer to perform these work items with its own forces. This may include, where possible, establishing flexible timeframes for performance and delivery schedules in a manner that encourages and facilitates S/M/WBE participation

Documentation: Written correspondence with the subcontractor's name, address, contact person, and the date of the written notice should be documented. Written notification must be dated and transmitted at least ten business days prior to the bid due date and include verification of transmission date. Such verification may include emails and copies of certified mail return receipts.

Initial Contact Follow-up (15 points)

Effort: The contractor should follow-up on initial solicitations by contacting S/M/WBE subcontractors prior to the bid opening to determine with certainty whether the subcontractors are interested in performing the specific items of work on the project. Such contact shall be within a reasonable amount of time to allow the prospective S/M/WBE subcontractor an opportunity to submit a competitive sub-bid. The content of the initial letter, email, or facsimile should be different for the follow-up correspondence to indicate the additional effort expended to secure bidders.

Documentation: The list of subcontractors that were contacted, including results of that contact, documented by a telephone log, e-mail printout, automated facsimile journal, or fax transmittal documents, is required. The record should include the S/M/WBE's name, telephone number, contacted person, dates of contact, and the outcome.

Identify Items of Work (15 points)

Effort: Bidder is to provide interested S/M/WBE's with adequate information about the plans, specifications, and requirements of the contract in a timely manner, not less than five (5) days, to provide their Sub-Contractor Proposal.

Documentation: The list utilized to define the specific items of work solicited, including the identification process for S/M/WBE firms from which such work was solicited, is required. Documents should be reviewed to determine if the work is specific to that listed in the goal-setting document for the specific work item.

Sunrise, Florida 33351

(754) 321-0505

Document 00467: Supplier Diversity Outreach Program Guidelines

Negotiate in Good Faith (15 points)

Effort: The contractor should negotiate in good faith with the S/M/WBE, and not unjustifiably reject bids, quotes, and proposals prepared by the S/M/WBE as unsatisfactory.

Documentation: Written statements of the names, addresses, and telephone numbers of subcontractors contacted by the contractor to negotiate price or services should be submitted. Dates of the negotiations and the results should be included, as well as documentation of the quotes and/or proposals received from S/M/WBEs; and evidence as to why additional Agreements could not be reached for S/M/WBEs to perform the work.

Bonding Requirements (5 points)

Effort: Waive or reduce bonding requirements for subcontractors where possible

Documentation: Copies of Respondent's commitment to waive or reduce bond requirements.

Assist in Financing, Bonding, and Insurance (5 points)

Effort: Where applicable, the contractor should advise and make efforts to assist interested S/M/WBEs in obtaining bonds, lines of credit, or insurance that SBBC requires.

Documentation: Written statements of the type of assistance offered to S/M/WBEs are required. The contractor should provide the name, contact person, and telephone number of the bonding company or financial institution offering assistance.

Acceptability of S/M/WBE Subcontractor Good Faith Efforts Submission:

- 1. If the SDOP questions the acceptability of the Firm's S/M/WBE subcontractor submissions, the Contractor shall present information to substantiate its compliance with the applicable requirements to SDOP within five (5) consecutive business days.
- 2. Not later than seven (7) calendar days after the communication with the Contractor, SDOP shall make a written recommendation. Recommendations to reject the Firm's Good Faith Efforts Submission may be submitted to the Director of Procurement & Warehousing Services for review and acceptance. The rejection will result in the Firm's response being deemed nonresponsive.
 - a. For RFQs, in the event, the highest ranked firm is deemed nonresponsive by PWS, District Staff will evaluate and negotiate with the next responsive /responsible firm/
 - **b.** For Construction Hard Bids, in the event the or lowest bidder is deemed nonresponsive, district staff will evaluate the next responsive / responsible bidder

Sunrise, Florida 33351

(754) 321-0505

Document 00467: Supplier Diversity Outreach Program Guidelines

Calculation of S/M/WBE Subcontractor Participation toward Contract Goal:

For projects utilizing a CMAR project delivery method, exact M/WBE or SBE participation during construction phase will not be known until proposals are received, subcontractors are selected, and the Guarantee Maximum Price (GMP) is developed. Please state the CM's commitment for M/WBE or SBE participation during the construction phase. The Construction Manager (CM) will be obligated to meet this commitment as a condition to gain SBBC approval of the GMP.

S/M/WBE subcontractor participation shall be calculated as follows:

- 1. Once the SBBC has determined that a firm is an eligible S/M/WBE subcontractor, the total dollar value of the contract awarded to the S/M/WBE subcontractor is counted toward the goal.
- 2. The SBBC shall count toward the goal only expenditures to S/M/WBE Subcontractors that perform a commercially useful contractual function in the work. An S/M/WBE Subcontractor is considered to perform a useful function when it is responsible for execution of a distinct element of the work of a contract and carrying out the responsibilities by actually performing and supervising the work involved.
- Consistent with normal industry practices, an S/M/WBE subcontractor may enter into subcontracts. If such an entity
 subcontracts a significantly greater portion of the work of the contract than would be expected on the basis of normal
 industry practices to non-S/M/WBE subcontractors, the entity shall not be considered as an S/M/WBE subcontractor.

Issuance of an NTP or ATP:

Provided the Firm submitted the completed forms and information as required by these Guidelines and submitted information to satisfy SDOP that it has met the API or has made good faith efforts to meet the API, an NTP or ATP may be issued.

Requirements and Procedures Subsequent to the Issuance of a NTP or an ATP:

Subcontracts:

After issuance of an NTP or an ATP, the Firm shall submit, executed subcontract agreements corresponding in all respects to its S/M/WBE Subcontractor Utilization Plan (including authorized substitutions).

If there is a change to the S/M/WBE Subcontractor Utilization Plan, the prime contractor/prime consultant shall submit a request for change to EDDC. Only EDDC approved Termination/Deletions/Additions & Substitution of S/M/WBE Subcontractors will be counted towards the committed participation goal.

<u>Termination/Deletions/Additions & Substitution of S/M/WBE Subcontractors:</u>

Changes are not accepted pre-award. Post-award, the School Board Policy 3330 and Section 9 of the Standard Operating Procedures requires that a prime may not deviate from the accepted certified subcontracting team on the project without the written approval of EDDC and Program Manager or designee. The reason for review of such requests is to prevent bid shopping and prevent undue or arbitrary removal of Broward Schools certified small businesses.

- 1. Termination of a S/M/WBE Certified Firm: A certified firm may not be removed from a subcontracting team without the prior written consent of EDDC and the Program Manager or designee. The acceptance of the request shall be in writing to the requesting prime contractor/prime consultant.
 - a. If a request to delete or substitute a certified firm is granted by EDDC, the prime must replace the removed firm with another M/WBE of at least equal contracting value to the overall project. Failure to do so will result in a notice to cure from EDDC and potentially a notice of violation if the matter is not resolved. The prime

Sunrise, Florida 33351

(754) 321-0505

Document 00467: Supplier Diversity Outreach Program Guidelines

contractor/prime consultant is responsible for utilizing good faith efforts to obtain the level of certified participation agreed upon at project inception.

- 2. Additions, Deletions, & Substitutions: Changes are not accepted pre-award. Post-award, the School Board Policy 3330 requires that a prime may not deviate from the accepted certified subcontracting team on the project without the written approval of the Program Manager or designee. The reason for review of such requests is to prevent undue or arbitrary removal of S/M/WBE certified small businesses.
- 3. Reasons to request a substitution of S/M/WBEs in a Contractor's S/M/WBE Subcontractor Utilization Plan include, but are not limited to the following:
 - b. Subcontractor is no longer qualified, Subcontractor has unreasonably refused to execute the subcontract agreement without cause, Subcontractor is no longer able to perform the work, Subcontractor materially breaches its subcontract agreement with the contractor.
- 4. If after a "reasonable" good faith effort measures, the Prime is unable to find a substitute S/M/WBE, a post-award waiver may be requested, which documents the reasons for the contractor's inability to meet the goal requirement. In the event the contractor is found not to have performed good faith efforts to find a suitable substitute for the initial S/M/WBE Subcontractor Utilization Plan (Doc.00485) the contract may, in the District's sole discretion, be terminated for material breach.
 - c. The prime shall make every reasonable effort to propose and enter into an alternative subcontract agreement or agreements for the same work to be performed by other certified S/M/WBE subcontractor(s) for a contract price for such work equal to or less than the price originally scheduled for such work (less all amounts previously paid thereof).
- 5. Compliance with Contract Goal:

The Contractor shall be deemed to be in continued compliance with the established goal for the S/M/WBE Subcontractor Utilization Plan if the Contractor meets or exceeds the goal established in the S/M/WBE Subcontractor Utilization Plan or if the Contractor's good faith efforts are approved.

6. Monthly Monitoring Report:

The Contractor shall submit a S/M/WBE Monthly Subcontractor Utilization Report as required in section XXX of these guidelines) when the Contractor's agreement with SBBC includes an S/M/WBE Subcontractor Utilization Plan. The S/M/WBE Monthly Subcontractor Utilization Report can be obtained from SDOP.

Should the Contractor fail to comply with the provisions required for monthly reporting requirements, it shall be considered in default of its contract. (See Sanctions and Penalties)

Sunrise, Florida 33351

(754) 321-0505

Document 00467: Supplier Diversity Outreach Program Guidelines

Non-Compliance with Contract Goal:

1. Should the Contractor fail to comply with the provisions required by its S/M/WBE Subcontractor Utilization Plan, this shall be considered a material breach and therefore the Contractor shall be considered in default of its Contract.

2. Compliance Requests for Information; Non-Receipt of Compliance Information

- a. The EDDC may request information to assist with producing compliance reviews/reports. Requests for information must be properly titled, sufficiently detailed, printed on letterhead or via internal email (including staff name and contact information), and must have a date for response within five (5) business days or less as determined by compliance staff.
- b. After efforts to ascertain the information have been exhausted and the information requested has not been provided, the compliance process will proceed with the information on hand. Mention will be made of any missing information, in official correspondence, and the status of the request(s).



SMALL/MINORITY/WOMEN BUSINESS ENTERPRISE SUBCONTRACTOR PARTICIPATION SCHEDULE

DATE

SOLICITATION INFORMATION										
Contract #:		e:								
Project Name:										
Project Location:										
Bidder/Proposer:										
Address:										
Contact Person:		Email Address:		Ph	one #:					
		ORGANIZATION S	TATUS							
Business Association	Business Name	Business Address	Business Phone #	Type of Work to be Performed	% of Work	\$ Amount				
Prime Bidder/ Proposer					%	\$				
Non S/M/WBE Subcontractor					%	\$				
S/M/WBE Subcontractor					%	\$				
S/M/WBE Subcontractor					%	\$				
S/M/WBE Subcontractor					%	\$				
S/M/WBE Subcontractor					%	\$				
		TOTAL PAR	RTICIPATION % & DOLL	AR AMOUNT:	100%	\$				
	ACT AMOUNT:	\$								
		PIDDED/DDADACED CI	CNATURE							
BIDDER/PROPOSER SIGNATURE The listing of S/M/WBE(s) shall constitute a representation by the bidder/proposer to the SBBC that the bidder/proposer believes such S/M/WBE(s) to be technically and financially qualified and available to perform the work described. Bidders/Proposers are advised that the information contained herein may be verified. I certify that all information contained in this form is true and accurate to the best of my knowledge.										
Bidder/Pr	oposer Signature	Name &	Name & Title (Print) Date							



STATEMENT OF INTENT TO PERFORM AS AN S/M/WBE SUBCONTRACTOR

SOLICITATION #:	
CONTRACT #:	

A signed Statement of Intent to Perform as a Small/Minority/Women Business Enterprise (S/M/WBE) Subcontractor form must be completed by the owner or authorized principal of each S/M/WBE firm listed in the S/M/WBE Subcontractor Participation Schedule.

	STATEMENT OF INTENT									
The undersi	gned is certified as an S/M/WBE vendor by The School Board of Broward County, F	lorida's (SBBC) Supplier	Diversity							
Outreach Pr	=	, , , , ,	•							
	(Name of S/M/WBE Subcontract	or) agrees to perform v	vork on the							
above contr	ract as a (check one):	or, agrees to perform v	vork on the							
		Cornoration								
	·	Corporation								
	BE subcontractor will enter into a formal agreement with									
(Name of Bidder/Proposer) conditioned upon the Bidder/Proposer executing a contract with SBBC.										
DESCRIPTION OF WORK & VALUE										
Please provide the details and value of the work to be performed:										
Item No.	Type of Work	Agreed Upon Price	% of Work							
1		\$	%							
2		\$	%							
3		\$	%							
	TOTAL VALUE OF WORK	\$	%							
		_								
	S/M/WBE SUBCONTRACTOR SIGNATURE									
	(Signature) S/M/WBE Subcontractor	Title								
	(Print) Name of S/M/WBE Subcontractor	Date								

Procurement & Warehousing Services • 7720 West Oakland Park Blvd., Suite 323, Sunrise, FL 33351 • (754) 321-0505 www.BrowardSchools.com/SDOP

S/M/WBE MONTHLY SUBCONTRACTOR UTILIZATION REPORT

SECTION I - GENERAL INFORMATION										
Project Name:					Contract Number and Work Order Number (if applicable):					
Report #:		Reporting Period:	Reporting Period:			ct Goal:	Contract Compl	etion Date:		
		to					-			
Prime Contractor	· Name:				Project Manager	(PM) Name:				
Prime Contractor	Street Address:									
Prime Contractor	· Phone #:	Prime Contractor E	mail Addres	s:	PM Phone #: PM Email Address:					
		SEC	TION II - UT	ILIZATION INFORM	ATION					
Prime Contracto	or must list ALL C	ertified and non-certified su	ubcontracto	rs that will be utilize	ed for the entire c	ontract perio	od.			
For assistance in	completing this	form, please call the Suppl		Outreach Program	at (754) 321-050					
ROLE	FEDERAL IDENTIFICATION NUMBER	BUSINESS NAME	S/M/WBE CERTIFIED BY SBBC (Y/N)	DESCRIPTION OF WORK	TOTAL PROJECT AMOUNT	AMOUNT PA DURING REPORTING PERIOD	INVOICE #	TOTAL PAID TO DATE		
PRIME CONTRACTOR					\$	\$		\$		
SUBCONTRACTOR					\$	\$		\$		
SUBCONTRACTOR					\$	\$		\$		
SUBCONTRACTOR					\$	\$		\$		
SUBCONTRACTOR					\$	\$		\$		
SUBCONTRACTOR					\$	\$		\$		
SUBCONTRACTOR					\$	\$		\$		
SUBCONTRACTOR					\$	\$		\$		
SUBCONTRACTOR					\$	\$		\$		
SUBCONTRACTOR					\$	\$		\$		
		Total Paid to Dat	e for All S/M	/WBE Subcontractors	s \$	\$		\$		
				N III - AFFIDAVIT						
I hereby affirm th	at the informatio	n on this form is true and co	mplete to th	e best of my knowle	edge.					
Prime Contractor Authorized Personnel (Signature) Prime Contractor Authorized Personnel (Print) Title Date										

Procurement & Warehousing Services • 7720 West Oakland Park Blvd., Sunrise, FL 33351 • (754) 321-0505 • www.BrowardSchools.com/SDOP

FORM INSTRUCTIONS: S/M/WBE MONTHLY SUBCONTRACTOR UTILIZATION REPORT

SECTION I: GENERAL INFORMATION

- **1. Project Name**: Enter the entire name of the Project.
- Contract Number (work order): Enter the District contract number and work order number
- 3. Report Number: Enter the S/M/WBE Monthly Subcontractor Utilization Report number.
- **4.** Reports must be in a numerical series (i.e., 1, 2, 3).
- 5. Reporting Period: Enter the beginning and end dates for which this report covers (i.e., 10/01/2016: 9/30/2018)
- **6. S/M/WBE Contract Goal**: Enter the S/M/WBE Contract Goal on entire contract.
- 7. Contract Completion Date: Enter the expiration date of the contract, (not work order).
- 8. Prime Contractor Name: Enter the complete legal business name of the Prime Contractor.
- 9. Prime Contractor Street Address: Enter the mailing address of the Prime Contractor.
- **10. Prime Contractor Phone Number**: Enter the telephone number of the Prime Contractor.
- 11. Prime Contractor Email Address: Enter the email address of the Prime Contractor.
- 12. Project Manager (PM) Name: Enter the name of the Project Manager for the Prime Contractor on the project.
- **13.** PM Telephone Number: Enter the direct telephone number of the Prime Contractor's Project Manager.
- **14.** PM Email Address: Enter the email address of the Prime Contractor's Project Manager.

SECTION II: UTILIZATION INFORMATION

- 15. Federal Identification Number: Enter the Federal Identification Number of the S/M/WBE Subcontractor(s)
- 16. Business Name: Enter the complete legal business name of the S/M/WBE Subcontractor(s)
- 17. S/M/WBE Certified by BCPS (Yes/No): Enter "yes" or "no" to indicate if the subcontractor is S/M/WBE Certified by BCPS
- **18. Description of Work**: Enter the type of work being performed by the S/M/WBE Subcontractors(s) (i.e., electrical services).
- 19. Total Project Amount: Enter the dollar amount allocated to the S/M/WBE Subcontractors(s) for the entire project (i.e., amount in the subcontract agreement).
- 20. Amount Paid During Reporting Period: Enter the total amount paid to the S/M/WBE Subcontractor(s) during the reporting period.
- 21. Invoice Number: Enter the S/M/WBE Subcontractor's invoice number related to the payment reported this period.
- 22. Total Paid (to Each Subcontractor) to Date: Enter the total amount paid to the S/M/WBE Subcontractor(s) to date.
- 23. Total Paid to All Subcontractors to Date: Enter the total amount paid to all subcontractors during reporting period.

SECTION III: AFFIDAVIT

- **24. Affidavit**: Statement attesting to the contents of the report.
- 25. Prime Contractor Name Authorized Personnel (signature): Signature of the employee that is authorized to execute the S/M/WBE Subcontractor Utilization Report.
- 26. Prime Contractor Name Authorized Personnel (print): Printed name of the employee that is authorized to execute the S/M/WBE Subcontractor Utilization Report.
- 27. Title: Enter the title of authorized employee completing the S/M/WBE Subcontractor Utilization Report.
- 28. Date: Enter the date of submission of the S/M/WBE Subcontractor Utilization Report to the District.

Procurement & Warehousing Services • 7720 West Oakland Park Blvd., Sunrise, FL 33351 • (754) 321-0505 • www.BrowardSchools.com/SDOP

Effective July 1, 2017 Page 2 of 2



ECONOMIC DEVELOPMENT & DIVERSITY COMPLIANCE DEPARTMENT (EDDC)

Document Number 00480

S/M/W	BE PARTICIPATION GOOD FA	ITH EFFORT FORM	DATE:						
CONTACT INFORMATION									
Solicitation Title:									
Solicitation #:									
Bidder/ Submitting Firm:									
Contact Person:									
Phone #:		Email:							
	GOOD EAITH FEE	OPT DOCUMENTATION							

The purpose of this form is to demonstrate good faith efforts in meeting the required subcontracting goal. Respondents must obtain a total of seventy (70) or more points in order to pass, indicating that good faith efforts were sufficient. The good faith efforts demonstrated must be with Broward County Public Schools S/M/WBE Certified firms.

The following actions outline evidence of the minimum good faith effort to meet the SDOP S/M/WBE subcontracting goal and describe the criteria to quantify good faith efforts.

SECTION A – Advertising Effort	5 points
Please provide documentation and supporting evidence to show how the criteria was fulfilled.	
Advertise opportunities for SBBC-certified S/M/WBEs a minimum of three (3) times in print an outlets at least two (2) weeks prior to the bid opening.	d/or digital media
Publish these opportunities in the general circulation media, minority-focused media, trade as publications, or trade-related publications, unless SBBC waives this requirement due to time co	
Additional efforts, if any: (Provide information in space provided below)	
Requirements: Attach copy of advertisements (i.e. newspaper Tear Sheet) with the name and location of the i	project the location

Economic Development & Diversity Compliance Department (EDDC) · 7720 West Oakland Park Blvd., Suite 323, Sunrise, FL 33351· (754) 321-0505

where plans and specifications can be viewed, the subcontractor proposal due date, and the items of work or

specialties being solicited.

ECONOMIC DEVELOPMENT & DIVERSITY COMPLIANCE DEPARTMENT (EDDC)

Document Number 00480

S/M/WBE PARTICIPATION GOOD FAITH EFFORT FORM

SECTION B – Outreach to Identify S/M/WBE Firms	15 points
Please provide documentation and supporting evidence to show how the criteria was fulfilled.	
Identify the scope of work required Seek relevant companies to perform the specified work	
Contact S/M/WBEs by utilizing the EDDC approved online database of certified firms	
Additional efforts, if any: (Provide information in space provided below)	
Requirements: Attach dated documentation of all correspondence to target S/M/WBE Certified firms listed on www.BrowardSchools.com/sdop	
Include in dated documentation the business name, telephone number, fax number, email add work solicited to perform	lress, and type of

Please provide documentation and supporting evidence to show how the criteria was fulfilled. Attend pre-bid meeting to comply with the good faith effort requirement. Attendance is mandatory. Note: If a pre-bid meeting is not offered, five (5) points will be deducted from the seventy (70) point minimum score. Additional efforts, if any: (Provide information in space provided below) Requirements: Signature on SBBC's pre-bid meeting sign-in sheet to document the proposer's presence.

Economic Development & Diversity Compliance Department (EDDC) · 7720 West Oakland Park Blvd., Suite 323, Sunrise, FL 33351 · (754) 321-0505



ECONOMIC DEVELOPMENT & DIVERSITY COMPLIANCE DEPARTMENT (EDDC)

Document Number 00480

S/M/WBE PARTICIPATION GOOD FAITH EFFORT FORM

SECTION D - Timely Written Notification Effort

20 Points

Please provide documentation and supporting evidence to show how the criteria was fulfilled.

Solicit, in writing, relevant S/M/WBE subcontractors for bids and material quotes.

Solicit relevant S/M/WBEs, in a timely manner, to result in meeting the goal.

Note: Relevant S/M/WBEs are firms that could feasibly provide services or supplies required to complete the scope of services provided in the bid document.

Additional efforts, if any: (Provide information in space provided below)

Requirements:

Attach dated written correspondence with the subcontractor's name, address, and contact person.

Correspondence must be transmitted at least ten (10) business days prior to the bid due date.

Correspondence must include verification of transmission date. Such verification may include emails, or copies of certified mail return receipts.

SECTION E – Initial Contact Follow-up

15 Points

Please provide documentation and supporting evidence to show how the criteria was fulfilled.

Follow-up with S/M/WBEs, in writing, on initial outreach prior to the bid opening to determine with certainty their interest in performing the specific scope of work on the project.

Follow-up contact must be made in a timely manner to allow the prospective S/M/WBE subcontractor an opportunity to submit a competitive sub-bid.

Follow-up correspondence date should differ from original outreach to verify the additional effort expended to secure bidders.

Additional efforts, if any: (Provide information in space provided below)

Requirements:

List of subcontractors contacted for follow-up

Results of follow-up effort, documented by a telephone log, e-mail printout, copies of certified mail return receipts, or fax transmittal documents

Follow-up record should include the S/M/WBEs name, telephone number, person contacted, date(s) of contact, and the result.

Economic Development & Diversity Compliance Department (EDDC) · 7720 West Oakland Park Blvd., Suite 323, Sunrise, FL 33351 · (754) 321-0505

Document Number 00480

S/M/WBE PARTICIPATION GOOD FAITH EFFORT FORM

Please provide documentation and supporting evidence to show how the criteria was fulfilled. Identify specific items of the work to be performed by subcontractors. Offer S/M/WBEs portions of the work or other assistance that could reasonably be expected to produce a level of S/M/WBE participation to meet the goals Additional efforts, if any: (Provide information in space provided below) Requirements:

Include list utilized to define the specific items of work solicited, including the identification process to solicit

SECTION G - Negotiate in Good Faith

S/M/WBE firms

15 Points

Please provide documentation and supporting evidence to show how the criteria was fulfilled.

Negotiate in good faith with the S/M/WBE

Note: Proposals, bids, or quotes prepared by S/M/WBE firms may not be unjustifiably rejected.

Additional efforts, if any: (Provide information in space provided below)

List of specific items of work must meet goal-setting requirements

Requirements:

Submit written statements of the subcontractors contacted to negotiate price or services

Include name, address, and telephone number of each subcontractor contacted for negotiation

Include documentation of quotes and/or proposals received from S/M/WBEs and date of negotiations

Economic Development & Diversity Compliance Department · 7720 West Oakland Park Blvd., Suite 323, Sunrise, FL 33351 · (754) 321-0505

Document Number 00480

S/M/WBE PARTICIPATION GOOD FAITH EFFORT FORM

SECTION H – Bonding Requirements Ef	fort	5 Points							
Please provide documentation and sup	porting evidence to show how the crite	ria was fulfilled.							
Waive or reduce bonding requireme	ents for subcontractors.								
Additional efforts, if any: (Provide in	nformation in space provided below)								
Requirements: Include copy of commitment to waive or reduce bonding requirements for S/M/WBEs.									
SECTION I – Assist in Financing, Bonding, and Insurance Effort 5 Points									
Please provide documentation and sup	porting evidence to show how the crite	ria was fulfilled.							
Advise and make efforts to assist interested S/M/WBEs in obtaining bonds, lines of credit, or insurance that SBBC requires.									
Additional efforts, if any: (Provide in	nformation in space provided below)								
Requirements:									
	pe of assistance offered to S/M/WBEs, in ompany or financial institution offering a	•							
Respondents will be considered non-responsive upon denial of the S/M/WBE Participation Good Faith Effort Form and Subcontractor Utilization Plan. For more information, please contact the EDDC Office at (754) 321-0505.									
AFFIRMATION I CERTIFY THAT ALL INFORAMTION CONTAINED IN THEIS FORM IS ACCURATE AND COMPLETE, AND UNDERSTAND THAT IF THIS REQUEST FOR WAIVER IS DENIED AND I FAIL TO MEET THE REQUIREMENTS OF THIS SOLICITATION, MY RESPONSE TO THIS SOLICITATION WILL BE DEEMED NON-RESPONSIVE.									
Signature:	Print Name/Title:								
Economic Development & Diversity	Approved	TOTAL SCORE:							
Compliance Department (EDDC)	Denied	/100							
Economic Development & Diver	sity Compliance Department (FDDC) - 77	20 West Oakland Bark Plyd Suite 222							

REQUIRED RESPONSE FORM - Proposer Information

RFQ Issued Date:	Title of Request or Solicita	tion:
	Note: For Joint Venture Proposals, see inst PROPOSER INFO	
PROPOSER'S (COMPANY) NAM	ME:	
PROPOSER TELEPHONE:	PROPOSE	R FAX:
CONTACT PERSON:		
CONTACT TELEPHONE:	CONTACT	FAX:
E-MAIL ADDRESS TO SEND PU	JRCHASE ORDERS TO:	
INTERNET E-MAIL ADDRESS: _	INTE	RNET URL:
PROPOSER FIRM - TAX IDENT	IFICATION NUMBER:	
 Proposer has not discussed, o Proposer, its principals, or the Proposer is attempting to qualify, silence" period for any solicitation School Board Policy School Board Policy Proposer acknowledges that a Records Laws. All responses, data and informated. Proposer agrees to acceptance issued Addenda. Proposer agrees to be bound to the proposer agrees to be bound to the proposer agrees to be proposer agrees. 	eir lobbyists has not provided any campaign of to provide Design Services to the School Boan for a competitive procurement as described I 3320, Part II, Section HH 1007, Section 5.4 Campaign Contribution Fun all information contained herein is part of the ation contained in this Proposal are true and a e of the contents of all pages in this Request for to all terms, conditions and requirements identi	osers and has not colluded with any other Proposer. ontributions to School Board Members during the period in which the rd. This period of limitation shall commence at the time of the "cone of by: draising public record as defined by the State of Florida Sunshine and Public
	er (blue ink preferred on original)	Date
Name of Proposer's Officer		Title of Proposer's Officer.

JOINT VENTURES: In the event multiple Proposers submit a joint Proposal in response to the RFQ, a single Proposer shall be identified as the Prime Proposer. If offering a joint Proposal, Prime Proposer must include the name and address of all parties of the joint Proposal. Prime Proposer shall provide all bonding and insurance requirements, execute any Contract, complete the <u>REQUIRED RESPONSE FORM</u> shown herein, have overall and complete accountability to resolve any dispute arising within this contract. Only a single contract with one Proposer shall be acceptable. Prime Proposer responsibilities shall include, but not be limited to, performing of overall contract administration, preside over other Proposers participating or present at SBBC meetings, oversee preparation of reports and presentations, and file any notice of protest and final protest as described herein. Prime Proposer shall also prepare and present a consolidated invoice(s) for services performed. SBBC shall issue only one check for each consolidated invoice to the Prime Proposer for services performed. Prime Proposer shall remain responsible for performing services associated with response to this RFQ.



The School Board of Broward County, Florida Procurement & Warehousing Services Department 7720 W. Oakland Park Blvd., Suite 323 Sunrise, Florida 33351 (754) 321-0505

Document 00220: Submitting Firm's Request for Information									
	Varehousing Services Department d Park Blvd., Suite 323	(For Owner's Use Only) Firm's RFI No.:							
Project: Facility Name:	RFQ/RFP#, #, & Locatio Project Cons	on #:							
Category: Information not sh	iown on RFQ/RFP Documents FQ/RFP Documents FP Requirements	Reference: Drawing Reference Spec/Project Manual Reference Other:							
Subject:									
Description:									
Submitting Firm: Company Name & Address: Phone:	Ву:	Signature							

Replies will be issued by Addendum and made available on Demand Star to all prospective Submitting Firms. No verbal or written interpretations of the RFQ / RFP Documents will be given to individual Submitting Firms.

PROCUREMENT & WAREHOUSING SERVICES

Broward County Public Schools

PROGRAM / PROJECT RELEVANCY / REFERENCE FORM (PMOR SERVICES)

Instructions to Submitting Firm: Submit a separate Form for each related program or project listed in the firm's RFQ Response. It is the firm's responsibility to provide accurate, current information for each program or project and Owner listed under sections 1, 2, and 3. Item 2 below along with any attachments about relevant projects will be reviewed and evaluated by QSEC. References will equal up to 1.5 points per program or project. This form does not count towards the page count for Relevant Projects, nor should it be used in lieu of the relevant project information. Relevant programs are eligible for up to 3.5 points per program. Relevant projects that meet the minimum requirements are eligible for up to 3.0 points per program.

1.	NAME OF PRIME FIRM: Enter Subm	itting Firm's name			RFQ #: FY20-192					
2.	SOLICITATION FOR: Program Management / Owner's Representative Services 2. RELEVANT PROJECT/PROGRAM NAME/TITLE & BRIEF SCOPE, INCLUDING APPROACH INFORMATION (list your firm's scope on the project):									
	OWNER'S NUMBER: PROJECT/PROGRAM LOCATION (Address, City, State & Zip):									
	CONSTRUCTION DELIVERY METH			nal (D/B/B) □CM at-Risk □I Owner's Representative Ser						
	ORIGINAL EST. COST OF CONSTR		PLANNED CONSTRUCTION SCHEDULE (DAYS):							
	FINAL CONSTRUCTION COST:	FINAL CONSTRUCTION COST:								
	PROPOSER'S CONTRACTUAL REI	LATIONSHIP TO OWNER (NC	PLANNED COMPLETION DATE: Month/Year						
				ACTUAL COMPLETION D	ATE: Month/Year					
MINORITY/WOMEN BUSINESS ENTERPRISE OR SMALL BUSINESS ENTERPRISE PERCENTAGE GOALS VS ACTUAL PERCENTAGES:										
3.	PROJECT OWNER:	C	OFFI	FICE / DEPARTMENT:						
	REFERENCE NAME:		TITLE:							
	E-MAIL:	F	РНО	ONE NO.:						
	ADDRESS (Address, City, State & Zip): Click here to enter address									

Revised 4/14/2020 Page 1 of 2

PROGRAM / PROJECT RELEVANCY / REFERENCE FORM (PMOR SERVICES) Scoring Rubric for Relevant Program / Projects / References (PMOR Services)

ı	R	FI	(٦I	M	19	TC) RE	LITH	17ED	RV	RC	DS	OSFC:	

Re	eference	(Circle only one box for each to indicate response received and the points allocated to the response.)						
a.	Is the information listed in Section 2 , above, accurate? (Staff to verify) (0.25 points)			YES (0.25 Points)	NO If "no", explain under 4.d. Clarification			
b.	How would you rate the <u>overall</u> performance of this firm in terms of: (1 Point Maximum)			GOOD (1 Point)	FAIR (or N/A) POOR (0.5 Points)			
	 Timely completion of the project/program; Cost effectiveness (was project within budget); Quality of the communications/coordination with the program/project team members; Responsiveness to the owner; and Quality of the project. 							
C.	Would you do business with this firm again? (0.25 Points) If "no" or "maybe", explain under 4.d. Clarification			YES (0.25Points)	NO MAYBE (0 Points)			
d.	d. Clarification, if required (as it relates to Section 2 , above):							
Re	elevant Program/Project							
• F	Relevancy of Program/Project and Scope)						
a.	Is this a Program Reference? (0.5 Maximu	ım Points)						
(0.5 Point) YES				(0 Points) NO				
b	b. How relevant is the scope provided to the scope of this RFQ? (1.75 Maximum Points)							
(1.75 Points) (1.0 Point) Very Somewhat				inimal (0 Points) Not at all				
Comments:								
c. Project/Program is a public educational facility project/program in the State of Florida (required to follow SREF). (1.0 Maximum Points)								
(1.0 Points) YES				(0 Points) NO				
 d. Project/Program is an educational facility project/program in the United States of America. (0.25 Maximum Points) 								
(0.25 Points) YES				(0 Points) NO				
Name of representative completing this form (print name):								
<u>Signatory:</u> By signing and submitting this project reference form, I hereby attest that I am authorized to provide this information and that the above information is true, accurate and correct to the best of my knowledge (Staff Signature)								
Si	gnature:		Date:					

Revised 4/14/2020 Page 2 of 2

Reference Total Score: _____

The School Board of Broward Count	y, Florida	
RFQ # and NAME		
DISCLOSURE OF POTE	NTIAL CONFLICT OF INTEREST AND CONFL CONTRACTUAL RELATIONSHIP	ICTING EMPLOYMENT OR
	Proposer must disclose, in its RFQ, the names of SBBC. Persons identified below may have of stutes.	
Name of Proposer's Employee	SBBC Title or Position of Proposer's Employee	SBBC Department/ School of Proposer's Employee
Check one of the following and sign:		
☐ I hereby affirm that there are no	known persons employed by Proposer who are a	also an employee of SBBC.
☐ I hereby affirm that all known pe identified above.	ersons who are employed by Proposer, who are	also an employee of SBBC, have been
Signature	Company	Name
Name of Official	Business A	address
	City, State, Zip Code	



The School Board of Broward County, Florida Office of Facilities & Construction 2301 N.W. 26th Street Fort Lauderdale, Florida 33311

(754) 321-1500

			(754) 321-1500
		thorization To Pi ne: PSA □ / CSA	
Project No.: Location No.:		Date:	
Project Title:		SBBC P.O. No./I	ine No.:
		RFQ No.:	
Facility Name:		Project Manager:	
Project Consultant:		ATP Start Date (Da	ıy1):
proceed with the Schematic Design Bidding Other Services:	Design Develope Construction Co Attached: on to Proceed is subject to	ment ntract Administration the following attachmen vices and Fee Schedule (Genda Request Form)	Construction Documents Warranty
	ne required services is s st (FLCC) for this project		onal Services Required with a Fixed Limit of
Original FLCC	Curre	ent Cost Estimate	Revised FLCC by ATP
\$		\$	\$
for the Project lis	rofessional services will b sted above: Discipline	be provided by the Project Description	Consultant as a normal part of its Basic Services



The School Board of Broward County, Florida Office of Facilities & Construction 2301 N.W. 26th Street Fort Lauderdale, Florida 33311

(754) 321-1500

Authorization To Proceed

Project Sc	hedule		
Project No.: Project Titl	le		
Location No.:			
Facility Name:			
Project Consultant:			
The required project schedule milestones for this project are presented	below. (Mandatory)		
	Date Required Or		
ACTIVITY –	Estimated Time		
Must Match (PSA/CSA/Term Per RFQ Phasing Plan) # Days per PSA/CSA/Term RFQ Schedule Final Negotiated Schedule	Period Start	Finish	
Phase I – Scope Validation	Day 1	Day	=
Phase I – Plan Review	Day	Day	
Phase I – Schematic Design	Day	Day	
Phase I – Plan Review	Day	Day	
Phase II – Design Development Submittal – 30% CDs	Day	Day	
Phase II – Plan Review	Day	Day	
Phase III – Construction Document – 50% CDs	Day	Day	
Phase III – Plan Review	Day	Day	
Phase III – Construction Document – 60% CDs	Day	Day	
Phase III – Plan Review	Day	Day	
Phase III – Submittal – 90% CDs	Day	Day	
Phase III – Review – 90 % CDs	Day	Day	

Day

Day

Day

Day

Day

Day

Day

Day

Day

Phase III – Submittal – 100% CDs

Phase III – Bldg. Dept. Review – 100% CDs

Plan Revision / Re-Submittal to Bldg. Dept. -

Construction Through Close-out (Contract Administration) –

Phase III – Review – 100% CDs

Bldg. Dept. Follow-up Review -

Procurement & Award/Bidding

Plan Approval –

Warranty -

Day

Day

Day

Day

Day

Day

Day

Day

Day



The School Board of Broward County, Florida Office of Facilities & Construction 2301 N.W. 26th Street Fort Lauderdale, Florida 33311

(754) 321-1500

Consultant's Authorization To Proceed							
	P	rofessio	nal Fees				
Project No. & Location No.:		Project Ti	tle:				
Facility Name:							
Project Consultant:							
	T					1	
Phase Original Fee Previously Must Match PSA/CSA/Term RFQ Basic Fee Fee Authorized by ATP Authorized Fee Balance							e Ralance
Phase I – Scope Validation	\$	\$	%	\$	%	\$	<u>%</u>
Phase I – Scope variation Phase I – Schematic Design	\$	\$	%	\$	%	\$	%
Phase II – Design Development Submittal – 30% CDs	\$	\$	%	\$	%	\$	%
Phase II – Construction Document – 50% CDs / 60% CDs	\$	\$	%	\$	%	\$	%
Phase III – 90% CDs	\$	\$	%	\$	%	\$	%
Phase III – 100% CDs	\$	\$	%	\$	%	\$	%
Pre-Construction Services (CMAR Only)	\$	\$	%	\$	%	\$	%
Phase IV - Procurement & Award/Bidding	\$	\$	%	\$	%	\$	%
Phase V – Construction Through Close-out (Contract Administration)	\$	\$	%	\$	%	\$	%
Phase IV – Warranty	\$	\$	%	\$	%	\$	%
Allowance – Document Reproduction	\$	\$	%	\$	%	\$	%
Allowance – Specific Purpose (list) Site Survey, GPR, Other	\$	\$	%	\$	%	\$	%
Allowance – Non-Destructive / Destructive Testing	\$	\$	%	\$	%	\$	%
Supplement Services	\$	\$	%	\$	%	\$	%
Total	\$	\$	%	\$	%	\$	%
Payment for these services s	hall be made in acc	ordance wi	th the provisions	of the Profe	essional Serv	vices Agre	eement
Approved By Consultant Certified by SBBC							
Name:		N	Name: Shelley N. Meloni				
Title:		Т	Title: Director, Pr	e-Construc	etion		
Signature:	Date:	S	signature:			Date	: [

This document is part of the Professional Services Agreement for Open End Services between The School Board of Broward County, Florida (Owner) and the Project Consultant and is incorporated by reference into the terms and conditions of that agreement.

Date:

Name:

Title:

Signature:

Approval by SBBC

Name: Title:

Signature:

Certified By SBBC / Owner's Representative

Date:

ACH Payment Agreement Form (ACH CREDITS)

VENDOR NAME:		
	Authorization Agreement	

I (we) hereby authorize <u>The School Board of Broward County</u> to initiate automatic deposits (credits) to my account at the financial institution named below. Additionally, I authorize <u>The School Board of Broward County</u> to make the necessary debit entries/adjustments in the event that a credit entry is made in error.

Further, I agree not to hold **The School Board of Broward County** responsible for any delay or loss of funds due to incorrect or incomplete information supplied by me or by my financial institution or due to an error on the part of my financial institution in depositing funds to my account.

This agreement will remain in effect until **The School Board of Broward County** receives written notification of cancellation from me or my financial institution and that the origination of ACH transactions to my (our) account must comply with the provisions of U.S. law.

Name of Bank or Financial Institution:		
Branch/ State:		
Routing No:		
	Che <u>cki</u> ng	Saving
VENDOR AREA: Remittance Confirmation:	Fax	Em
(please select one)	· · · · · · · · · · · · · · · · · · ·	
Federal Identification No.	TAX ID#	S
Vendor		
Update Pu	rchase Order Fax & Email Address	
Centralized Fax Number	Dep	ot
Centralized Email	Dep	ot
Centralized Phone No.	Dep	ot
	•	
Authorized Signature		e:
Authorized Signature (Primary) and Business title: Authorized Signature	Signature Date	
Authorized Signature (Primary) and Business title:	Signature Date	e: ::
Authorized Signature (Primary) and Business title: Authorized Signature (Joint) and Business title:	Signature Date	
Authorized Signature (Primary) and Business title: Authorized Signature (Joint) and Business title: Please attach a VOIDED chec	Signature Date	
Authorized Signature (Primary) and Business title: Authorized Signature (Joint) and Business title: Please attach a VOIDED chec This form must be returned to 7720 W. Oakland Park Blvd, Sunrise	Signature Date Lambda to verify bank details and routing number. SBBC - Purchasing - Data Strategy Group	
Authorized Signature (Primary) and Business title: Authorized Signature (Joint) and Business title: Please attach a VOIDED chec This form must be returned to 7720 W. Oakland Park Blvd, Sunrise	Signature Date k to verify bank details and routing number. SBBC – Purchasing – Data Strategy Group FL 33351 call: 754-321-0516 or fax # 754-321-0533 Use by DATA STRATEGY GROUP	

School Board of Broward County ACH payment agreement form

AGREEMENT

THIS AGREEMENT	is made and entered into as of thi	s day of	
2020, by and between			

THE SCHOOL BOARD OF BROWARD COUNTY, FLORIDA

(hereinafter referred to as "SBBC"), a body corporate and political subdivision of the State of Florida, whose principal place of business is 600 Southeast Third Avenue, Fort Lauderdale, Florida 33301

and

[VENDOR'S FULL NAME]

(hereinafter referred to as "VENDOR"), whose principal place of business is [VENDOR's address]

WHEREAS, SBBC issued a Request for proposal identified as RFQ FY20-192 – Program Manager – Owner's Representative Services dated and amended herein referred to as "RFQ" all of which are incorporated by reference as if fully set forth herein for the purpose of receiving proposals for Program Manager – Owner's Representative Services; and

WHEREAS, VENDOR offered a proposal dated (hereinafter referred to as "Proposal" which is incorporated by reference herein) in response to this RFQ; and

WHEREAS, VENDOR is willing to provide Program Manager – Owner's Representative Services as more fully described in the Scope of Services of the RFQ.

NOW, THEREFORE, in consideration of the premises and of the mutual covenants contained herein and other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the Parties hereby agree as follows:

ARTICLE 1 - RECITALS

- 1.01 **Recitals**. The parties agree that the foregoing recitals are true and correct and that such recitals are incorporated herein by reference.
- 1.02 **FEMA Requirements**. NOTE: Articles within this Agreement which include **"FEMA Requirements"** in the title are only applicable to services to be performed concerning disaster recovery work.

ARTICLE 2 – SPECIAL CONDITIONS

2.01 <u>Term of Agreement</u>. Unless terminated earlier pursuant to Section 3.05 of this Agreement, the term of this Agreement shall be three (3) years commencing on **[commencement date]** and ending on **[expiration date]**. The term of the contract may, at the sole discretion of

SBBC together with the agreement of VENDOR, be extended for two additional one-year periods and, if needed, 180 days beyond the expiration date of the renewal period. Procurement & Warehousing Services Department, will, if considering renewing, request a letter to renew from VENDOR, prior to the end of the contract period. Any renewal period shall be approved by the Amendment to this Agreement.

2.02 Contract Documents and Priority of Documents. The Contract Documents shall consist of this Agreement for provision of Program Manager – Owner's Representative Services between the SBBC and VENDOR, RFQ FY20-192 inclusive of its attachments, and any addenda thereto, Statement of Qualifications submitted by VENDOR in response to RFQ FY20-192, Scope of Services contained in the RFQ FY20-192, Notice(s) to Proceed, Project Specific Drawings, Project Specific Specifications, current District Design Criteria, post-award meeting checklist and current District Master Specifications. The Contract Documents supersede all prior negotiations, correspondence, conversations, agreements, and understandings applicable to the matters contained herein and the parties agree that there are no commitments, agreements, or understandings concerning the subject matter of this Contract that are not contained in this document. Accordingly, the parties agree that no deviation from the terms hereof shall be predicated upon any prior representations or agreements, whether oral or written.

In the event of a conflict between Contract Documents, the following priority of documents shall govern.

- A. This Agreement; then
- B. RFQ FY20-192 Program Manager Owner's Representative Services, inclusive of all its attachments, as may be amended via Addenda (the "RFQ"); then
- C. Statement of Qualifications submitted in response to the RFQ by VENDOR.
- Cost of Services. SBBC shall pay VENDOR for services rendered under this Agreement and Fee Schedule on a time and material basis with a not to exceed amount for the term of the Agreement, as may be adjusted pursuant to the Performance-Based Fees provision set forth in Section 6.6 of the RFQ. If SBBC and VENDOR desire to renew this Agreement for one or two additional years beyond the original term, the Fee Schedule for years may be renegotiated with VENDOR. If there is no negotiation regarding the Fee Schedule, the Fees Schedule for years four and five shall remain the same as presented in this Agreement. Fee Schedule is defined as the maximum rate per hour or task for which services can be billed to SBBC. These listed fees are full compensation for VENDOR's direct personnel expenses, mandatory and customary contribution, overhead costs, project costs and profit. For all services rendered, the VENDOR will be paid upon actual hourly or salaried rates paid by the VENDOR, times the actual hours worked, times a fixed multiplier of. As payments for services are based upon actual salaries paid, SBBC shall not be billed for hours worked for salaried employees in excess of forty (40) hours per week. The VENDOR shall obtain prior approval, by SBBC, for all materials and reimbursables. All SBBC approved materials and reimbursables will be paid by SBBC at actual costs as a passthrough expense. Attachment A is only a reference document used for the estimate of the not to exceed amount. VENDOR shall obtain prior approval of any and all staff, including their hourly rate selected for commissioning services.

2.04 <u>Billing Instructions and Payment</u>. VENDOR shall be required to bill on not less than a monthly basis, and in no event shall VENDOR submit an invoice for services and costs more than ninety (90) days after the services and/or costs occurred, otherwise such services and costs shall not be considered by SBBC and same services and costs are admitted by the VENDOR as waived and not due to the VENDOR. All costs shall be reasonable and subject to SBBC's approval.

Pursuant to Section 6.7.7 of the RFQ, during the initiation phase, as a condition precedent to payment, VENDOR shall timely submit a 12-month staffing plan, acceptable to SBBC, that correlates with the projected status of individual projects and the overall program over the 12-month period.

Pursuant to Section 6.7.7 of the RFQ, on a quarterly basis, as a condition precedent to payment, VENDOR shall timely submit a 12-month staffing plan that evaluates each team according to the projected status of individual projects and the overall program over the 12-month period.

On a monthly basis, as a condition precedent to payment, provide a written three (3) month look ahead document, acceptable to SBBC, which, among other things, lists the staffing plan (including the staff rates and other fees) for the three (3) months following the date of the three (3) month plan, and the month prior to the date of the three (3) month plan. SBBC, at its sole discretion shall approve the monthly written three (3) month look ahead document prior to its implementation.

- 2.05 <u>Services</u> VENDOR shall provide Program Management Services pursuant to Section 6 of RFQ FY20-192, all addenda issued to the RFQ, if any, the VENDOR's response to the RFQ, and any SBBC ordering or authorizing documents such as the NTP are also made a part of this Agreement. VENDOR will attend meetings as requested by the Executive Director or his/her designee, Capital Programs. Meetings may be conducted during or after regular business hours. Examples of types of meetings which VENDOR may be requested to attend:
 - a. Senior Administration Meetings
 - b. Community Meetings
 - c. Project Review Meetings
 - d. Construction Oversight Review Advisory Committee (CORC)
 - e. Coordination Meeting with District Staff
 - f. Local Governmental Agencies
 - g. Operations/Budget Staff Meetings
 - h. Capital Program Staff Meetings
 - i. School Board Meetings
- 2.06 <u>M/WBE Participation</u>. VENDOR shall provide for M/WBE participation during its performance of services under this Agreement by listing SBBC Certified MWBEs' participation commitment percentage set forth in the VENDOR's proposal (Attachment B).

VENDOR shall obtain prior written approval from the Officer, Economic Development & Diversity Compliance Department for any replacement of any of the entities listed above. Utilizing any entity other than the ones listed, respectively will be considered a breach of this

Agreement. VENDOR is subject to debarment and any other remedy available for any breaches to this Agreement.

Minority and/or Women-Owned Business Enterprise (M/WBE) Subcontracting Goals Program:

SBBC has identified a **twenty-nine percent (29%) participation goal** for Submitting Firms that subcontract with eligible SBBC-Certified M/WBE firms for at least twenty-nine percent (29%) of the total contract value. See the definition of M/WBE in Section C.2. of Policy 3330. BIPM Comment – Should the reference Section read C.2. of the Policy?

Penalties and Sanctions

In the absence of a waiver granted by the SDOP or the self-performance of a portion or all of the SBE subcontracting goal by a certified SBE proposer, the failure of a proposer to attain a subcontracting goal for SBE participation in the performance of its contract or otherwise comply with the provisions of this API shall be considered a material breach of contract, grounds for termination of that contract with the SBBC, and shall be subject to any penalties and sanctions available under the terms of the SDOP policy, its contract terms with the SBBC, or by law pursuant to the penalties and sanctions set forth in Section 13 of the Standard Operating Procedures for this Policy.

Suspension BIPM Comment – For clarification – Since the Goal is specific to M/WBE, should M/WBE be used below. Or is "SBE" being used in general terms?

The temporary stoppage of a M/WBE firm's beneficial participation in the District's SDOP for a finite period of time due to cumulative contract payments the M/WBE firm received during a fiscal year that exceed a certain dollar threshold as set forth in Section (7) of the Standard Operating Procedures for this Policy or pursuant to the penalties and sanctions set forth in Section 13 of the Standard Operating Procedures for this Policy.

The SBBC Supplier Diversity Outreach Program works to increase the participation of small, minority and women-owned business enterprises in construction and purchasing contracts. It is the intent of the School Board of Broward County to have a diverse group of vendors to participate in the procurement process. The current list of SBBC S/M/WBE-Certified firms can be viewed at: browardschools.com/Page/46981.

2.07 <u>Inspection of VENDOR's Records by SBBC</u>. VENDOR shall establish and maintain books, records and documents (including electronic storage media) sufficient to reflect all income and expenditures of funds provided by SBBC under this Agreement. All VENDOR's Records, regardless of the form in which they are kept, shall be open to inspection and subject to audit, inspection, examination, evaluation and/or reproduction, during normal working hours, by SBBC's agent or its authorized representative to permit SBBC to evaluate, analyze and verify the satisfactory performance of the terms and conditions of this Agreement and to evaluate, analyze and verify any and all invoices, billings, payments and/or claims submitted by VENDOR or any of VENDOR's payees pursuant to this Agreement. VENDOR's Records subject to examination shall include, without limitation, those records necessary to evaluate and verify direct and indirect costs (including overhead allocations) as they may apply to costs associated with this Agreement. VENDOR's Records subject to this section shall include any and all documents pertinent to the evaluation, analysis, verification and reconciliation of any and all expenditures under this Agreement without regard to funding sources.

- (a) <u>VENDOR's Records Defined</u>. For the purposes of this Agreement, the term "VENDOR's Records" shall include, without limitation, and any supporting documents that would substantiate, reconcile or refute any charges and/or expenditures related to this Agreement.
- (b) <u>Duration of Right to Inspect</u>. For the purpose of such audits, inspections, examinations, evaluations and/or reproductions, SBBC's agent or authorized representative shall have access to VENDOR's Records from the effective date of this Agreement, for the duration of the term of this Agreement, and until the later of five (5) years after the termination of this Agreement or five (5) years after the date of final payment by SBBC to VENDOR pursuant to this Agreement.
- (c) <u>Notice of Inspection</u>. SBBC's agent or its authorized representative shall provide VENDOR reasonable advance notice (not to exceed two (2) weeks) of any intended audit, inspection, examination, evaluation and or reproduction.
- (d) <u>Audit Site Conditions</u>. SBBC's agent or its authorized representative shall have access to VENDOR's facilities and to any and all records related to this Agreement, and shall be provided adequate and appropriate work space in order to exercise the rights permitted under this section.
- (e) <u>Failure to Permit Inspection</u>. Failure by VENDOR to permit audit, inspection, examination, evaluation and/or reproduction as permitted under this Section shall constitute grounds for termination of this Agreement by SBBC for cause and shall be grounds for the denial of some or all of any VENDOR's claims for payment by SBBC.
- (f) Overcharges and Unauthorized Charges. If an audit conducted in accordance with this Section discloses overcharges or unauthorized charges to SBBC by VENDOR in excess of two percent (2%) of the total billings under this Agreement, the actual cost of SBBC's audit shall be paid by VENDOR. If the audit discloses billings or charges to which VENDOR is not contractually entitled, VENDOR shall pay said sum to SBBC within twenty (20) days of receipt of written demand under otherwise agreed to in writing by both parties.
- (g) <u>Inspection of Subcontractor's Records</u>. VENDOR shall require any and all subcontractors, insurance agents and material suppliers (hereafter referred to as "Payees") providing services or goods with regard to this Agreement to comply with the requirements of this section by insertion of such requirements in any written subcontract. Failure by VENDOR to include such requirements in any subcontract shall constitute grounds for termination of this Agreement by SBBC for cause and shall be grounds for the exclusion of some or all of any Payee's costs from amounts payable by SBBC to VENDOR pursuant to this Agreement and such excluded costs shall become the liability of VENDOR.
- (h) <u>Inspector General Audits</u>. VENDOR shall comply and cooperate immediately with any inspections, reviews, investigations, or audits deemed necessary by the Florida Office of the Inspector General or by any other state or federal officials.
 - 2.07.01 **Quarterly Evaluation.** An evaluation shall be conducted by RSM US LLP and the assigned District Staff on a quarterly basis to measure performance. Such evaluation shall be communicated to VENDOR to identify areas for betterment and shall be used as basis for continuous improvement.

2.08 <u>Notice</u>. When any of the parties desire to give notice to the other, such notice must be in writing, sent by U.S. Mail, postage prepaid, addressed to the party for whom it is intended at the place last specified; the place for giving notice shall remain such until it is changed by written notice in compliance with the provisions of this paragraph. For the present, the Parties designate the following as the respective places for giving notice:

To SBBC: Superintendent of Schools

The School Board of Broward County, Florida

600 Southeast Third Avenue Fort Lauderdale, Florida 33301

With a Copy to: Director of Procurement and Warehousing Services

The School Board of Broward County, Florida 7720 West Oakland Park Blvd., Suite 323

Sunrise, Florida 33351

Executive Director, Capital Program

The School Board of Broward County, Florida

600 Southeast Third Avenue Fort Lauderdale, Florida 33301

To VENDOR:

- 2.09 **Background Screening:** VENDOR agrees to comply with all requirements of Sections 1012.32 and 1012.465, Florida Statutes, and all of its personnel who (1) are to be permitted access to school grounds when students are present, (2) will have direct contact with students, or (3) have access or control of school funds, will successfully complete the background screening required by the referenced statutes and meet the standards established by the statutes. This background screening will be conducted by SBBC in advance of VENDOR or its personnel providing any services under the conditions described in the previous sentence. VENDOR shall bear the cost of acquiring the background screening required by Section 1012.32, Florida Statutes, and any fee imposed by the Florida Department of Law Enforcement to maintain the fingerprints provided with respect to VENDOR and its personnel. The parties agree that the failure of VENDOR to perform any of the duties described in this section shall constitute a material breach of this Agreement entitling SBBC to terminate immediately with no further responsibilities or duties to perform under this Agreement. VENDOR agrees to indemnify and hold harmless SBBC, its officers and employees from any liability in the form of physical or mental injury, death or property damage resulting from VENDOR failure to comply with the requirements of this Section or with Sections 1012.32 and 1012.465, Florida Statutes.
- 2.10 <u>Liability</u>. This section shall survive the termination of all performance or obligations under this Agreement and shall be fully binding until such time as any proceeding brought on account of this Agreement is barred by any applicable statute of limitations.
 - A. By SBBC: SBBC agrees to be fully responsible up to the limits of Section 768.28, Florida Statutes, for its acts of negligence, or its employees' acts of negligence when acting within the scope of their employment and agrees to be liable for any damages resulting from said negligence.

- By VENDOR: VENDOR agrees to indemnify and hold harmless SBBC, В. its agents, servants and employees from any and all claims, judgments, costs, and expenses including, but not limited to, reasonable attorney's fees, reasonable investigative and discovery costs, court costs and all other sums which SBBC, its agents, servants and employees may pay or become obligated to pay on account of any, all and every claim or demand, or assertion of liability, or any claim or action founded thereon, arising out of the products, goods or services negligently furnished by VENDOR, its agents, servants or employees; the equipment of VENDOR, its agents, servants or employees while such equipment is on premises owned or controlled by SBBC; or the negligence of VENDOR or the negligence of VENDOR's agents when acting within the scope of their employment, whether such claims, judgments, costs and expenses be for damages, damage to property including SBBC's property, and injury or death of any person whether employed by VENDOR, SBBC or otherwise. This indemnification provision does not extend to require indemnification to the SBBC for its own negligence and, therefore, certain requirements of Section 725.06, Florida Statutes, are not required. However, should this clause be interpreted by a court to require compliance with other provisions of Section 725.06, Florida Statutes, then the following shall be incorporated by reference into the Agreement: (i) the indemnification as provided in this paragraph shall be subject to a monetary limitation equal to Six Hundred Million Dollars (\$600,000,000.00), or the total value (including fees and cost) of all projects assigned to VENDOR, whichever is less, and (ii) the SBBC and VENDOR both acknowledge that the monetary limitation referenced above bears a reasonable commercial relationship to this Agreement.
- Professional Liability (Errors and Omissions): The VENDOR shall Insurance. procure a Professional Liability Insurance Policy with coverage of not less than Five Million (\$5,000,000) Dollars and a deductible, to be paid by VENDOR, of not more than Fifty Thousand (\$50,000) Dollars, providing for all sums which the VENDOR shall be legally obligated to pay as damages for claims arising out of or relating to the Work performed by the VENDOR or any VENDOR's employed or acting on the behalf, but not limited Subconsultants/Subcontractors, in connection with this Agreement. The insurance policy, or a policy with the same terms and conditions, shall remain in full force and effect during the project and for a period of time terminating three (3) years after final completion of the project. If the Owner, at its sole discretion, agrees that such coverage is not commercially reasonably available. the Owner may authorize the VENDOR to alter the coverage by substituting a lower aggregate or changing any other terms and conditions of the coverage, including but not limited to deductible amounts, based upon the scope of the Project. Notwithstanding any provision in this Agreement, nothing in this Agreement shall be construed as a limitation of the VENDOR's liability to SBBC.
- A. <u>Worker's Compensation Insurance.</u> The VENDOR shall maintain Worker's Compensation Insurance in accordance with Florida Statutory Limits and Employers Liability Insurance with a limit of not less than Five Hundred Thousand \$500,000/\$500,000/\$500,000 (each accident/disease-each employee/disease-policy limit).
- B. <u>Commercial General Liability Insurance</u>. The VENDOR shall maintain Comprehensive General Liability Insurance, including Products & Completed Operations, Personal and Bodily Injury, and Contractual Liability, to cover the indemnification language set forth herein. Limits shall not be less than \$1,000,000 per occurrence; \$2,000,000 general aggregate. Products Completed Operations aggregate shall not be less than \$2,000,000. **The**

School Board of Broward County must be named as an Additional Insured on the Certificate along with the RFO Number.

C. <u>Automobile Liability Insurance</u>. The VENDOR shall maintain Automobile Liability Insurance covering all Owned, Non-Owned and Hired vehicles in the amount of not less than One Million Dollars (\$1,000,000) per occurrence Combined Single Limit for Bodily Injury and Property Damage and shall not be any less restrictive than the standard ISO Business Auto Policy CA 00 01. In the event the VENDOR does not own any vehicles, VENDOR shall obtain hired and non-owned coverage for \$1,000,000 Combined Single Limits and provide an affidavit signed by the VENDOR indicating the following:

(Insured) does not own any vehicles. In the event we acquire any vehicles throughout the term of this Agreement, Insured agrees to provide of "Any Auto" coverage effective date of acquisition.

D. <u>Certificate of Insurance Requirements</u>. Prior to the commencement of any Work, as evidence of required coverage, VENDOR must provide a Certificate of Insurance to The School Board of Broward County, Florida's Risk Management Department for approval. Certificates must be faxed to the School Board of Broward County's Certificate Tracking System at 1-866-897-0425.

Liability Policies are to contain the following provisions. <u>In addition, the following wording must be included on the Certificate of Insurance</u>:

The School Board of Broward County, Florida, its memb	ers, officers,
employees, and agents are added as additional insured. T	`he
endorsement # is .	

All liability policies are primary of all other valid and collectable coverage maintained by The School Board of Broward County, Florida.

Contractual liability is included in the General Liability policy.

2.12 <u>Nonconformance to Contract Conditions</u>. Services offered must comply with RFQ conditions and specifications, specifically including but not limited to Sections 6.2.1 and 6.3.4 of the RFQ, and any resulting Agreement at all times. Service(s) not conforming to RFQ conditions, specifications or time frames will have a written nonconformance letter sent to the vendor to provide an opportunity to remedy the nonconformance. Failure of the VENDOR to remedy the nonconformance, within thirty (30) days from receipt of notification, shall result in termination of contract. Additionally, for a period of two (2) years, any RFQ submitted by VENDOR will not be considered or recommended for award, as such, all departments being advised not to do business with VENDOR.

Key Personnel* indicated in the PM's Qualifications Statement are expected to actually perform services in this program. NOTE: The substitution of Key Personnel listed in the Submitting Firm's Qualifications Statement without the prior written approval by the Superintendent of Schools or his/her designee, shall be a material breach of this Agreement. The substitution of the Principal in Charge and/or Program Director (a/k/a PM Lead) listed in the Submitting Firm's Qualifications Statement without the prior

written approval by <u>The School Board of Broward County</u>, <u>Florida</u>, shall be a material breach of the Agreement.

2.13 <u>Travel</u>. Local travel will be billed as a reimbursable expense in accordance with the current or updated School Board Policy 3400 and any other relevant School Board Policies. Out of county travel and per diem may be allowable at the sole discretion of SBBC. SBBC has delegated authority to the Superintendent of Schools or his/her designee to provide prior approval to VENDOR for any and all travel and per diem. Should any out of county travel and/or per diem be allowed, then it shall be billed and reimbursed in compliance with the current or updated School Board Policy 3400 and/or other relevant School Board Policies.

PLEASE BE ADVISED THAT SCHOOL BOARD POLICY 3400, UNDER THE "ALL TRAVEL" SECTION PARAGRAPH 2 STATES "NEITHER SUBSISTENCE NOR PER DIEM SHALL BE ALLOWED FOR ANY CLASS OF TRAVEL IN BROWARD COUNTY EXCEPT FOR AUTHORIZED BUSINESS OR WORKSHOPS OCCURING BEFORE 6:00 A.M. OR EXTENDING BEYOND 8:00 PM." Additionally, the travel requires prior approval.

2.14 Working Environment. SBBC will provide a functional working environment within an existing school to include, but not limited to, office space, office furniture and appliances, housekeeping, internet service and office network cabling, utilities, use of SBBC's email service, server(s) for file storage, office supplies, and in-house printing, copying, plotting and fax equipment services. VENDOR shall obtain prior approval from SBBC for any additional goods or services required for the working environment.

ARTICLE 3 – GENERAL CONDITIONS

- 3.01 <u>No Waiver of Sovereign Immunity</u>. Nothing herein is intended to serve as a waiver of sovereign immunity by any agency or political subdivision to which sovereign immunity may be applicable or of any rights or limits to liability existing under Section 768.28, Florida Statutes. This section shall survive the termination of all performance or obligations under this Agreement and shall be fully binding until such time as any proceeding brought on account of this Agreement is barred by any applicable statute of limitations.
- 3.02 No Third Party Beneficiaries. The parties expressly acknowledge that it is not their intent to create or confer any rights or obligations in or upon any third person or entity under this Agreement. None of the parties intend to directly or substantially benefit a third party by this Agreement. The parties agree that there are no third party beneficiaries to this Agreement and that no third party shall be entitled to assert a claim against any of the parties based upon this Agreement. Nothing herein shall be construed as consent by an agency or political subdivision of the State of Florida to be sued by third parties in any matter arising out of any contract.
- 3.03 <u>Independent Contractor</u>. The parties to this agreement shall at all times be acting in the capacity of independent contractors and not as an officer, employee or agent of one another. Neither party or its respective agents, employees, subcontractors or assignees shall represent to others that it has the authority to bind the other party unless specifically authorized in writing to do so. No right to SBBC retirement, leave benefits or any other benefits of SBBC employees shall exist as a result of the performance of any duties or responsibilities under this Agreement. SBBC shall not be responsible for social security, withholding taxes, contributions to unemployment compensation funds or insurance for the other party or the other party's officers, employees, agents, subcontractors or assignees.

3.04 **Equal Opportunity Provision**. The parties agree that no person shall be subjected to discrimination because of age, race, color, disability, gender identity, gender expression marital status, national origin, religion, sex or sexual orientation in the performance of the parties' respective duties, responsibilities and obligations under this Agreement.

3.04.01 **FEMA Requirement - Equal Employment Opportunity.**

The following language must be included in any contract involving any grant, contract, loan, insurance, or guarantee involving federally assisted construction which is not exempt from the requirements of the equal opportunity clause. The term "contractor" is used to refer to both the VENDOR and contractors performing work.

The following is hereby incorporated into this an any other agreement for construction work, or modification thereof, as defined in the regulations of the Secretary of Labor at 41 CFR Chapter 60, which is paid for in whole or in part with funds obtained from the Federal Government or borrowed on the credit of the Federal Government pursuant to a grant, contract, loan, insurance, or guarantee, or undertaken pursuant to any Federal program involving such grant, contract, loan, insurance, or guarantee, the following equal opportunity clause: During the performance of this contract, the contractor agrees as follows:

- (1) The contractor will not discriminate against any employee or applicant for employment because of race, color, religion, sex, sexual orientation, gender identity, or national origin. The contractor will take affirmative action to ensure that applicants are employed, and that employees are treated during employment without regard to their race, color, religion, sex, sexual orientation, gender identity, or national origin. Such action shall include, but not be limited to the following: Employment, upgrading, demotion, or transfer; recruitment or recruitment advertising; layoff or termination; rates of pay or other forms of compensation; and selection for training, including apprenticeship. The contractor agrees to post in conspicuous places, available to employees and applicants for employment, notices to be provided setting forth the provisions of this nondiscrimination clause.
- (2) The contractor will, in all solicitations or advertisements for employees placed by or on behalf of the contractor, state that all qualified applicants will receive consideration for employment without regard to race, color, religion, sex, sexual orientation, gender identity, or national origin.
- (3) The contractor will not discharge or in any other manner discriminate against any employee or applicant for employment because such employee or applicant has inquired about, discussed, or disclosed the compensation of the employee or applicant or another employee or applicant. This provision shall not apply to instances in which an employee who has access to the compensation information of other employees or applicants as a part of such employee's essential job functions discloses the compensation of such other employees or applicants to individuals who do not otherwise have access to such information, unless such disclosure is in response to a formal complaint or charge, in furtherance of an

investigation, proceeding, hearing, or action, including an investigation conducted by the employer, or is consistent with the contractor's legal duty to furnish information.

- (4) The contractor will send to each labor union or representative of workers with which he has a collective bargaining agreement or other contract or understanding, a notice to be provided advising the said labor union or workers' representatives of the contractor's commitments under this section, and shall post copies of the notice in conspicuous places available to employees and applicants for employment.
- (5) The contractor will comply with all provisions of Executive Order 11246 of September 24, 1965, and of the rules, regulations, and relevant orders of the Secretary of Labor.
- (6) The contractor will furnish all information and reports required by Executive Order 11246 of September 24, 1965, and by rules, regulations, and orders of the Secretary of Labor, or pursuant thereto, and will permit access to his books, records, and accounts by the administering agency and the Secretary of Labor for purposes of investigation to ascertain compliance with such rules, regulations, and orders.
- (7) In the event of the contractor's noncompliance with the nondiscrimination clauses of this contract or with any of the said rules, regulations, or orders, this contract may be canceled, terminated, or suspended in whole or in part and the contractor may be declared ineligible for further Government contracts or federally assisted construction contracts in accordance with procedures authorized in Executive Order 11246 of September 24, 1965, and such other sanctions may be imposed and remedies invoked as provided in Executive Order 11246 of September 24, 1965, or by rule, regulation, or order of the Secretary of Labor, or as otherwise provided by law.
- (8) The contractor will include the portion of the sentence immediately preceding paragraph (1) and the provisions of paragraphs (1) through (8) in every subcontractor purchase order unless exempted by rules, regulations, or orders of the Secretary of Labor issued pursuant to section 204 of Executive Order 11246 of September 24, 1965, so that such provisions will be binding upon each subcontractor or vendor. The contractor will take such action with respect to any subcontract or purchase order as the administering agency may direct as a means of enforcing such provisions, including sanctions for noncompliance:

Provided, however, that in the event a contractor becomes involved in, or is threatened with, litigation with a subcontractor or vendor as a result of such direction by the administering agency, the contractor may request the United States to enter into such litigation to protect the interests of the United States. The applicant further agrees that it will be bound by the above equal opportunity clause with respect to its own employment practices when it participates in

federally assisted construction work: *Provided*, That if the applicant so participating is a State or local government, the above equal opportunity clause is not applicable to any agency, instrumentality or subdivision of such government which does not participate in work on or under the contract.

The applicant [School Board] and VENDOR agree that to assist and cooperate actively with the administering agency and the Secretary of Labor in obtaining the compliance of contractors and subcontractors with the equal opportunity clause and the rules, regulations, and relevant orders of the Secretary of Labor, that it will furnish the administering agency and the Secretary of Labor such information as they may require for the supervision of such compliance, and that it will otherwise assist the administering agency in the discharge of the agency's primary responsibility for securing compliance.

- 3.05 <u>Termination</u>. This Agreement may be canceled with or without cause by SBBC during the term hereof upon thirty (30) days written notice to the other parties of its desire to terminate this Agreement. SBBC shall have no liability for any property left on SBBC's property by any party to this Agreement after the termination of this Agreement. Any party contracting with SBBC under this Agreement agrees that any of its property placed upon SBBC's facilities pursuant to this Agreement shall be removed within ten (10) business days following the termination, conclusion or cancellation of this Agreement and that any such property remaining upon SBBC's facilities after that time shall be deemed to be abandoned, title to such property shall pass to SBBC, and SBBC may use or dispose of such property as SBBC deems fit and appropriate. VENDOR will be compensated for services performed up to the date of termination and be reimbursed for any and all reimbursable expenses allowed under the Agreement.
- 3.06 <u>Default</u>. The parties agree that, in the event that either party is in default of its obligations under this Agreement, the non-defaulting party shall provide to the defaulting party (30) days written notice to cure the default. However, in the event said default cannot be cured within said thirty (30) day period and the defaulting party is diligently attempting in good faith to cure same, the time period shall be reasonably extended to allow the defaulting party additional cure time. Upon the occurrence of a default that is not cured during the applicable cure period, this Agreement may be terminated by the non-defaulting party upon thirty (30) days notice. This remedy is not intended to be exclusive of any other remedy, and each and every such remedy shall be cumulative and shall be in addition to every other remedy now or hereafter existing at law or in equity or by statute or otherwise. No single or partial exercise by any party of any right, power, or remedy hereunder shall preclude any other or future exercise thereof. Nothing in this section shall be construed to preclude termination for convenience pursuant to Section 3.05.
- Annual Appropriation. The performance and obligations of SBBC under this Agreement shall be contingent upon an annual budgetary appropriation by its governing body. If SBBC does not allocate funds for the payment of services or products to be provided under this Agreement, this Agreement may be terminated by SBBC at the end of the period for which funds have been allocated. SBBC shall notify the other party at the earliest possible time before such termination. No penalty shall accrue to SBBC in the event this provision is exercised, and SBBC shall not be obligated or liable for any future payments due or any damages as a result of termination under this section.

- 3.08 Excess Funds. Any party receiving funds paid by SBBC under this Agreement agrees to promptly notify SBBC of any funds erroneously received from SBBC upon the discovery of such erroneous payment or overpayment. Any such excess funds shall be refunded to SBBC with interest calculated from the date of the erroneous payment or overpayment. Interest shall be calculated using the interest rate for judgments under Section 55.03, Florida Statutes, applicable at the time the erroneous payment or overpayment was made by SBBC.
- Pursuant to Section 119.0701, Florida Statutes, any party 3.09 **Public Records.** contracting with SBBC is required to (a) keep and maintain available for public inspection any records that pertain to services rendered under this Agreement; (b) provide the public with access to public records on the same terms and conditions that SBBC would provide such records and at a cost that does not exceed the cost provided in Chapter 119, Florida Statutes or as otherwise provided by law; (c) ensure that public records that are exempt or confidential and exempt from public records disclosure requirements are not disclosed except as authorized by law; and (d) meet all requirements for retaining public records and transfer, at no cost to SBBC, all public records in that party's possession upon termination of its Agreement with SBBC and destroy any duplicate public records that are exempt or confidential and exempt from public records disclosure requirements. All of such party's records stored electronically must be provided to SBBC in a format that is compatible with SBBC's information technology systems. Each party shall maintain its own respective records and documents associated with this Agreement in accordance with the records retention requirements applicable to public records. Each party shall be responsible for compliance with any public documents request served upon it pursuant to Section 119.07, Florida Statutes, and any resultant award of attorney's fees for non-compliance with that law. Each party acknowledges that this Agreement and all attachments thereto are public records and do not constitute trade secrets.
- 3.10 Student Records: Notwithstanding any provision to the contrary within this Agreement, any party contracting with SBBC under this Agreement shall fully comply with the requirements of Sections 1002.22 and 1002.221, Florida Statutes; FERPA, and any other state or federal law or regulation regarding the confidentiality of student information and records. Each such party agrees, for itself, its officers, employees, agents, representatives, contractors or subcontractors, to fully indemnify and hold harmless SBBC and its officers and employees for any violation of this section, including, without limitation, defending SBBC and its officers and employees against any complaint, administrative or judicial proceeding, payment of any penalty imposed upon SBBC, or payment of any and all costs, damages, judgments or losses incurred by or imposed upon SBBC arising out of a breach of this covenant by the party, or an officer, employee, agent, representative, contractor of the party to the extent that the party or an officer, employee, agent, representative, contractor, or sub-contractor of the party shall either intentionally or negligently violate the provisions of this section or of Sections 1002.22 and/or 1002.221, Florida Statutes.
- 3.11 <u>Compliance with Laws</u>. Each party shall comply with all applicable federal and state laws, codes, rules and regulations in performing its duties, responsibilities and obligations pursuant to this Agreement.
- 3.12 <u>Place of Performance</u>. All obligations of SBBC under the terms of this Agreement are reasonably susceptible of being performed in Broward County, Florida and shall be payable and performable in Broward County, Florida.
- 3.13 Governing Law and Venue. This Agreement shall be interpreted and construed in accordance with and governed by the laws of the State of Florida. Any

controversies or legal problems arising out of this Agreement and any action involving the enforcement or interpretation of any rights hereunder shall be submitted to the jurisdiction of the State courts of the Seventeenth Judicial Circuit of Broward County, Florida.

- 3.14 Entirety of Agreement. This document incorporates and includes all prior negotiations, correspondence, conversations, agreements and understandings applicable to the matters contained herein and the parties agree that there are no commitments, agreements or understandings concerning the subject matter of this Agreement that are not contained in this document. Accordingly, the parties agree that no deviation from the terms hereof shall be predicated upon any prior representations or agreements, whether oral or written.
- 3.15 <u>Binding Effect</u>. This Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns.
- 3.16 <u>Assignment</u>. Neither this Agreement nor any interest herein may be assigned, transferred or encumbered by any party without the prior written consent of the other party. There shall be no partial assignments of this Agreement including, without limitation, the partial assignment of any right to receive payments from SBBC.
- 3.17 <u>Incorporation by Reference</u>. Exhibits attached hereto **Attachment A** and referenced herein shall be deemed to be incorporated into this Agreement by reference.
- 3.18 <u>Captions</u>. The captions, section designations, section numbers, article numbers, titles and headings appearing in this Agreement are inserted only as a matter of convenience, have no substantive meaning, and in no way define, limit, construe or describe the scope or intent of such articles or sections of this Agreement, nor in any way effect this Agreement and shall not be construed to create a conflict with the provisions of this Agreement.
- 3.19 **Severability**. In the event that any one or more of the sections, paragraphs, sentences, clauses or provisions contained in this Agreement is held by a court of competent jurisdiction to be invalid, illegal, unlawful, unenforceable or void in any respect, such shall not affect the remaining portions of this Agreement and the same shall remain in full force and effect as if such invalid, illegal, unlawful, unenforceable or void sections, paragraphs, sentences, clauses or provisions had never been included herein.
- 3.20 <u>Preparation of Agreement</u>. The parties acknowledge that they have sought and obtained whatever competent advice and counsel as was necessary for them to form a full and complete understanding of all rights and obligations herein and that the preparation of this Agreement has been their joint effort. The language agreed to herein expresses their mutual intent and the resulting document shall not, solely as a matter of judicial construction, be construed more severely against one of the parties than the other.
- 3.21 <u>Amendments</u>. No modification, amendment, or alteration in the terms or conditions contained herein shall be effective unless contained in a written document prepared with the same or similar formality as this Agreement and executed by each party hereto.
- 3.22 <u>Waiver</u>. The parties agree that each requirement, duty and obligation set forth herein is substantial and important to the formation of this Agreement and, therefore, is a material term hereof. Any party's failure to enforce any provision of this Agreement shall not be deemed a waiver of such provision or modification of this Agreement unless the waiver is in writing and signed by the party waiving such provision. A written waiver shall only be effective

as to the specific instance for which it is obtained and shall not be deemed a continuing or future waiver.

- 3.23 **Force Majeure**. Neither party shall be obligated to perform any duty, requirement or obligation under this Agreement if such performance is prevented by fire, hurricane, earthquake, explosion, wars, sabotage, accident, flood, acts of God, strikes, or other labor disputes, riot or civil commotions, or by reason of any other matter or condition beyond the control of either party, and which cannot be overcome by reasonable diligence and without unusual expense ("Force Majeure"). In no event shall a lack of funds on the part of either party be deemed Force Majeure.
- 3.24 <u>Survival</u>. All representations and warranties made herein, indemnification obligations, obligations to reimburse SBBC, obligations to maintain and allow inspection and audit of records and property, obligations to maintain the confidentiality of records, reporting requirements, and obligations to return public funds shall survive the termination of this Agreement.
- 3.25 <u>Contract Administration</u>: SBBC has delegated authority to the Superintendent of Schools or his/her designee to take any actions necessary to implement and administer this Agreement.
- 3.26 <u>Authority</u>. Each person signing this Agreement on behalf of either party individually warrants that he or she has full legal power to execute this Agreement on behalf of the party for whom he or she is signing, and to bind and obligate such party with respect to all provisions contained in this Agreement.
- 3.27 **FEMA Requirements.** The following requirements of this Article are only applicable to services to be performed concerning disaster recovery work.
 - A. Recovered/Recycled Materials. Offered products will make maximum use of recovered/recycled materials, but if not available, then materials used for the manufacture or construction of any supplies, materials or equipment covered by this Agreeement shall be new unless otherwise specified. Product(s) offered that have not been previously used in any way and are being actively marketed by the manufacturer will be accepted. Minor parts within the product(s) may have remanufactured components. If recovered/recycled materials are not available, the items bid must be new, the latest model, of the best quality and highest-grade workmanship as needed for the particular procurement. Note, however, that if a proposed recovered/recycled material has been determined to be unsafe for children in the schools, then only new material shall be used.
 - B. <u>Contract Work Hours and Safety Standards.</u> VENDOR will comply with 40 U.S.C. 3702 and 3704, as supplemented by Department of Labor regulations (29 C.F.R. Part 5).
 - C. <u>Energy Efficiency</u>. VENDOR must comply with the mandatory standards and policies relating to energy efficiency which are contained in the state energy conservation plan issued in compliance with the Energy Policy and Conservation Act.

- D. <u>Rights To Inventions Made Under A Contract Or Agreement</u>. Any invention developed during the performance of work under a federal award or grant shall provide for a nonexclusive, nontransferable, irrevocable, paid-up license to practice or have practiced for or on behalf of the United States the subject invention throughout the world.
- 3.28 **e-Builder.** The VENDOR shall use the Owner's Project Management software, e-Builder, as a conduit for all project management tasks, including, but not limited to: communications to, from and between Owner, VENDOR and contractors, CMs, Design Consultants and other vendors providing goods or services on projects; pay applications/invoicing; requests for change orders; material, equipment and systems submittals; requests for information; Architect's Supplemental Instructions; SMWBE Monthly Utilization Reporting; periodic Project observations; Weekly Progress Reports and meeting minutes.
- A. Licenses shall be provided to VENDOR to permit access and use of e- Builder for all projects awarded by Owner. Such licenses(s) shall be valid throughout the duration of the project(s).
- B. <u>Form Module</u>. The e-Builder Forms Module shall be used as the exclusive method to create Action Items that require a response from another Project Construction Team member. The required use of the Forms Module includes All e-mailed communications.
- C. <u>Work Flows</u>. Any and all responses or required responses to an open Action Item or to an initiated Work Flow process shall be input and managed through e-Builder. Work Flow processes that will be executed through e-Builder include but are not limited to those processes identified in Article 3.1.
- D. <u>Calendar Module</u>. The identification of Project events and required deliverables shall be input and maintained in the Calendar Module. At a minimum, such events include biweekly design meetings (while in design) weekly construction meetings, public meetings for the project (ex. Project Charter Meetings, etc.) and other design and/or construction milestones and deadlines.
- E. <u>Meetings</u>. Information to be input into e-Agenda related to any meeting includes, but is not limited to an agenda, a reminder of the meeting (which must occur a minimum of two (2) days prior to the meeting), meeting minutes (using the approved meeting minutes template) and confirmation of actual meeting attendees.
- F. <u>Access to e-Builder and Licensing</u>. VENDOR shall designate and identify the employee(s) that shall personally access e-Builder, the projects to which the employees(s) is assigned, and the employee(s)'s duties and responsibilities as it relates to e-Builder.
- G. This information together with a request for licensing shall be sent to Program Controls Support, telephone number (754) 321-1537, ebuilderLicense@browardschools.com. Upon receipt, review and acceptance of the request, access information and logins shall be proved to VENDOR.

H. and licenses b based on avail	Training shall be coordinated, scheduled and provided to those p by the Cost and Program Controls Manager. Additional training m	
	(Remainder of page intentionally left blank)	

IN WITNESS WHEREOF, the Parties hereto have made and executed this Agreement on the date first above written.

FOR S	<u>BBC</u>
(Corporate Seal)	THE SCHOOL BOARD OF BROWARD COUNTY, FLORIDA
ATTEST:	By Donna P. Korn, Chair
Robert W. Runcie, Superintendent of Schools	Approved as to Form and Legal Content:
	Office of the General Counsel

(Remainder of page intentionally left blank)

FOR VENDOR

(Corporate Seal)	
ATTEST:	[VENDOR'S FULL NAME]
	By
, Secretary	
Witness	
Witness	
	uired for Every Agreement Without Regard to Secretary's Attestation or Two (2) Witnesses.
STATE OF	
COUNTY OF	
presence or □ online notarization, this	nowledged before me, by means of □ physical day of, 202, by [PRINT FULL NAME OF FIRM] on behalf of the corporation
He/She is personally known to me or prodidentification and did/did not first take an o	oath. Type of Identification
My Commission Expires:	
	Signature – Notary Public
(SEAL)	Printed Name of Notary
	Notary's Commission No.

ATTACHMENT A

ATTACHMENT B

THE FOLLOWING VERSION OF THE SAMPLE PROGRAM MANAGER AGREEMENT (ATTACHMENT H REDLINED) IS FOR REFERENCE PURPOSES ONLY. IT CONTAINS THE TRACKED CHANGES FROM THE PREVIOUS UPLOADED VERSION.

AGREEMENT

THIS AGREEMENT is made and entered into as of this _____ day of ______.

2020, by and between

THE SCHOOL BOARD OF BROWARD COUNTY, FLORIDA

(hereinafter referred to as "SBBC"), a body corporate and political subdivision of the State of Florida, whose principal place of business is 600 Southeast Third Avenue, Fort Lauderdale, Florida 33301

and

[VENDOR'S FULL NAME]

(hereinafter referred to as "VENDOR"), whose principal place of business is [VENDOR's address]

WHEREAS, SBBC issued a Request for proposal identified as RFQ FY20-192 – Program Manager – Owner's Representative Services dated and amended herein referred to as "RFQ" all of which are incorporated by reference as if fully set forth herein for the purpose of receiving proposals for Program Manager – Owner's Representative Services; and

WHEREAS, VENDOR offered a proposal dated (hereinafter referred to as "Proposal" which is incorporated by reference herein) in response to this RFQ; and

WHEREAS, VENDOR is willing to provide Program Manager – Owner's Representative Services as more fully described in the Scope of Services of the RFQ.

NOW, THEREFORE, in consideration of the premises and of the mutual covenants contained herein and other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the Parties hereby agree as follows:

ARTICLE 1 - RECITALS

- 1.01 **Recitals**. The parties agree that the foregoing recitals are true and correct and that such recitals are incorporated herein by reference.
- 1.02 **FEMA Requirements.** NOTE: Articles within this Agreement which include **"FEMA Requirements"** in the title are only applicable to services to be performed concerning disaster recovery work.

ARTICLE 2 – SPECIAL CONDITIONS

2.01 <u>Term of Agreement</u>. Unless terminated earlier pursuant to Section 3.05 of this Agreement, the term of this Agreement shall be three (3) years commencing on **[commencement date]** and ending on **[expiration date]**. The term of the contract may, at the sole discretion of SBBC together with the agreement of VENDOR, be extended for two additional one-year

periods and, if needed, 180 days beyond the expiration date of the renewal period. Procurement & Warehousing Services Department, will, if considering renewing, request a letter to renew from VENDOR, prior to the end of the contract period. Any renewal period shall be approved by the Amendment to this Agreement.

2.02 Contract Documents and Priority of Documents. The Contract Documents shall consist of this Agreement for provision of Program Manager — Owner's Representative Services between the SBBC and VENDOR, RFQ FY20-192 inclusive of its attachments, and any addenda thereto, Statement of Qualifications submitted by VENDOR in response to RFQ FY20-192, Scope of Services contained in the RFQ FY20-192, Notice(s) to Proceed, Project Specific Drawings, Project Specific Specifications, current District Design Criteria, post-award meeting checklist and current District Master Specifications. The Contract Documents supersede all prior negotiations, correspondence, conversations, agreements, and understandings applicable to the matters contained herein and the parties agree that there are no commitments, agreements, or understandings concerning the subject matter of this Contract that are not contained in this document. Accordingly, the parties agree that no deviation from the terms hereof shall be predicated upon any prior representations or agreements, whether oral or written.

In the event of a conflict between Contract Documents, the following priority of documents shall govern.

A. This Agreement; then

- B. RFQ FY20-192 Program Manager Owner's Representative Services, inclusive of all its attachments, as may be amended via Addenda (the "RFQ"); then
- C. <u>Statement of Qualifications submitted in response to the RFQ by VENDOR.</u>

Cost of Services. SBBC shall pay VENDOR for services rendered under this Agreement and Fee Schedule on a time and material basis with a not to exceed amount for the term of the Agreement, as may be adjusted pursuant to the Performance-Based Fees provision set forth in Section 6.6 of the RFQ. If SBBC and VENDOR desire to renew this Agreement for one or two additional years beyond the original term, the Fee Schedule for years may be renegotiated with VENDOR. If there is no negotiation regarding the Fee Schedule, the Fees Schedule for years four and five shall remain the same as presented in this Agreement. Fee Schedule is defined as the maximum rate per hour or task for which services can be billed to SBBC. These listed fees are full compensation for VENDOR's direct personnel expenses, mandatory and customary contribution, overhead costs, project costs and profit. For all services rendered, the VENDOR will be paid upon actual hourly or salaried rates paid by the VENDOR, times the actual hours worked, times a fixed multiplier of. As payments for services are based upon actual salaries paid, SBBC shall not be billed for hours worked for salaried employees in excess of forty (40) hours per week. The VENDOR shall obtain prior approval, by SBBC, for all materials and reimbursables. All SBBC approved materials and reimbursables will be paid by SBBC at actual costs as a passthrough expense. Attachment A is only a reference document used for the estimate of the not to exceed amount. VENDOR shall obtain prior approval of any and all staff, including their hourly rate selected for commissioning services.

2.04 <u>Billing Instructions and Payment</u>. VENDOR shall be required to bill on not less than a monthly basis, and in no event shall VENDOR submit an invoice for services and costs more than ninety (90) days after the services and/or costs occurred, otherwise such services

Deleted: d

Deleted: Proposal

Deleted: of

Deleted: contract

Deleted: contract

Deleted:

and costs shall not be considered by SBBC and same services and costs are admitted by the VENDOR as waived and not due to the VENDOR. All costs shall be reasonable and subject to SBBC's approval.

Pursuant to Section 6.7.7 of the RFQ, during the initiation phase, as a condition precedent to payment, VENDOR shall timely submit a 12-month staffing plan, acceptable to SBBC, that correlates with the projected status of individual projects and the overall program over the 12-month period.

Pursuant to Section 6.7.7 of the RFQ, on a quarterly basis, as a condition precedent to payment, VENDOR shall timely submit a 12-month staffing plan that evaluates each team according to the projected status of individual projects and the overall program over the 12-month period.

On a monthly basis, as a condition precedent to payment, provide a written three (3) month look ahead document, acceptable to SBBC, which, among other things, lists the staffing plan (including the staff rates and other fees) for the three (3) months following the date of the three (3) month plan, and the month prior to the date of the three (3) month plan. SBBC, at its sole discretion shall approve the monthly written three (3) month look ahead document prior to its implementation.

2.05 Services - VENDOR shall provide Program Management Services pursuant to Section 6 of RFQ FY20-192, all addenda issued to the RFQ, if any, the VENDOR's response to the RFQ, and any SBBC ordering or authorizing documents such as the NTP are also made a part of this Agreement. VENDOR will attend meetings as requested by the Executive Director or his/her designee, Capital Programs. Meetings may be conducted during or after regular business hours. Examples of types of meetings which VENDOR may be requested to attend:

a. Senior Administration Meetings

- b. Community Meetings
- c. Project Review Meetings
- d. Construction Oversight Review Advisory Committee (CORC)
- e. Coordination Meeting with District Staff
- f. Local Governmental Agencies
- g. Operations/Budget Staff Meetings
- h. Capital Program Staff Meetings
- i. School Board Meetings

2.06 <u>M/WBE Participation</u>. VENDOR shall provide for M/WBE participation during its performance of services under this <u>Agreement</u> by listing SBBC Certified MWBEs' participation commitment percentage set forth in the VENDOR's proposal (Attachment B).

VENDOR shall obtain prior written approval from the Officer, Economic Development & Diversity Compliance Department for any replacement of any of the entities listed above. Utilizing any entity other than the ones listed, respectively will be considered a breach of this Agreement. VENDOR is subject to debarment and any other remedy available for any breaches to this Agreement.

Minority and/or Women-Owned Business Enterprise (M/WBE) Subcontracting Goals Program:

Deleted: ¶

Deleted: Additionally, VENDOR shall timely, o

Deleted: SBBC with

 $\label{eq:Deleted:VENDOR} \textbf{Deleted:}: VENDOR shall provide SBBC with services as proposed in its Proposal and in compliance with this Agreement and the RFQ and its Addenda. \P$

Deleted: contract

Deleted: .

Deleted:

RFQ FY20-192 Program Management Services - Owner's Representative

Page 3 of 20

SBBC has identified a **twenty-nine percent (29%) participation goal** for Submitting Firms that subcontract with eligible SBBC-Certified M/WBE firms for at least twenty-nine percent (29%) of the total contract value. See the definition of M/WBE in Section C.2. of Policy 3330. BIPM Comment – Should the reference Section read C.2. of the Policy?

Penalties and Sanctions

In the absence of a waiver granted by the SDOP or the self-performance of a portion or all of the SBE subcontracting goal by a certified SBE proposer, the failure of a proposer to attain a subcontracting goal for SBE participation in the performance of its contract or otherwise comply with the provisions of this API shall be considered a material breach of contract, grounds for termination of that contract with the SBBC, and shall be subject to any penalties and sanctions available under the terms of the SDOP policy, its contract terms with the SBBC, or by law pursuant to the penalties and sanctions set forth in Section 13 of the Standard Operating Procedures for this Policy.

Suspension BIPM Comment – For clarification – Since the Goal is specific to M/WBE, should M/WBE be used below. Or is "SBE" being used in general terms?

The temporary stoppage of a M/WBE firm's beneficial participation in the District's SDOP for a finite period of time due to cumulative contract payments the M/WBE firm received during a fiscal year that exceed a certain dollar threshold as set forth in Section (7) of the Standard Operating Procedures for this Policy or pursuant to the penalties and sanctions set forth in Section 13 of the Standard Operating Procedures for this Policy.

The SBBC Supplier Diversity Outreach Program works to increase the participation of small, minority and women-owned business enterprises in construction and purchasing contracts. It is the intent of the School Board of Broward County to have a diverse group of vendors to participate in the procurement process. The current list of SBBC S/M/WBE-Certified firms can be viewed at: browardschools.com/Page/46981.

- 2.07 <u>Inspection of VENDOR's Records by SBBC</u>. VENDOR shall establish and maintain books, records and documents (including electronic storage media) sufficient to reflect all income and expenditures of funds provided by SBBC under this Agreement. All VENDOR's Records, regardless of the form in which they are kept, shall be open to inspection and subject to audit, inspection, examination, evaluation and/or reproduction, during normal working hours, by SBBC's agent or its authorized representative to permit SBBC to evaluate, analyze and verify the satisfactory performance of the terms and conditions of this Agreement and to evaluate, analyze and verify any and all invoices, billings, payments and/or claims submitted by VENDOR or any of VENDOR's payees pursuant to this Agreement. VENDOR's Records subject to examination shall include, without limitation, those records necessary to evaluate and verify direct and indirect costs (including overhead allocations) as they may apply to costs associated with this Agreement. VENDOR's Records subject to this section shall include any and all documents pertinent to the evaluation, analysis, verification and reconciliation of any and all expenditures under this Agreement without regard to funding sources.
- (a) <u>VENDOR's Records Defined</u>. For the purposes of this Agreement, the term "VENDOR's Records" shall include, without limitation, and any supporting documents that would substantiate, reconcile or refute any charges and/or expenditures related to this Agreement.

- (b) <u>Duration of Right to Inspect</u>. For the purpose of such audits, inspections, examinations, evaluations and/or reproductions, SBBC's agent or authorized representative shall have access to VENDOR's Records from the effective date of this Agreement, for the duration of the term of this Agreement, and until the later of five (5) years after the termination of this Agreement or five (5) years after the date of final payment by SBBC to VENDOR pursuant to this Agreement.
- (c) <u>Notice of Inspection</u>. SBBC's agent or its authorized representative shall provide VENDOR reasonable advance notice (not to exceed two (2) weeks) of any intended audit, inspection, examination, evaluation and or reproduction.
- (d) <u>Audit Site Conditions</u>. SBBC's agent or its authorized representative shall have access to VENDOR's facilities and to any and all records related to this Agreement, and shall be provided adequate and appropriate work space in order to exercise the rights permitted under this section.
- (e) <u>Failure to Permit Inspection</u>. Failure by VENDOR to permit audit, inspection, examination, evaluation and/or reproduction as permitted under this Section shall constitute grounds for termination of this Agreement by SBBC for cause and shall be grounds for the denial of some or all of any VENDOR's claims for payment by SBBC.
- (f) Overcharges and Unauthorized Charges. If an audit conducted in accordance with this Section discloses overcharges or unauthorized charges to SBBC by VENDOR in excess of two percent (2%) of the total billings under this Agreement, the actual cost of SBBC's audit shall be paid by VENDOR. If the audit discloses billings or charges to which VENDOR is not contractually entitled, VENDOR shall pay said sum to SBBC within twenty (20) days of receipt of written demand under otherwise agreed to in writing by both parties.
- (g) <u>Inspection of Subcontractor's Records</u>. VENDOR shall require any and all subcontractors, insurance agents and material suppliers (hereafter referred to as "Payees") providing services or goods with regard to this Agreement to comply with the requirements of this section by insertion of such requirements in any written subcontract. Failure by VENDOR to include such requirements in any subcontract shall constitute grounds for termination of this Agreement by SBBC for cause and shall be grounds for the exclusion of some or all of any Payee's costs from amounts payable by SBBC to VENDOR pursuant to this Agreement and such excluded costs shall become the liability of VENDOR.
- (h) <u>Inspector General Audits</u>. VENDOR shall comply and cooperate immediately with any inspections, reviews, investigations, or audits deemed necessary by the Florida Office of the Inspector General or by any other state or federal officials.
 - 2.07.01 Quarterly Evaluation. An evaluation shall be conducted by RSM US LLP and the assigned District Staff on a quarterly basis to measure performance. Such evaluation shall be communicated to VENDOR to identify areas for betterment and shall be used as basis for continuous improvement.
- 2.08 Notice. When any of the parties desire to give notice to the other, such notice must be in writing, sent by U.S. Mail, postage prepaid, addressed to the party for whom it is intended at the place last specified; the place for giving notice shall remain such until it is changed by written notice in compliance with the provisions of this paragraph. For the present, the Parties designate the following as the respective places for giving notice:

Deleted: ¶

To SBBC: Superintendent of Schools

The School Board of Broward County, Florida

600 Southeast Third Avenue Fort Lauderdale, Florida 33301

With a Copy to: Director of Procurement and Warehousing Services

The School Board of Broward County, Florida 7720 West Oakland Park Blvd., Suite 323

Sunrise, Florida 33351

Executive Director, Capital Program

The School Board of Broward County, Florida

600 Southeast Third Avenue Fort Lauderdale, Florida 33301

To VENDOR:

Background Screening: VENDOR agrees to comply with all requirements of Sections 1012.32 and 1012.465, Florida Statutes, and all of its personnel who (1) are to be permitted access to school grounds when students are present, (2) will have direct contact with students, or (3) have access or control of school funds, will successfully complete the background screening required by the referenced statutes and meet the standards established by the statutes. This background screening will be conducted by SBBC in advance of VENDOR or its personnel providing any services under the conditions described in the previous sentence. VENDOR shall bear the cost of acquiring the background screening required by Section 1012.32, Florida Statutes, and any fee imposed by the Florida Department of Law Enforcement to maintain the fingerprints provided with respect to VENDOR and its personnel. The parties agree that the failure of VENDOR to perform any of the duties described in this section shall constitute a material breach of this Agreement entitling SBBC to terminate immediately with no further responsibilities or duties to perform under this Agreement. VENDOR agrees to indemnify and hold harmless SBBC, its officers and employees from any liability in the form of physical or mental injury, death or property damage resulting from VENDOR failure to comply with the requirements of this Section or with Sections 1012.32 and 1012.465, Florida Statutes.

- 2.10 <u>Liability</u>. This section shall survive the termination of all performance or obligations under this Agreement and shall be fully binding until such time as any proceeding brought on account of this Agreement is barred by any applicable statute of limitations.
 - A. By SBBC: SBBC agrees to be fully responsible up to the limits of Section 768.28, Florida Statutes, for its acts of negligence, or its employees' acts of negligence when acting within the scope of their employment and agrees to be liable for any damages resulting from said negligence.
 - B. By VENDOR: VENDOR agrees to indemnify and hold harmless SBBC, its agents, servants and employees from any and all claims, judgments, costs, and expenses including, but not limited to, reasonable attorney's fees, reasonable investigative and discovery costs, court costs and all other sums which SBBC, its agents, servants and employees may pay or become obligated to pay on account of any, all and every claim or demand, or assertion of liability, or any claim or action founded thereon,

Deleted: ¶
¶
¶
.¶

arising out of the products, goods or services negligently furnished by VENDOR, its agents, servants or employees; the equipment of VENDOR, its agents, servants or employees while such equipment is on premises owned or controlled by SBBC; or the negligence of VENDOR or the negligence of VENDOR's agents when acting within the scope of their employment, whether such claims, judgments, costs and expenses be for damages, damage to property including SBBC's property, and injury or death of any person whether employed by VENDOR, SBBC or otherwise. This indemnification provision does not extend to require indemnification to the SBBC for its own negligence and, therefore, certain requirements of Section 725.06, Florida Statutes, are not required. However, should this clause be interpreted by a court to require compliance with other provisions of Section 725.06, Florida Statutes, then the following shall be incorporated by reference into the Agreement: (i) the indemnification as provided in this paragraph shall be subject to a monetary limitation equal to Six Hundred Million Dollars (\$600,000,000.00), or the total value (including fees and cost) of all projects assigned to VENDOR, whichever is less, and (ii) the SBBC and VENDOR both acknowledge that the monetary limitation referenced above bears a reasonable commercial relationship to this Agreement.

- 2.11 <u>Insurance</u>. Professional Liability (Errors and Omissions): The VENDOR shall procure a Professional Liability Insurance Policy with coverage of not less than Five Million (\$5,000,000) Dollars and a deductible, to be paid by VENDOR, of not more than Fifty Thousand (\$50,000) Dollars, providing for all sums which the VENDOR shall be legally obligated to pay as damages for claims arising out of or relating to the Work performed by the VENDOR or any person employed or acting on the VENDOR's behalf, but not limited to Subconsultants/Subcontractors, in connection with this Agreement. The insurance policy, or a policy with the same terms and conditions, shall remain in full force and effect during the project and for a period of time terminating three (3) years after final completion of the project. If the Owner, at its sole discretion, agrees that such coverage is not commercially reasonably available, the Owner may authorize the VENDOR to alter the coverage by substituting a lower aggregate or changing any other terms and conditions of the coverage, including but not limited to deductible amounts, based upon the scope of the Project. Notwithstanding any provision in this Agreement, nothing in this Agreement shall be construed as a limitation of the VENDOR's liability to SBBC.
- A. <u>Worker's Compensation Insurance.</u> The VENDOR shall maintain Worker's Compensation Insurance in accordance with Florida Statutory Limits and Employers Liability Insurance with a limit of not less than Five Hundred Thousand \$500,000/\$500,000/\$500,000 (each accident/disease-each employee/disease-policy limit).
- B. <u>Commercial General Liability Insurance</u>. The VENDOR shall maintain Comprehensive General Liability Insurance, including Products & Completed Operations, Personal and Bodily Injury, and Contractual Liability, to cover the indemnification language set forth herein. Limits shall not be less than \$1,000,000 per occurrence; \$2,000,000 general aggregate. Products Completed Operations aggregate shall not be less than \$2,000,000. The School Board of Broward County must be named as an Additional Insured on the Certificate along with the RFQ Number.
- C. <u>Automobile Liability Insurance</u>. The VENDOR shall maintain Automobile Liability Insurance covering all Owned, Non-Owned and Hired vehicles in the amount of not less than One Million Dollars (\$1,000,000) per occurrence Combined Single Limit for Bodily Injury and Property Damage and shall not be any less restrictive than the standard ISO Business

Auto Policy CA 00 01. In the event the VENDOR does not own any vehicles, VENDOR shall obtain hired and non-owned coverage for \$1,000,000 Combined Single Limits and provide an affidavit signed by the VENDOR indicating the following:

(Insured) does not own any vehicles. In the event we acquire any vehicles throughout the term of this Agreement, Insured agrees to provide of "Any Auto" coverage effective date of acquisition.

D. <u>Certificate of Insurance Requirements</u>. Prior to the commencement of any Work, as evidence of required coverage, VENDOR must provide a Certificate of Insurance to The School Board of Broward County, Florida's Risk Management Department for approval. Certificates must be faxed to the School Board of Broward County's Certificate Tracking System at 1-866-897-0425.

Liability Policies are to contain the following provisions. <u>In addition, the following</u> wording must be included on the Certificate of Insurance:

The School Board of Broward County, Florida, its members, officers,
employees, and agents are added as additional insured. The
endorsement # is

All liability policies are primary of all other valid and collectable coverage maintained by The School Board of Broward County, Florida.

Contractual liability is included in the General Liability policy.

2.12 Nonconformance to Contract Conditions. Services offered must comply with RFQ conditions and specifications, specifically including but not limited to Sections 6.2.1 and 6.3.4 of the RFQ, and any resulting Agreement at all times. Service(s) not conforming to RFQ conditions, specifications or time frames will have a written nonconformance letter sent to the vendor to provide an opportunity to remedy the nonconformance. Failure of the VENDOR to remedy the nonconformance, within thirty (30) days from receipt of notification, shall result in termination of contract. Additionally, for a period of two (2) years, any RFQ submitted by VENDOR will not be considered or recommended for award, as such, all departments being advised not to do business with VENDOR.

Key Personnel* indicated in the PM's Qualifications Statement are expected to actually perform services in this program. NOTE: The substitution of Key Personnel listed in the Submitting Firm's Qualifications Statement without the prior written approval by the Superintendent of Schools or his/her designee, shall be a material breach of this Agreement. The substitution of the Principal in Charge and/or Program Director (a/k/a PM Lead) listed in the Submitting Firm's Qualifications Statement without the prior written approval by The School Board of Broward County, Florida, shall be a material breach of the Agreement.

2.13 <u>Travel</u>. Local travel will be billed as a reimbursable expense in accordance with the current or updated School Board Policy 3400 and any other relevant School Board Policies. Out of county travel and per diem may be allowable at the sole discretion of SBBC. SBBC has delegated authority to the Superintendent of Schools or his/her designee to provide prior approval to VENDOR for any and all travel and per diem. Should any out of county travel

Deleted: Executive Director, Capital Programs

Deleted: the Program Manager – Owner's Representative Services this

Deleted: T

Deleted: Program Manager - Owner's Representative Services

and/or per diem be allowed, then it shall be billed and reimbursed in compliance with the current or updated School Board Policy 3400 and/or other relevant School Board Policies.

PLEASE BE ADVISED THAT SCHOOL BOARD POLICY 3400, UNDER THE "ALL TRAVEL" SECTION PARAGRAPH 2 STATES "NEITHER SUBSISTENCE NOR PER DIEM SHALL BE ALLOWED FOR ANY CLASS OF TRAVEL IN BROWARD COUNTY EXCEPT FOR AUTHORIZED BUSINESS OR WORKSHOPS OCCURING BEFORE 6:00 A.M. OR EXTENDING BEYOND 8:00 PM." Additionally, the travel requires prior approval.

2.14 **Working Environment**. SBBC will provide a functional working environment within an existing school to include, but not limited to, office space, office furniture and appliances, housekeeping, internet service and office network cabling, utilities, use of SBBC's email service, server(s) for file storage, office supplies, and in-house printing, copying, plotting and fax equipment services. VENDOR shall obtain prior approval from SBBC for any additional goods or services required for the working environment.

ARTICLE 3 – GENERAL CONDITIONS

- 3.01 No Waiver of Sovereign Immunity. Nothing herein is intended to serve as a waiver of sovereign immunity by any agency or political subdivision to which sovereign immunity may be applicable or of any rights or limits to liability existing under Section 768.28, Florida Statutes. This section shall survive the termination of all performance or obligations under this Agreement and shall be fully binding until such time as any proceeding brought on account of this Agreement is barred by any applicable statute of limitations.
- 3.02 No Third Party Beneficiaries. The parties expressly acknowledge that it is not their intent to create or confer any rights or obligations in or upon any third person or entity under this Agreement. None of the parties intend to directly or substantially benefit a third party by this Agreement. The parties agree that there are no third party beneficiaries to this Agreement and that no third party shall be entitled to assert a claim against any of the parties based upon this Agreement. Nothing herein shall be construed as consent by an agency or political subdivision of the State of Florida to be sued by third parties in any matter arising out of any contract.
- 3.03 <u>Independent Contractor</u>. The parties to this agreement shall at all times be acting in the capacity of independent contractors and not as an officer, employee or agent of one another. Neither party or its respective agents, employees, subcontractors or assignees shall represent to others that it has the authority to bind the other party unless specifically authorized in writing to do so. No right to SBBC retirement, leave benefits or any other benefits of SBBC employees shall exist as a result of the performance of any duties or responsibilities under this Agreement. SBBC shall not be responsible for social security, withholding taxes, contributions to unemployment compensation funds or insurance for the other party or the other party's officers, employees, agents, subcontractors or assignees.
- 3.04 **Equal Opportunity Provision**. The parties agree that no person shall be subjected to discrimination because of age, race, color, disability, gender identity, gender expression marital status, national origin, religion, sex or sexual orientation in the performance of the parties' respective duties, responsibilities and obligations under this Agreement.

3.04.01 **FEMA Requirement - Equal Employment Opportunity.**

The following language must be included in any contract involving any grant, contract, loan, insurance, or guarantee involving federally assisted construction which is not exempt from the requirements of the equal opportunity clause. The term "contractor" is used to refer to both the VENDOR and contractors performing work.

The following is hereby incorporated into this an any other agreement for construction work, or modification thereof, as defined in the regulations of the Secretary of Labor at 41 CFR Chapter 60, which is paid for in whole or in part with funds obtained from the Federal Government or borrowed on the credit of the Federal Government pursuant to a grant, contract, loan, insurance, or guarantee, or undertaken pursuant to any Federal program involving such grant, contract, loan, insurance, or guarantee, the following equal opportunity clause: During the performance of this contract, the contractor agrees as follows:

- (1) The contractor will not discriminate against any employee or applicant for employment because of race, color, religion, sex, sexual orientation, gender identity, or national origin. The contractor will take affirmative action to ensure that applicants are employed, and that employees are treated during employment without regard to their race, color, religion, sex, sexual orientation, gender identity, or national origin. Such action shall include, but not be limited to the following: Employment, upgrading, demotion, or transfer; recruitment or recruitment advertising; layoff or termination; rates of pay or other forms of compensation; and selection for training, including apprenticeship. The contractor agrees to post in conspicuous places, available to employees and applicants for employment, notices to be provided setting forth the provisions of this nondiscrimination clause.
- (2) The contractor will, in all solicitations or advertisements for employees placed by or on behalf of the contractor, state that all qualified applicants will receive consideration for employment without regard to race, color, religion, sex, sexual orientation, gender identity, or national origin.
- (3) The contractor will not discharge or in any other manner discriminate against any employee or applicant for employment because such employee or applicant has inquired about, discussed, or disclosed the compensation of the employee or applicant or another employee or applicant. This provision shall not apply to instances in which an employee who has access to the compensation information of other employees or applicants as a part of such employee's essential job functions discloses the compensation of such other employees or applicants to individuals who do not otherwise have access to such information, unless such disclosure is in response to a formal complaint or charge, in furtherance of an investigation, proceeding, hearing, or action, including an investigation conducted by the employer, or is consistent with the contractor's legal duty to furnish information.
- (4) The contractor will send to each labor union or representative of workers with which he has a collective bargaining agreement or other contract or understanding, a notice to be provided advising the said labor union or workers' representatives of the contractor's commitments under this section, and shall post

copies of the notice in conspicuous places available to employees and applicants for employment.

- (5) The contractor will comply with all provisions of Executive Order 11246 of September 24, 1965, and of the rules, regulations, and relevant orders of the Secretary of Labor.
- (6) The contractor will furnish all information and reports required by Executive Order 11246 of September 24, 1965, and by rules, regulations, and orders of the Secretary of Labor, or pursuant thereto, and will permit access to his books, records, and accounts by the administering agency and the Secretary of Labor for purposes of investigation to ascertain compliance with such rules, regulations, and orders.
- (7) In the event of the contractor's noncompliance with the nondiscrimination clauses of this contract or with any of the said rules, regulations, or orders, this contract may be canceled, terminated, or suspended in whole or in part and the contractor may be declared ineligible for further Government contracts or federally assisted construction contracts in accordance with procedures authorized in Executive Order 11246 of September 24, 1965, and such other sanctions may be imposed and remedies invoked as provided in Executive Order 11246 of September 24, 1965, or by rule, regulation, or order of the Secretary of Labor, or as otherwise provided by law.
- (8) The contractor will include the portion of the sentence immediately preceding paragraph (1) and the provisions of paragraphs (1) through (8) in every subcontractor purchase order unless exempted by rules, regulations, or orders of the Secretary of Labor issued pursuant to section 204 of Executive Order 11246 of September 24, 1965, so that such provisions will be binding upon each subcontractor or vendor. The contractor will take such action with respect to any subcontract or purchase order as the administering agency may direct as a means of enforcing such provisions, including sanctions for noncompliance:

Provided, however, that in the event a contractor becomes involved in, or is threatened with, litigation with a subcontractor or vendor as a result of such direction by the administering agency, the contractor may request the United States to enter into such litigation to protect the interests of the United States. The applicant further agrees that it will be bound by the above equal opportunity clause with respect to its own employment practices when it participates in federally assisted construction work: Provided, That if the applicant so participating is a State or local government, the above equal opportunity clause is not applicable to any agency, instrumentality or subdivision of such government which does not participate in work on or under the contract.

The applicant [School Board] and VENDOR agree that to assist and cooperate actively with the administering agency and the Secretary of Labor in obtaining the compliance of contractors and subcontractors with the equal opportunity clause and the rules, regulations, and relevant orders of the Secretary of Labor,

that it will furnish the administering agency and the Secretary of Labor such information as they may require for the supervision of such compliance, and that it will otherwise assist the administering agency in the discharge of the agency's primary responsibility for securing compliance.

- 3.05 **Termination**. This Agreement may be canceled with or without cause by SBBC during the term hereof upon thirty (30) days written notice to the other parties of its desire to terminate this Agreement. SBBC shall have no liability for any property left on SBBC's property by any party to this Agreement after the termination of this Agreement. Any party contracting with SBBC under this Agreement agrees that any of its property placed upon SBBC's facilities pursuant to this Agreement shall be removed within ten (10) business days following the termination, conclusion or cancellation of this Agreement and that any such property remaining upon SBBC's facilities after that time shall be deemed to be abandoned, title to such property shall pass to SBBC, and SBBC may use or dispose of such property as SBBC deems fit and appropriate. VENDOR will be compensated for services performed up to the date of termination and be reimbursed for any and all reimbursable expenses allowed under the Agreement.
- 3.06 <u>Default</u>. The parties agree that, in the event that either party is in default of its obligations under this Agreement, the non-defaulting party shall provide to the defaulting party (30) days written notice to cure the default. However, in the event said default cannot be cured within said thirty (30) day period and the defaulting party is diligently attempting in good faith to cure same, the time period shall be reasonably extended to allow the defaulting party additional cure time. Upon the occurrence of a default that is not cured during the applicable cure period, this Agreement may be terminated by the non-defaulting party upon thirty (30) days notice. This remedy is not intended to be exclusive of any other remedy, and each and every such remedy shall be cumulative and shall be in addition to every other remedy now or hereafter existing at law or in equity or by statute or otherwise. No single or partial exercise by any party of any right, power, or remedy hereunder shall preclude any other or future exercise thereof. Nothing in this section shall be construed to preclude termination for convenience pursuant to Section 3.05.
- 3.07 Annual Appropriation. The performance and obligations of SBBC under this Agreement shall be contingent upon an annual budgetary appropriation by its governing body. If SBBC does not allocate funds for the payment of services or products to be provided under this Agreement, this Agreement may be terminated by SBBC at the end of the period for which funds have been allocated. SBBC shall notify the other party at the earliest possible time before such termination. No penalty shall accrue to SBBC in the event this provision is exercised, and SBBC shall not be obligated or liable for any future payments due or any damages as a result of termination under this section.
- 3.08 Excess Funds. Any party receiving funds paid by SBBC under this Agreement agrees to promptly notify SBBC of any funds erroneously received from SBBC upon the discovery of such erroneous payment or overpayment. Any such excess funds shall be refunded to SBBC with interest calculated from the date of the erroneous payment or overpayment. Interest shall be calculated using the interest rate for judgments under Section 55.03, Florida Statutes, applicable at the time the erroneous payment or overpayment was made by SBBC.
- 3.09 **Public Records**. Pursuant to Section 119.0701, Florida Statutes, any party contracting with SBBC is required to (a) keep and maintain available for public inspection any

records that pertain to services rendered under this Agreement; (b) provide the public with access to public records on the same terms and conditions that SBBC would provide such records and at a cost that does not exceed the cost provided in Chapter 119. Florida Statutes or as otherwise provided by law; (c) ensure that public records that are exempt or confidential and exempt from public records disclosure requirements are not disclosed except as authorized by law; and (d) meet all requirements for retaining public records and transfer, at no cost to SBBC, all public records in that party's possession upon termination of its Agreement with SBBC and destroy any duplicate public records that are exempt or confidential and exempt from public records disclosure requirements. All of such party's records stored electronically must be provided to SBBC in a format that is compatible with SBBC's information technology systems. Each party shall maintain its own respective records and documents associated with this Agreement in accordance with the records retention requirements applicable to public records. Each party shall be responsible for compliance with any public documents request served upon it pursuant to Section 119.07, Florida Statutes, and any resultant award of attorney's fees for non-compliance with that law. Each party acknowledges that this Agreement and all attachments thereto are public records and do not constitute trade secrets.

- 3.10 Student Records: Notwithstanding any provision to the contrary within this Agreement, any party contracting with SBBC under this Agreement shall fully comply with the requirements of Sections 1002.22 and 1002.221, Florida Statutes; FERPA, and any other state or federal law or regulation regarding the confidentiality of student information and records. Each such party agrees, for itself, its officers, employees, agents, representatives, contractors or subcontractors, to fully indemnify and hold harmless SBBC and its officers and employees for any violation of this section, including, without limitation, defending SBBC and its officers and employees against any complaint, administrative or judicial proceeding, payment of any penalty imposed upon SBBC, or payment of any and all costs, damages, judgments or losses incurred by or imposed upon SBBC arising out of a breach of this covenant by the party, or an officer, employee, agent, representative, contractor of the party to the extent that the party or an officer, employee, agent, representative, contractor, or sub-contractor of the party shall either intentionally or negligently violate the provisions of this section or of Sections 1002.22 and/or 1002.221, Florida Statutes.
- 3.11 Compliance with Laws. Each party shall comply with all applicable federal and state laws, codes, rules and regulations in performing its duties, responsibilities and obligations pursuant to this Agreement.
- 3.12 <u>Place of Performance</u>. All obligations of SBBC under the terms of this Agreement are reasonably susceptible of being performed in Broward County, Florida and shall be payable and performable in Broward County, Florida.
- 3.13 Governing Law and Venue. This Agreement shall be interpreted and construed in accordance with and governed by the laws of the State of Florida. Any controversies or legal problems arising out of this Agreement and any action involving the enforcement or interpretation of any rights hereunder shall be submitted to the jurisdiction of the State courts of the Seventeenth Judicial Circuit of Broward County, Florida.
- 3.14 Entirety of Agreement. This document incorporates and includes all prior negotiations, correspondence, conversations, agreements and understandings applicable to the matters contained herein and the parties agree that there are no commitments, agreements or understandings concerning the subject matter of this Agreement that are not contained in this document. Accordingly, the parties agree that no deviation from the terms hereof shall be predicated upon any prior representations or agreements, whether oral or written.

- 3.15 <u>Binding Effect</u>. This Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns.
- 3.16 <u>Assignment</u>. Neither this Agreement nor any interest herein may be assigned, transferred or encumbered by any party without the prior written consent of the other party. There shall be no partial assignments of this Agreement including, without limitation, the partial assignment of any right to receive payments from SBBC.
- 3.17 <u>Incorporation by Reference</u>. Exhibits attached hereto **Attachment A** and referenced herein shall be deemed to be incorporated into this Agreement by reference.
- 3.18 <u>Captions</u>. The captions, section designations, section numbers, article numbers, titles and headings appearing in this Agreement are inserted only as a matter of convenience, have no substantive meaning, and in no way define, limit, construe or describe the scope or intent of such articles or sections of this Agreement, nor in any way effect this Agreement and shall not be construed to create a conflict with the provisions of this Agreement.
- 3.19 <u>Severability</u>. In the event that any one or more of the sections, paragraphs, sentences, clauses or provisions contained in this Agreement is held by a court of competent jurisdiction to be invalid, illegal, unlawful, unenforceable or void in any respect, such shall not affect the remaining portions of this Agreement and the same shall remain in full force and effect as if such invalid, illegal, unlawful, unenforceable or void sections, paragraphs, sentences, clauses or provisions had never been included herein.
- 3.20 <u>Preparation of Agreement</u>. The parties acknowledge that they have sought and obtained whatever competent advice and counsel as was necessary for them to form a full and complete understanding of all rights and obligations herein and that the preparation of this Agreement has been their joint effort. The language agreed to herein expresses their mutual intent and the resulting document shall not, solely as a matter of judicial construction, be construed more severely against one of the parties than the other.
- 3.21 <u>Amendments</u>. No modification, amendment, or alteration in the terms or conditions contained herein shall be effective unless contained in a written document prepared with the same or similar formality as this Agreement and executed by each party hereto.
- 3.22 **Waiver.** The parties agree that each requirement, duty and obligation set forth herein is substantial and important to the formation of this Agreement and, therefore, is a material term hereof. Any party's failure to enforce any provision of this Agreement shall not be deemed a waiver of such provision or modification of this Agreement unless the waiver is in writing and signed by the party waiving such provision. A written waiver shall only be effective as to the specific instance for which it is obtained and shall not be deemed a continuing or future waiver.
- 3.23 Force Majeure. Neither party shall be obligated to perform any duty, requirement or obligation under this Agreement if such performance is prevented by fire, hurricane, earthquake, explosion, wars, sabotage, accident, flood, acts of God, strikes, or other labor disputes, riot or civil commotions, or by reason of any other matter or condition beyond the control of either party, and which cannot be overcome by reasonable diligence and without unusual expense ("Force Majeure"). In no event shall a lack of funds on the part of either party be deemed Force Majeure.

- 3.24 <u>Survival</u>. All representations and warranties made herein, indemnification obligations, obligations to reimburse SBBC, obligations to maintain and allow inspection and audit of records and property, obligations to maintain the confidentiality of records, reporting requirements, and obligations to return public funds shall survive the termination of this Agreement.
- 3.25 <u>Contract Administration</u>: SBBC has delegated authority to the Superintendent of Schools or his/her designee to take any actions necessary to implement and administer this Agreement.
- 3.26 <u>Authority</u>. Each person signing this Agreement on behalf of either party individually warrants that he or she has full legal power to execute this Agreement on behalf of the party for whom he or she is signing, and to bind and obligate such party with respect to all provisions contained in this Agreement.
- 3.27 **FEMA Requirements.** The following requirements of this Article are only applicable to services to be performed concerning disaster recovery work.
 - A. <u>Recovered/Recycled Materials</u>. Offered products will make maximum use of recovered/recycled materials, but if not available, then materials used for the manufacture or construction of any supplies, materials or equipment covered by this <u>Agreeement</u> shall be new unless otherwise specified. Product(s) offered that have not been previously used in any way and are being actively marketed by the manufacturer will be accepted. Minor parts within the product(s) may have remanufactured components. If recovered/recycled materials are not available, the items bid must be new, the latest model, of the best quality and highest-grade workmanship as needed for the particular procurement. Note, however, that if a proposed recovered/recycled material has been determined to be unsafe for children in the schools, then only new material shall be used.
 - B. <u>Contract Work Hours and Safety Standards.</u> VENDOR will comply with 40 U.S.C. 3702 and 3704, as supplemented by Department of Labor regulations (29 C.F.R. Part 5).
 - C. <u>Energy Efficiency</u>. VENDOR must comply with the mandatory standards and policies relating to energy efficiency which are contained in the state energy conservation plan issued in compliance with the Energy Policy and Conservation Act.
 - D. <u>Rights To Inventions Made Under A Contract Or Agreement.</u> Any invention developed during the performance of work under a federal award or grant shall provide for a nonexclusive, nontransferable, irrevocable, paid-up license to practice or have practiced for or on behalf of the United States the subject invention throughout the world.
- 3.28 **e-Builder.** The VENDOR shall use the Owner's Project Management software, e-Builder, as a conduit for all project management tasks, including, but not limited to: communications to, from and between Owner, VENDOR and contractors, CMs, Design Consultants and other vendors providing goods or services on projects; pay applications/invoicing; requests for change orders; material, equipment and systems submittals;

Deleted:

Deleted: RFP

requests for information; Architect's Supplemental Instructions; SMWBE Monthly Utilization Reporting; periodic Project observations; Weekly Progress Reports and meeting minutes.

- A. Licenses shall be provided to VENDOR to permit access and use of e-Builder for all projects awarded by Owner. Such licenses(s) shall be valid throughout the duration of the project(s).
- B. <u>Form Module</u>. The e-Builder Forms Module shall be used as the exclusive method to create Action Items that require a response from another Project Construction Team member. The required use of the Forms Module includes All e-mailed communications.
- C. <u>Work Flows</u>. Any and all responses or required responses to an open Action Item or to an initiated Work Flow process shall be input and managed through e-Builder. Work Flow processes that will be executed through e-Builder include but are not limited to those processes identified in Article 3.1.
- D. <u>Calendar Module</u>. The identification of Project events and required deliverables shall be input and maintained in the Calendar Module. At a minimum, such events include biweekly design meetings (while in design) weekly construction meetings, public meetings for the project (ex. Project Charter Meetings, etc.) and other design and/or construction milestones and deadlines.
- E. <u>Meetings</u>. Information to be input into e-Agenda related to any meeting includes, but is not limited to an agenda, a reminder of the meeting (which must occur a minimum of two (2) days prior to the meeting), meeting minutes (using the approved meeting minutes template) and confirmation of actual meeting attendees.
- F. <u>Access to e-Builder and Licensing</u>. VENDOR shall designate and identify the employee(s) that shall personally access e-Builder, the projects to which the employees(s) is assigned, and the employee(s)'s duties and responsibilities as it relates to e-Builder.
- G. This information together with a request for licensing shall be sent to <u>Program</u> Controls Support, telephone number (754) 321-1537, <u>ebuilderLicense@browardschools.com</u>. Upon receipt, review and acceptance of the request, access information and logins shall be proved to VENDOR.
- <u>H.</u> Training shall be coordinated, scheduled and provided to those provided access and licenses by the Cost and Program Controls Manager, Additional training may be provided based on availability.

Deleted: Atkins,

Deleted: Atkins

(Remainder of page intentionally left blank)

Page 16 of 20

IN WITNESS WHEREOF, the Parties hereto have made and executed this Agreement on the date first above written.

FOR SBBC

(Corporate Seal) THE SCHOOL BOARD OF BROWARD COUNTY, FLORIDA

ATTEST: Donna P. Korn, Chair

Robert W. Runcie, Superintendent of Schools

Approved as to Form and Legal Content:

Office of the General Counsel

(Remainder of page intentionally left blank)

 Deleted: ¶
Page Break

FOR VENDOR

(Corporate Seal)		
ATTEST:	[VENDOR'S FULL NAME]	
	By	
, Secretary		
Witness		
Witness		
	nired for Every Agreement Without Regard Secretary's Attestation or Two (2) Witness	
STATE OF		
COUNTY OF		
presence or □ online notarization, this	owledged before me, by means of □ physical _ day of, 202, by [PFFULL NAME OF FIRM] on behalf of the contraction.	RINT
	iced	_as
identification and did/did not first take an o	eath. Type of Identification	
My Commission Expires:		
	Signature – Notary Public	
(SEAL)	Printed Name of Notary	-
	Notary's Commission No.	-

ATTACHMENT A

ATTACHMENT B

Project ID	Project Name	P	TOTAL PROJECTED BUDGET	Current Phase (FEB 2020)	Phase Percent Complete
Phase 2 - Pre-De	l ocign		DODGET		Complete
P.001676		Τċ	F 061 766	2 Dro Dosign	95%
P.001676	Cresthaven ES SMART Program Renovations (CMAR)	\$	5,961,766	2-Pre-Design	95%
P.002134	Deerfield Beach HS Phase 2 SMART Program Renovations (CMAR)	\$	5 281 200	2-Pre-Design	95%
P.002134	Deerfield Beach MS SMART Program Renovations (CMAR)	\$		2-Pre-Design	95%
P.002142	Lyons Creek MS SMART Program Renovations (CMAR)	\$		2-Pre-Design	95%
P.002148	Monarch HS SMART Program Renovations (CMAR)	\$		2-Pre-Design	95%
P.002143	New Renaissance MS SMART Program Renovations (CMAR)	\$		2-Pre-Design	95%
P.002128	Sheridan Technical HS SMART Program Renovations (CMAR)	\$		2-Pre-Design	95%
P.002144	Silver Lakes MS SMART Program Renovations (CMAR)	\$		2-Pre-Design	95%
Phase 3 - Design		1 *	_,	1 - 1 - 2 - 20 - 8 - 1	1 2271
P.002110	Apollo MS SMART Program Renovations	Τć	6.015.000	2 Dosign/Dormit	96%
P.002110	Apollo MS SMART Program Renovations	\$	6,915,000	3-Design/Permit	96%
P.000415	Atlantic Tachnical HS & Tachnical Callage SMART Renovations	ے ا	12 005 200	2 Dosign/Dormit	0.49/
P.000415 P.001796	Atlantic Technical HS & Technical College SMART Renovations Atlantic West ES SMART Program Renovations	\$ \$		3-Design/Permit 3-Design/Permit	94% 95%
P.001796 P.001686	Attucks MS Phase 1 SMART Program Renovations	\$		3-Design/Permit	98%
P.001080	Bennett ES SMART Program Renovations	\$		3-Design/Permit	88%
P.002083 P.001818		\$		-	95%
P.001818 P.001848	Charles Drew ERC SMART Program Renovations	\$		3-Design/Permit	
P.001848 P.001753	Charles Drew FRC SMART Program Renovations	\$		3-Design/Permit	95% 91%
	Coconut Creek HS SMART Program Renovations			3-Design/Permit	
P.002088	Coconut Palm ES SMART Program Renovations	\$		3-Design/Permit	99%
P.002133	Cooper City HS SMART Program Renovations (CMAR)	\$		3-Design/Permit	84%
P.002080	Coral Glades HS SMART Program Renovations	\$		3-Design/Permit	95%
P.002045	Coral Park ES SMART Program Renovations	\$		3-Design/Permit	96%
P.001979	Coral Springs MS SMART Program Renovations	\$		3-Design/Permit	95%
P.001982	Coral Springs PK8 Phase 2 SMART Program Renovations	\$		3-Design/Permit	95%
P.002063	Country Hills ES SMART Program Renovations	\$		3-Design/Permit	95%
P.002086	Croissant Park ES SMART Program Renovations	\$		3-Design/Permit	96%
P.000816	Crystal Lake MS SMART Program Renovations	\$		3-Design/Permit	97%
P.002061	Dania ES SMART Program Renovations	\$		3-Design/Permit	55%
P.002064	Driftwood ES SMART Program Renovations	\$		3-Design/Permit	88%
P.002115	Hallandale Magnet HS Phase 2 SMART Program Renovations	\$		3-Design/Permit	96%
P.001845	Hollywood Hills ES SMART Program Renovations	\$		3-Design/Permit	40%
P.001980	Indian Trace ES SMART Program Renovations	\$		3-Design/Permit	35%
P.002059	James S. Hunt ES SMART Program Renovations	\$		3-Design/Permit	95%
P.001801	Lauderhill 6-12 SMART Program Renovations	\$	7,841,280	3-Design/Permit	96%
	Lauderhill Comm School at Park Lakes LC (fka Castle Hill Annex)				
P.002092	SMART Program Renovations (CC-CMAR)	\$		3-Design/Permit	97%
P.001824	Lloyd Estates ES SMART Program Renovations	\$		3-Design/Permit	98%
P.001647-NCA	Margate ES Phase 2 SMART Program Renovations	_	in Ph 1	3-Design/Permit	50%
P.001836	Margate MS SMART Program Renovations	\$	11,054,080	3-Design/Permit	99%
		١.			
P.000817	Marjory Stoneman Douglas HS SMART Program Renovations (CMAR)	\$		3-Design/Permit	96%
P.002125	Mary M. Bethune ES SMART Program Renovations	\$		3-Design/Permit	35%
P.001954	McArthur HS Phase 1 SMART Program Renovations (CMAR)	\$		3-Design/Permit	85%
P.001954-1	McArthur HS Phase 2 SMART Program Renovations (CMAR)		Above	3-Design/Permit	15%
P.002003	Miramar HS SMART Program Renovations (CMAR)	\$		3-Design/Permit	97%
P.001890	North Andrews Gardens ES SMART Program Renovations	\$	3,075,300	3-Design/Permit	55%
-		,			
P.002301	Northeast HS New Addition and Renovation to Building 12 (CMAR)	\$		3-Design/Permit	50%
P.002149	Nova Blanche Forman ES SMART Program Renovations	\$	3,169,542	3-Design/Permit	95%
	Nova Dwight D. Eisenhower ES SMART Program Renovations (CC-				
P.002145	CMAR)	\$		3-Design/Permit	95%
P.001970	Oriole ES SMART Program Renovations	\$		3-Design/Permit	97%
P.002084	Palmview ES SMART Program Renovations	\$		3-Design/Permit	91%
P.002062	Park Springs ES SMART Program Renovations	\$		3-Design/Permit	91%
P.001807	Parkway MS SMART Program Renovations	\$	6,500,000	3-Design/Permit	85%

				Ι .	
P.001634	Pasadena Lakes ES SMART Program Renovations	\$		3-Design/Permit	99%
P.002041	Peters ES SMART Program Renovations	\$		3-Design/Permit	93%
P.001916	Plantation HS SMART Program Renovations (CMAR)	\$		3-Design/Permit	85%
P.002136	Plantation Park ES SMART Program Renovations	\$		3-Design/Permit	96%
P.002091	Pompano Beach HS SMART Program Renovations	\$		3-Design/Permit	91%
P.002039	Riverside ES SMART Program Renovations	\$	2,025,000	3-Design/Permit	90%
P.002132	Sanders Park Elementary Magnet School SMART Program Renovations	\$	6,761,385	3-Design/Permit	93%
P.002127	Sawgrass ES SMART Renovations	\$	5,079,895	3-Design/Permit	95%
P.002047	Seminole MS SMART Program Renovations	\$		3-Design/Permit	93%
P.001636	Sheridan Hills ES SMART Program Renovations	\$		3-Design/Permit	97%
P.002060	Shoridan Tashnisal Callaga SMAPT Program Panayations (CMAP)	خ	12 520 202	2 Dosign/Bormit	92%
	Sheridan Technical College SMART Program Renovations (CMAR)	\$ \$		3-Design/Permit	
P.002090	South Plantation HS SMART Program Renovations	Þ	5,028,100	3-Design/Permit	91%
P.002163	Stranahan HS Cafe/Kitchen SMART Program Renovations (CMAR)	\$	6,675,000	3-Design/Permit	55%
P.002079	Sunshine ES SMART Program Renovations	\$	2,304,000	3-Design/Permit	91%
P.002042	Tequesta Trace MS SMART Program Renovations	\$	4,936,500	3-Design/Permit	50%
P.002010	Walter C. Young MS SMART Program Renovations (CMAR)	\$	13,691,256	3-Design/Permit	97%
P.002114	Welleby ES SMART Program Renovations	\$	4,415,688	3-Design/Permit	95%
P.001967	Western HS SMART Program Renovations	\$	5,409,280	3-Design/Permit	94%
P.001711	Whiddon-Rogers EC SMART Program Renovations	\$		3-Design/Permit	95%
					2=0/
P.002089	Whispering Pines EC SMART Program Renovations (CC-CMAR)	\$	3,412,864	3-Design/Permit	95%
P.001658	William T. McFatter Technical College SMART Program Renovations	\$	17.678.578	3-Design/Permit	97%
P.001917	Wilton Manors ES SMART Program Renovations	\$		3-Design/Permit	80%
Phase 4 - Bid		1 *	.,0,000	o Designy remine	0070
			4.477.000	I	700/
P.001633	Attucks MS Phase 2 SMART Program Renovations (CC-A)	\$		4-Bid & Award	70%
P.002044	Bair MS SMART Program Renovations	\$		4-Bid & Award	5%
P.002065	Boulevard Heights ES SMART Program Renovations	\$		4-Bid & Award	5%
P.001846	Boyd H. Anderson HS SMART Program Renovations	\$		4-Bid & Award	98%
P.001638	Broadview ES SMART Program Renovations	\$		4-Bid & Award	99%
P.002037	Broward Estates ES SMART Program Renovations	\$		4-Bid & Award	5%
P.001920	C. Robert Markham ES SMART Program Renovations (CMAR)	\$		4-Bid & Award	85%
P.001757	Central Park ES SMART Program Renovations	\$		4-Bid & Award	15%
P.002040	Challenger ES SMART Program Renovations	\$		4-Bid & Award	5%
P.001659	Collins ES SMART Program Renovations (CC-A)(CC-CMAR)	\$		4-Bid & Award	25%
P.002150	Cooper City ES SMART Program Renovations	\$		4-Bid & Award	98%
P.001765	Coral Springs HS SMART Program Renovations	\$		4-Bid & Award	99%
P.002002	Country Isles ES SMART Program Renovations	\$		4-Bid & Award	65%
P.002081	Cross Creek School SMART Program Renovations	\$		4-Bid & Award	5%
P.001899	Davie ES SMART Program Renovations	\$		4-Bid & Award	25%
P.001820	Deerfield Beach ES SMART Program Renovations	\$		4-Bid & Award	55%
P.001694	Deerfield Beach HS Phase 1 SMART Program Renovations	\$	5 358 149	4-Bid & Award	5%
P.002036	Deerfield Park ES SMART Program Renovations	\$	7,316,977	4-Bid & Award	5%
P.001915	Dillard ES SMART Program Renovations	\$ \$	7,316,977 4,358,734	4-Bid & Award 4-Bid & Award	5% 5%
P.001915 P.001837	Dillard ES SMART Program Renovations Driftwood MS SMART Program Renovations	\$ \$ \$	7,316,977 4,358,734 11,836,879	4-Bid & Award 4-Bid & Award 4-Bid & Award	5% 5% 20%
P.001915	Dillard ES SMART Program Renovations	\$ \$	7,316,977 4,358,734 11,836,879	4-Bid & Award 4-Bid & Award	5% 5% 20% 55%
P.001915 P.001837	Dillard ES SMART Program Renovations Driftwood MS SMART Program Renovations Eagle Point ES SMART Program Renovations	\$ \$ \$ \$	7,316,977 4,358,734 11,836,879 6,145,450	4-Bid & Award 4-Bid & Award 4-Bid & Award 4-Bid & Award	5% 5% 20% 55%
P.001915 P.001837 P.001746 P.002111	Dillard ES SMART Program Renovations Driftwood MS SMART Program Renovations Eagle Point ES SMART Program Renovations Endeavour Primary LC ES SMART Program Renovations (CC-CMAR)	\$ \$ \$ \$	7,316,977 4,358,734 11,836,879 6,145,450 2,360,590	4-Bid & Award 4-Bid & Award 4-Bid & Award 4-Bid & Award 4-Bid & Award	5% 5% 20% 55% 75%
P.001915 P.001837 P.001746 P.002111 P.002135	Dillard ES SMART Program Renovations Driftwood MS SMART Program Renovations Eagle Point ES SMART Program Renovations Endeavour Primary LC ES SMART Program Renovations (CC-CMAR) Flamingo ES SMART Program Renovations (CC-A)	\$ \$ \$ \$	7,316,977 4,358,734 11,836,879 6,145,450 2,360,590 3,199,085	4-Bid & Award	5% 5% 20% 55% 75% 15%
P.001915 P.001837 P.001746 P.002111 P.002135 P.002001	Dillard ES SMART Program Renovations Driftwood MS SMART Program Renovations Eagle Point ES SMART Program Renovations Endeavour Primary LC ES SMART Program Renovations (CC-CMAR) Flamingo ES SMART Program Renovations (CC-A) Floranada ES SMART Program Renovations	\$ \$ \$ \$ \$	7,316,977 4,358,734 11,836,879 6,145,450 2,360,590 3,199,085 2,650,744	4-Bid & Award	5% 5% 20% 55% 75% 15% 5%
P.001915 P.001837 P.001746 P.002111 P.002135 P.002001 P.001973	Dillard ES SMART Program Renovations Driftwood MS SMART Program Renovations Eagle Point ES SMART Program Renovations Endeavour Primary LC ES SMART Program Renovations (CC-CMAR) Flamingo ES SMART Program Renovations (CC-A) Floranada ES SMART Program Renovations Fox Trail ES SMART Program Renovations	\$ \$ \$ \$ \$ \$	7,316,977 4,358,734 11,836,879 6,145,450 2,360,590 3,199,085 2,650,744 1,541,265	4-Bid & Award 4-Bid & Award 4-Bid & Award 4-Bid & Award 4-Bid & Award 4-Bid & Award 4-Bid & Award 4-Bid & Award 4-Bid & Award	5% 5% 20% 55% 75% 15% 5%
P.001915 P.001837 P.001746 P.002111 P.002135 P.002001	Dillard ES SMART Program Renovations Driftwood MS SMART Program Renovations Eagle Point ES SMART Program Renovations Endeavour Primary LC ES SMART Program Renovations (CC-CMAR) Flamingo ES SMART Program Renovations (CC-A) Floranada ES SMART Program Renovations	\$ \$ \$ \$ \$	7,316,977 4,358,734 11,836,879 6,145,450 2,360,590 3,199,085 2,650,744 1,541,265	4-Bid & Award	5% 5% 20% 55% 75% 15% 5%
P.001915 P.001837 P.001746 P.002111 P.002135 P.002001 P.001973	Dillard ES SMART Program Renovations Driftwood MS SMART Program Renovations Eagle Point ES SMART Program Renovations Endeavour Primary LC ES SMART Program Renovations (CC-CMAR) Flamingo ES SMART Program Renovations (CC-A) Floranada ES SMART Program Renovations Fox Trail ES SMART Program Renovations Glades MS SMART Program Renovations	\$ \$ \$ \$ \$ \$	7,316,977 4,358,734 11,836,879 6,145,450 2,360,590 3,199,085 2,650,744 1,541,265 386,000	4-Bid & Award 4-Bid & Award 4-Bid & Award 4-Bid & Award 4-Bid & Award 4-Bid & Award 4-Bid & Award 4-Bid & Award 4-Bid & Award	5% 5% 20% 55% 75% 15% 5% 40%
P.001915 P.001837 P.001746 P.002111 P.002135 P.002001 P.001973 P.001968	Dillard ES SMART Program Renovations Driftwood MS SMART Program Renovations Eagle Point ES SMART Program Renovations Endeavour Primary LC ES SMART Program Renovations (CC-CMAR) Flamingo ES SMART Program Renovations (CC-A) Floranada ES SMART Program Renovations Fox Trail ES SMART Program Renovations Glades MS SMART Program Renovations Glades MS SMART Program Renovations Gulfstream Academy of Hallandale Beach (FKA Hallandale ES) SMART	\$ \$ \$ \$ \$ \$ \$	7,316,977 4,358,734 11,836,879 6,145,450 2,360,590 3,199,085 2,650,744 1,541,265 386,000	4-Bid & Award 4-Bid & Award 4-Bid & Award 4-Bid & Award 4-Bid & Award 4-Bid & Award 4-Bid & Award 4-Bid & Award 4-Bid & Award 4-Bid & Award 4-Bid & Award	5% 5% 20% 55% 75% 15% 5% 40%
P.001915 P.001837 P.001746 P.002111 P.002135 P.002001 P.001973 P.001968	Dillard ES SMART Program Renovations Driftwood MS SMART Program Renovations Eagle Point ES SMART Program Renovations Endeavour Primary LC ES SMART Program Renovations (CC-CMAR) Flamingo ES SMART Program Renovations (CC-A) Floranada ES SMART Program Renovations Fox Trail ES SMART Program Renovations Glades MS SMART Program Renovations Glades MS SMART Program Renovations Gulfstream Academy of Hallandale Beach (FKA Hallandale ES) SMART Program Renovations	\$ \$ \$ \$ \$ \$ \$	7,316,977 4,358,734 11,836,879 6,145,450 2,360,590 3,199,085 2,650,744 1,541,265 386,000 2,298,994	4-Bid & Award 4-Bid & Award 4-Bid & Award 4-Bid & Award 4-Bid & Award 4-Bid & Award 4-Bid & Award 4-Bid & Award 4-Bid & Award 4-Bid & Award 4-Bid & Award	5% 5% 20% 55% 75% 15% 5%

P.002147 Heron Heightt ES SMART Program Renovations \$.8386.001 4-Bit & Award 5-Bit	D 00100C	Harry D. Dawy FC CMART Drawns Barry stiers		7 422 060	4 Dial O A	
P.001983	P.001986	Henry D. Perry EC SMART Program Renovations	\$			5%
DOD1288			_			
P.002038			_			
P.001912		,	_			
D002070 Lakeside ES SMART Program Renovations \$ 1,891,350 4-Bit & Award 55			_			
D00073			_			
P.001635 Lauderdale Manors ELC SMART Program Renovations \$ 6,950,500 4-Bit & Award 15% P.002066 Lauderhill*Paul Turner ES SMART Program Renovations \$ 1,381,819 4-Bit & Award 5% P.002046 Millennium 6-12 Collegiate Academy SMART Program Renovations \$ 1,061,000 4-Bit & Award 5% P.001710 New River MS SMART Program Renovations \$ 1,061,000 4-Bit & Award 5% P.001710 New River MS SMART Program Renovations \$ 1,324,600 4-Bit & Award 25% P.001710 New River MS SMART Program Renovations \$ 1,318,2600 4-Bit & Award 25% P.001710 New River MS SMART Program Renovations \$ 1,318,2600 4-Bit & Award 25% P.001899 North Lauderdale PRS SMART Program Renovations (CC-A) \$ 0,650,000 4-Bit & Award 25% P.001898 North Lauderdale PRS SMART Program Renovations \$ 1,325,304 4-Bit & Award 50% P.001903 North Lauderdale PRS SMART Program Renovations \$ 1,561,424 4-Bit & Award 50% P.001898 Powen MS SMART Program Renovations \$ 1,361,424 4-Bit & Award			_			
P. 002066 Lauderhill-Paul Turner ES SMART Program Renovations S S,381,819 A-Bid & Award S9 P. 002083 Meadowbrook ES SMART Program Renovations S 1,061,000 A-Bid & Award S9 P. 002086 Meadowbrook ES SMART Program Renovations S 1,061,000 A-Bid & Award S9 P. 002086 Millennium G-12 Collegiate Academy SMART Program Renovations S 4,441,099 A-Bid & Award S9 P. 002086 Millennium G-12 Collegiate Academy SMART Program Renovations S 4,234,600 A-Bid & Award S9 P. 0020112 Not Phill ES SMART Program Renovations S 2,709,914 A-Bid & Navard S9 P. 0020119 Not Phill ES SMART Program Renovations S 3,182,500 A-Bid & Award S9 P. 0020199 North Fork ES SMART Program Renovations S 3,182,500 A-Bid & Award S9 P. 0020199 North Fork ES SMART Program Renovations S 6,609,000 A-Bid & Award S9 P. 0020199 North Fork ES SMART Program Renovations S 2,259,350 A-Bid & Award S9 P. 0020190 North Earl SMART Program Renovations S 2,259,350 A-Bid & Award S9 P. 0020190 North Earl SMART Program Renovations S 2,259,350 A-Bid & Award S9 P. 0020191 North Earl SMART Program Renovations S 3,651,424 A-Bid & Award S9 P. 0020195 Particle Filt SMART Program Renovations S 3,651,424 A-Bid & Award S9 P. 002059 Particle Filt SMART Program Renovations S 3,651,424 A-Bid & Award S9 P. 002059 Particle Filt SMART Program Renovations S 3,753,260 A-Bid & Award S9 P. 002059 Particle Filt SMART Program Renovations S 3,753,260 A-Bid & Award S9 P. 002059 Particle ES SMART Program Renovations S 3,651,424 A-Bid & Award S9 P. 002059 Particle ES SMART Program Renovations S 3,653,400 A-Bid & Award S9 P. 002069 Particle ES SMART Program Renovations S 3,653,400 A-Bid & Award S9 P. 002069 Particle ES SMART Program Renovations S 3,653,400 A-Bid & Award S9 P. 002074 Particle ES SMART Program Renovations S 3,224,400 A-Bid & Award S9		•	_			
P.001547 Margate ES Phase 1 SMART Program Renovations \$ 8,854,000 4818 & Award 55			_			
P.002083 Meadowbrook ES SMART Program Renovations \$ 1,061,000 4-Bid & Award 59 P.002046 Millennium 6-12 Collegiate Academy SMART Program Renovations \$ 4,441,099 4-Bid & Award 59 P.002112 New River MS SMART Program Renovations \$ 2,209,0914 4-Bid & Award 59 P.002109 Not Hill ES SMART Program Renovations (CC-MAR) \$ 2,709,914 4-Bid & Award 59 P.001099 Nortrest ES SMART Program Renovations (CC-MAR) \$ 3,182,500 4-Bid & Award 458 P.001090 North Fork ES SMART Program Renovations (CC-A) \$ 0,505,000 4-Bid & Award 458 P.001093 North Loaderdale PR8 SMART Program Renovations (CC-A) \$ 2,259,350 4-Bid & Award 590 P.001093 North Loaderdale PR8 SMART Program Renovations (CC-A) \$ 2,259,350 4-Bid & Award 590 P.001093 North Loaderdale PR8 SMART Program Renovations (CC-A) \$ 2,259,350 4-Bid & Award 590 P.001093 North Loaderdale PR8 SMART Program Renovations \$ 2,250,300 4-Bid & Award 590 P.001093 North Loaderdale PR8 SMART Program Renovations \$ 3,651,424 4-Bid & Award 590 P.001093 North Loaderdale PR8 SMART Program Renovations \$ 3,651,424 4-Bid & Award 590 P.001093 Park Likes ES SMART Program Renovations \$ 3,651,424 4-Bid & Award 590 P.001093 Park Likes ES SMART Program Renovations \$ 3,753,260 4-Bid & Award 590 P.001094 Park Trails ES SMART Program Renovations \$ 3,753,260 4-Bid & Award 590 P.001094 Park Trails ES SMART Program Renovations \$ 3,753,260 4-Bid & Award 590 P.001094 Park Trails ES SMART Program Renovations \$ 3,653,404 4-Bid & Award 590 P.001210 Park Trails ES SMART Program Renovations \$ 3,653,404 4-Bid & Award 590 P.001210 Park Trails ES SMART Program Renovations \$ 1,662,142 4-Bid & Award 590 P.001210 Park Trails ES SMART Program Renovations \$ 1,662,142 4-Bid & Award 590 P.001210 Park Trails ES SMART Program Renovations \$ 1,662,142 4-Bid & Award 590 P.001210 Park Trails ES SMART Program Renovations \$ 6,636,300			_			5%
P. 002046 Millennium 6-12 Collegiate Academy SMART Program Renovations \$ 4,441,099 4-Bid & Award 25% P. 001710 New River MS SMART Program Renovations \$ 4,324,600 4-Bid & Award 25% P. 0017112 Nob Hill ES SMART Program Renovations \$ 2,709,514 4-Bid & Award 25% P. 001899 North Fork ES SMART Program Renovations (CC-A) \$ 6,050,000 4-Bid & Award 45% P. 001899 North Fork ES SMART Program Renovations (CC-A) \$ 6,050,000 4-Bid & Award 25% P. 001899 North East HS SMART Program Renovations (CC-A) \$ 2,523,300 4-Bid & Award 25% P. 001903 North East HS SMART Program Renovations (CMAR) \$ 2,529,300 4-Bid & Award 50% P. 001903 Nova MS SMART Program Renovations \$ 6,633,300 4-Bid & Award 50% P. 001905 Olsen MS SMART Program Renovations \$ 13,651,424 4-Bid & Award 15% P. 001905 Olsen MS SMART Program Renovations \$ 3,461,665 4-Bid & Award 15% P. 001905 Panther Run ES SMART Program Renovations \$ 774,000 4-Bid & Award 50% P. 001908 Park Likes ES SMART Program Renovations \$ 7,905,219 4-Bid & Award 50% P. 001908 Park Likes ES SMART Program Renovations \$ 7,905,219 4-Bid & Award 50% P. 002010 Park Incline SMART Program Renovations \$ 2,965,219 4-Bid & Award 50% P. 0020210 Park Incline SMART Program Renovations \$ 2,965,219 4-Bid & Award 50% P. 0020210 Park Incline SMART Program Renovations \$ 2,666,953 4-Bid & Award 50% P. 0020210 Park Incline SMART Program Renovations \$ 5,241,914 4-Bid & Award 50% P. 0020210 Pines Likes ES SMART Program Renovations \$ 5,241,914 4-Bid & Award 50% P. 0020210 Pines Likes ES SMART Program Renovations \$ 6,263,000 4-Bid & Award 50% P. 002130 Pines MS SMART Program Renovations \$ 6,636,300 4-Bid & Award 50% P. 002140 Pines Likes ES SMART Program Renovations \$ 6,636,300 4-Bid & Award 50% P. 002140 Pines Likes ES SMART Program Renovations \$ 7,975,900 4-Bid & Award 50% P. 002140 Pines Likes ES SMART			_			
P.001710 New River MS SMART Program Renovations \$ 4,324,600 4-Bid & Award 5.99	P.002083	Meadowbrook ES SMART Program Renovations	\$	1,061,000	4-Bid & Award	65%
P.001710 New River MS SMART Program Renovations \$ 4,324,600 4-Bid & Award 5.99						
P.002112	P.002046	Millennium 6-12 Collegiate Academy SMART Program Renovations				5%
P.001969	P.001710	New River MS SMART Program Renovations	_	4,324,600	4-Bid & Award	25%
P.001943	P.002112	Nob Hill ES SMART Program Renovations (CC-CMAR)	_	2,709,914	4-Bid & Award	5%
P.001303	P.001969	Norcrest ES SMART Program Renovations	\$	3,182,500	4-Bid & Award	45%
Noorliéesat HIS SMART Program Renovations (CMAR) \$ 22,500,000 4-Bid & Award 50%	P.001849	North Fork ES SMART Program Renovations (CC-A)	\$	6,050,000	4-Bid & Award	35%
P.001388 Nova MS SMART Program Renovations \$ 6,39,300 4-Bid & Award 15% P.001955 Olisen MS SMART Program Renovations \$ 13,651,424 4-Bid & Award 15% P.002069 Panther Run ES SMART Program Renovations \$ 3,461,665 4-Bid & Award 5% P.001988 Park Lakes ES SMART Program Renovations \$ 774,000 4-Bid & Award 5% P.001944 Park Ridge ES SMART Program Renovations \$ 2,965,219 4-Bid & Award 5% P.002116 Park Trails ES SMART Program Renovations \$ 3,753,260 4-Bid & Award 5% P.002126 Park Ridge ES SMART Program Renovations \$ 3,753,260 4-Bid & Award 5% P.002082 Parkside ES SMART Program Renovations \$ 2,666,953 4-Bid & Award 5% P.002094 Penbroke Lakes ES SMART Program Renovations \$ 1,628,127 4-Bid & Award 15% P.002004 Pines Lakes ES SMART Program Renovations \$ 1,628,127 4-Bid & Award 5% P.002130 Pines MS SMART Program Renovations \$ 20,491,400 4-Bid & Award 5% P.001744 Pijer HS SMART Program Renovations \$ 20,491,400 4-Bid & Award 5% P.001749 Plantation MS SMART Program Renovations \$ 6,363,000 4-Bid & Award 15% P.001867 Ramblewood MS SMART Program Renovations \$ 6,876,241 4-Bid & Award 15% P.001869 Program Renovations \$ 6,876,241 4-Bid & Award 5% P.001860 Sawgrass Springs MS SMART Program Renovations \$ 7,975,900 4-Bid & Award 5% P.001870 Sawgrass Springs MS SMART Program Renovations \$ 9,275,500 4-Bid & Award 5% P.001871 Sheridan Park ES SMART Program Renovations \$ 1,174,885 4-Bid & Award 5% P.001872 South Broward HS SMART Program Renovations \$ 1,174,885 4-Bid & Award 5% P.001905 Stirling ES SMART Program Renovations \$ 1,174,885 4-Bid & Award 5% P.001905 Stirling ES SMART Program Renovations \$ 1,174,885 4-Bid & Award 5% P.001905 Stirling ES SMART Program Renovations \$ 1,286,0979 4-Bid & Award 5% P.001904 Tradewinds ES SMART Program Renovations \$ 1,287,099 4-Bid & Award 5% P.001905 Virginia Shumar Program Re	P.001903	North Lauderdale PK8 SMART Program Renovations (CC-A)	\$	2,529,350	4-Bid & Award	75%
P.001555 Olsen MS SMART Program Renovations \$ 13,651,424	P.001684	Northeast HS SMART Program Renovations (CMAR)	\$	22,500,000	4-Bid & Award	50%
P.001988	P.001898	Nova MS SMART Program Renovations	\$	6,639,300	4-Bid & Award	15%
P.001988	P.001955	Olsen MS SMART Program Renovations	\$	13,651,424	4-Bid & Award	15%
P.001844 Park Ridge ES SMART Program Renovations \$ 2,965,219 4-Bid & Award 59	P.002069	Panther Run ES SMART Program Renovations	\$	3,461,665	4-Bid & Award	5%
P.001844 Park Ridge ES SMART Program Renovations \$ 2,965,219 4-Bid & Award 5%	P.001988		_			5%
P.002116 Park Trails ES SMART Program Renovations \$ 3,753,260 4-Bid & Award 59% P.0020202 Parkside ES SMART Program Renovations \$ 2,666,953 4-Bid & Award 59% P.0021404 Pembroke Lakes ES SMART Program Renovations \$ 5,321,416 4-Bid & Award 15% P.002104 Pines Lakes ES SMART Program Renovations \$ 1,628,127 4-Bid & Award 31% P.002104 Pines Lakes ES SMART Program Renovations \$ 1,628,127 4-Bid & Award 31% P.002130 Pines MS SMART Program Renovations \$ 20,491,400 4-Bid & Award 31% P.001729 Piantation MS SMART Program Renovations \$ 6,636,300 4-Bid & Award 15% P.001867 Ramblewood MS SMART Program Renovations \$ 6,636,300 4-Bid & Award 25% P.001879 Program Renovations \$ 7,975,900 4-Bid & Award 25% P.001889 Program Renovations \$ 7,975,900 4-Bid & Award 5% P.001841 Sawgrass Springs MS SMART Program Renovations \$ 9,275,500 4-Bid & Award 5% P.001841 Sawgrass Springs MS SMART Program Renovations \$ 6,313,240 P.001842 Silver Palms ES SMART Program Renovations \$ 6,313,24 P.001843 South Broward HS SMART Program Renovations \$ 6,313,24 P.001846 Silver Palms ES SMART Program Renovations \$ 6,313,24 P.001858 Stephen Foster ES SMART Program Renovations \$ 1,0948,400 P.001808 Tedder ES SMART Program Renovations \$ 1,0948,400 P.001905 Stirling ES SMART Program Renovations \$ 1,0948,400 P.001906 Thurgood Marshall ES SMART Program Renovations \$ 1,0948,400 P.001906 Thurgood Marshall ES SMART Program Renovations \$ 1,0948,400 P.001906 Traplical ES SMART Program Renovations \$ 1,0948,400 P.001909 Tradewinds ES SMART Program Renovations \$ 1,0948,400 P.001909 Village ES SMART Program Renovations \$ 1,0948,400 P.00			_			5%
P.002082 Parkside ES SMART Program Renovations \$ 2,666,953 4-Bid & Award 15%			_			5%
P.001842 Pembroke Lakes ES SMART Program Renovations \$ 1,628,127 4-Bid & Award 60% P.0021004 Pines Lakes ES SMART Program Renovations \$ 1,628,127 4-Bid & Award 60% P.002130 Pines MS SMART Program Renovations \$ 20,411,400 4-Bid & Award 31% P.001744 Piper HS SMART Program Renovations \$ 20,411,400 4-Bid & Award 55% P.001729 Plantation MS SMART Program Renovations \$ 6,636,300 4-Bid & Award 15% P.001867 Ramblewood MS SMART Program Renovations \$ 6,878,241 4-Bid & Award 25% P.001876 Ramblewood MS SMART Program Renovations \$ 6,878,241 4-Bid & Award 25% P.001886 Program Renovations \$ 7,975,900 4-Bid & Award 5% P.001896 Program Renovations \$ 9,275,500 4-Bid & Award 5% P.001841 Sawgrass Springs MS SMART Program Renovations \$ 9,275,500 4-Bid & Award 5% P.002146 Silver Palms ES SMART Program Renovations \$ 6,131,239 4-Bid & Award 5% P.002146 Silver Palms ES SMART Program Renovations \$ 4,171,885 4-Bid & Award 5% P.001838 South Broward HS SMART Program Renovations \$ 10,948,406 4-Bid & Award 15% P.001905 Stirling ES SMART Program Renovations \$ 1,285,098 4-Bid & Award 15% P.001905 Stirling ES SMART Program Renovations \$ 4,215,616 4-Bid & Award 5% P.001905 Stirling ES SMART Program Renovations \$ 4,215,616 4-Bid & Award 5% P.001904 Tropical ES SMART Program Renovations \$ 4,215,616 4-Bid & Award 5% P.001904 Tropical ES SMART Program Renovations \$ 3,704,874 4-Bid & Award 5% P.001905 Virginia Shuman Young Montessori SMART Program Renovations \$ 3,704,874 4-Bid & Award 5% P.001905 Virginia Shuman Young Montessori SMART Program Renovations \$ 3,704,874 4-Bid & Award 5% P.002000 Virginia Shuman Young Montessori SMART Program Renovations \$ 3,320,000 4-Bid & Award 5% P.001905 Watkins ES SMART Program Renovations \$ 5,830,151 4-Bid & Award 5% P.001905 Watkins ES SMART Program Renovations \$ 5,830,151 4-Bid & Award 5%			_			5%
P.002004 Pines Lakes ES SMART Program Renovations \$ 1,628,127 4-Bid & Award 60% P.002130 Pines MS SMART Program Renovations (CC-CMAR) \$ 621,491 4-Bid & Award 31% P.001742 Piper HS SMART Program Renovations \$ 20,491,400 4-Bid & Award 55% P.001729 Plantation MS SMART Program Renovations \$ 6,636,300 4-Bid & Award 15% P.001867 Ramblewood MS SMART Program Renovations \$ 6,878,241 4-Bid & Award 25% P.001896 Program Renovations \$ 7,975,900 4-Bid & Award 5% P.001896 Program Renovations \$ 7,975,900 4-Bid & Award 5% P.001896 Program Renovations \$ 9,275,500 4-Bid & Award 5% P.0018141 Sawgrass Springs MS SMART Program Renovations \$ 9,275,500 4-Bid & Award 5% P.002146 Silver Palms ES SMART Program Renovations \$ 6,131,293 4-Bid & Award 5% P.001838 South Broward HS SMART Program Renovations \$ 10,948,406 4-Bid & Award 5% P.001838 South Broward HS SMART Program Renovations \$ 10,948,406 4-Bid & Award 5% P.001905 Stirling ES SMART Program Renovations \$ 10,948,406 4-Bid & Award 5% P.001808 Tedder ES SMART Program Renovations \$ 1,376,295 4-Bid & Award 5% P.001808 Tedder ES SMART Program Renovations \$ 4,215,616 4-Bid & Award 5% P.001904 Tropical ES SMART Program Renovations \$ 4,215,616 4-Bid & Award 5% P.001904 Tropical ES SMART Program Renovations \$ 4,215,616 4-Bid & Award 5% P.001904 Tropical ES SMART Program Renovations \$ 3,704,874 4-Bid & Award 5% P.001904 Tropical ES SMART Program Renovations \$ 1,788,337 4-Bid & Award 5% P.001905 Village ES SMART Program Renovations \$ 1,788,337 4-Bid & Award 5% P.001904 Watkins ES SMART Program Renovations \$ 1,788,337 4-Bid & Award 5% P.001904 Watkins ES SMART Program Renovations \$ 1,587,919 4-Bid & Award 5% P.001904 Watkins ES SMART Program Renovations \$ 1,587,919 4-Bid & Award 5% P.001904 Watkins ES SMART Program Renovations \$ 1,680,000 4-Bid & Award 5% P.			_			15%
P.002130 Pines MS SMART Program Renovations (CC-CMAR) \$ 621,491 4-Bid & Award 31%			_			
P.001744 Piper HS SMART Program Renovations \$ 20,491,400 4-Bid & Award 559 P.001729 Plantation MS SMART Program Renovations \$ 6,636,300 4-Bid & Award 159 P.001867 Ramblewood MS SMART Program Renovations \$ 6,878,241 4-Bid & Award 259 Royal Palm STEM Museum Magnet (fka Royal Palm ES) SMART P.001896 Program Renovations \$ 7,975,900 4-Bid & Award 59 P.001841 Sawgrass Springs MS SMART Program Renovations \$ 9,275,500 4-Bid & Award 59 P.001841 Sawgrass Springs MS SMART Program Renovations \$ 9,275,500 4-Bid & Award 59 P.002174 Sheridan Park ES SMART Program Renovations \$ 6,131,293 4-Bid & Award 59 P.002146 Silver Palms ES SMART Program Renovations \$ 4,171,885 4-Bid & Award 59 P.001838 South Broward HS SMART Program Renovations \$ 10,948,406 4-Bid & Award 159 P.001905 Stirling ES SMART Program Renovations \$ 4,376,295 4-Bid & Award 59 P.001905 Stirling ES SMART Program Renovations \$ 4,376,295 4-Bid & Award 59 P.001808 Tedder ES SMART Program Renovations \$ 4,215,616 4-Bid & Award 159 P.001674 Thurgood Marshall ES SMART Program Renovations \$ 4,215,616 4-Bid & Award 159 P.001904 Tropical ES SMART Program Renovations \$ 3,704,874 4-Bid & Award 59 P.001904 Tropical ES SMART Program Renovations \$ 3,704,874 4-Bid & Award 59 P.001905 Village ES SMART Program Renovations \$ 3,335,530 4-Bid & Award 59 P.001901 Watkins ES SMART Program Renovations \$ 3,120,000 4-Bid & Award 59 P.001901 Watkins ES SMART Program Renovations \$ 3,200,000 4-Bid & Award 59 P.002101 Westglades MS FMART Program Renovations \$ 3,200,000 4-Bid & Award 59 P.002101 Westglades MS FMART Program Renovations \$ 4,702,994 4-Bid & Award 59 P.001901 Watkins ES SMART Program Renovations \$ 5,830,151 4-Bid & Award 59 P.001918 Winston Park ES SMART Program Renovations \$ 5,830,151 4-Bid & Award 59 P.001918 Winston Park ES SMART Program Renovations \$ 5,831			_			
P.001729 Plantation MS SMART Program Renovations \$ 6,636,300 4-Bid & Award 15%			_			
P.001867 Ramblewood MS SMART Program Renovations \$ 6,878,241 4-Bid & Award 25%		· ·	<u> </u>			
Royal Palm STEM Museum Magnet (fka Royal Palm ES) SMART			_			
P.001896 Program Renovations \$ 7,975,900 4-Bid & Award 5% P.001841 Sawgrass Springs MS SMART Program Renovations \$ 9,275,500 4-Bid & Award 5% P.002171 Sheridan Park ES SMART Program Renovations \$ 6,131,293 4-Bid & Award 5% P.002146 Silver Palms ES SMART Program Renovations \$ 4,171,885 4-Bid & Award 5% P.001838 South Broward HS SMART Program Renovations \$ 10,948,406 4-Bid & Award 15% P.001907 Stephen Foster ES SMART Program Renovations \$ 12,948,406 4-Bid & Award 15% P.001905 Stirling ES SMART Program Renovations (CC-A) \$ 4,376,295 4-Bid & Award 70% P.001808 Tedder ES SMART Program Renovations \$ 4,215,616 4-Bid & Award 15% P.001674 Thurgood Marshall ES SMART Program Renovations \$ 4,560,979 4-Bid & Award 5% P.001229 Tradewinds ES SMART Program Renovations \$ 3,704,874 4-Bid & Award 5% P.001904 Tropical ES SMART Program Renovations (CC-A) \$ 1,778,337 4-Bid & Award 15% P.0	1.001007		7	0,070,241	T Bla & 7 (Wara	2370
P.001841 Sawgrass Springs MS SMART Program Renovations \$ 9,275,500 4-Bid & Award 5% P.002071 Sheridan Park ES SMART Program Renovations \$ 6,131,293 4-Bid & Award 5% P.002146 Siliver Palms ES SMART Program Renovations \$ 4,171,885 4-Bid & Award 5% P.001838 South Broward HS SMART Program Renovations \$ 10,948,406 4-Bid & Award 15% P.001907 Stephen Foster ES SMART Program Renovations \$ 1,0948,406 4-Bid & Award 5% P.001905 Stirling ES SMART Program Renovations (CC-A) \$ 4,376,295 4-Bid & Award 70% P.001905 Stirling ES SMART Program Renovations \$ 4,215,616 4-Bid & Award 15% P.001904 Tradewinds ES SMART Program Renovations \$ 4,560,979 4-Bid & Award 5% P.001904 Tropical ES SMART Program Renovations (CC-A) \$ 1,778,337 4-Bid & Award 5% P.001952 Village ES SMART Program Renovations (CC-A) \$ 1,787,937 4-Bid & Award 15% P.002100 Virginia Shuman Young Montessori SMAT Program Renovations \$ 3,335,530 4-Bid & Award 5% <td>P 001896</td> <td>1 '</td> <td>Ġ</td> <td>7 975 900</td> <td>1-Rid & Award</td> <td>5%</td>	P 001896	1 '	Ġ	7 975 900	1-Rid & Award	5%
P.002071 Sheridan Park ES SMART Program Renovations \$ 6,131,293 4-Bid & Award 59		<u> </u>				
P.002146 Silver Palms ES SMART Program Renovations \$ 4,171,885 4-Bid & Award 5% P.001838 South Broward HS SMART Program Renovations \$ 10,948,406 4-Bid & Award 15% P.002067 Stephen Foster ES SMART Program Renovations \$ 5,285,098 4-Bid & Award 5% P.001905 Stirling ES SMART Program Renovations (CC-A) \$ 4,376,295 4-Bid & Award 70% P.001808 Tedder ES SMART Program Renovations \$ 4,215,616 4-Bid & Award 15% P.001674 Thurgood Marshall ES SMART Program Renovations \$ 4,560,979 4-Bid & Award 5% P.0012129 Tradewinds ES SMART Program Renovations \$ 3,704,874 4-Bid & Award 5% P.001904 Tropical ES SMART Program Renovations (CC-A) \$ 1,778,337 4-Bid & Award 15% P.001952 Village ES SMART Program Renovations (CC-A) \$ 1,587,919 4-Bid & Award 15% P.002000 Virginia Shuman Young Montessori SMART Program Renovations \$ 3,335,530 4-Bid & Award 5% P.002174 Watkins ES SMART Program Renovations \$ 4,702,994 4-Bid & Award 5%		<u> </u>	_			
P.001838 South Broward HS SMART Program Renovations \$ 10,948,406 4-Bid & Award 15%		<u> </u>	_			
P.002067 Stephen Foster ES SMART Program Renovations \$ 5,285,098 4-Bid & Award 5% P.001905 Stirling ES SMART Program Renovations (CC-A) \$ 4,376,295 4-Bid & Award 70% P.001808 Tedder ES SMART Program Renovations \$ 4,215,616 4-Bid & Award 15% P.001674 Thurgood Marshall ES SMART Program Renovations \$ 4,560,979 4-Bid & Award 5% P.002129 Tradewinds ES SMART Program Renovations (CC-A) \$ 1,778,337 4-Bid & Award 5% P.001904 Tropical ES SMART Program Renovations (CC-A) \$ 1,787,337 4-Bid & Award 20% P.001952 Village ES SMART Program Renovations (CC-A) \$ 1,587,919 4-Bid & Award 15% P.002000 Virginia Shuman Young Montessori SMART Program Renovations \$ 3,335,530 4-Bid & Award 5% P.002074 Watkins ES SMART Program Renovations \$ 3,120,000 4-Bid & Award 5% P.002131 Westglades MS SMART Program Renovations \$ 4,702,994 4-Bid & Award 5% P.001965 Renovations (CC-A) \$ 614,512 4-Bid & Award 5%		<u> </u>	_			
P.001905 Stirling ES SMART Program Renovations (CC-A) \$ 4,376,295 4-Bid & Award 70% P.001808 Tedder ES SMART Program Renovations \$ 4,215,616 4-Bid & Award 15% P.001674 Thurgood Marshall ES SMART Program Renovations \$ 4,560,979 4-Bid & Award 5% P.002129 Tradewinds ES SMART Program Renovations \$ 3,704,874 4-Bid & Award 5% P.001904 Tropical ES SMART Program Renovations (CC-A) \$ 1,778,337 4-Bid & Award 20% P.001952 Village ES SMART Program Renovations (CC-A) \$ 1,587,919 4-Bid & Award 15% P.002000 Virginia Shuman Young Montessori SMART Program Renovations \$ 3,335,530 4-Bid & Award 5% P.002074 Watkins ES SMART Program Renovations \$ 3,120,000 4-Bid & Award 5% P.002131 Westglades MS SMART Program Renovations \$ 4,702,994 4-Bid & Award 5% P.002043 Westpine MS SMART Program Renovations \$ 5,830,151 4-Bid & Award 5% P.001965 Renovations (CC-A) \$ 614,512 4-Bid & Award 15% P.001741		<u> </u>	_			
P.001808 Tedder ES SMART Program Renovations \$ 4,215,616 4-Bid & Award 15%			_			
P.001674 Thurgood Marshall ES SMART Program Renovations \$ 4,560,979 4-Bid & Award 5%			_			
P.002129 Tradewinds ES SMART Program Renovations \$ 3,704,874 4-Bid & Award 5% P.001904 Tropical ES SMART Program Renovations (CC-A) \$ 1,778,337 4-Bid & Award 20% P.001952 Village ES SMART Program Renovations (CC-A) \$ 1,587,919 4-Bid & Award 15% P.002000 Virginia Shuman Young Montessori SMART Program Renovations \$ 3,335,530 4-Bid & Award 5% P.002074 Watkins ES SMART Program Renovations \$ 3,120,000 4-Bid & Award 5% P.002131 Westglades MS SMART Program Renovations \$ 4,702,994 4-Bid & Award 5% P.002043 Westpine MS SMART Program Renovations \$ 5,830,151 4-Bid & Award 5% William T. McFatter Technical Broward Fire Academy SMART Program P.001965 Renovations (CC-A) \$ 614,512 4-Bid & Award 91% P.001941 Wingate Oaks Center SMART Program Renovations \$ 5,831,000 4-Bid & Award 15% P.001981 Winston Park ES SMART Program Renovations \$ 2,681,000 4-Bid & Award 5% Phase 5 - Construction P.001728 Annabel C. Perry PK8 SMART Program Renovations \$ 5,015,037 5-Construction 92% Atlantic Technical College Arthur Ashe SMART Program Renovations \$ 3,078,449 5-Construction 85%			_			
P.001904 Tropical ES SMART Program Renovations (CC-A) \$ 1,778,337 4-Bid & Award 20% P.001952 Village ES SMART Program Renovations (CC-A) \$ 1,587,919 4-Bid & Award 15% P.002000 Virginia Shuman Young Montessori SMART Program Renovations \$ 3,335,530 4-Bid & Award 5% P.002074 Watkins ES SMART Program Renovations \$ 3,120,000 4-Bid & Award 5% P.002131 Westglades MS SMART Program Renovations \$ 4,702,994 4-Bid & Award 5% P.002043 Westpine MS SMART Program Renovations \$ 5,830,151 4-Bid & Award 5% William T. McFatter Technical Broward Fire Academy SMART Program P.001965 Renovations (CC-A) \$ 614,512 4-Bid & Award 91% P.001741 Wingate Oaks Center SMART Program Renovations \$ 5,831,000 4-Bid & Award 15% P.001981 Winston Park ES SMART Program Renovations \$ 2,681,000 4-Bid & Award 5% Phase 5 - Construction P.001728 Annabel C. Perry PK8 SMART Program Renovations \$ 5,015,037 5-Construction 92% Atlantic Technical College Arthur Ashe SMART Program Renovations \$ 3,078,449 5-Construction 85%			_			
P.001952 Village ES SMART Program Renovations (CC-A) \$ 1,587,919 4-Bid & Award 15% P.002000 Virginia Shuman Young Montessori SMART Program Renovations \$ 3,335,530 4-Bid & Award 5% P.002074 Watkins ES SMART Program Renovations \$ 3,120,000 4-Bid & Award 5% P.002131 Westglades MS SMART Program Renovations \$ 4,702,994 4-Bid & Award 5% P.002043 Westpine MS SMART Program Renovations \$ 5,830,151 4-Bid & Award 5% William T. McFatter Technical Broward Fire Academy SMART Program P.001965 Renovations (CC-A) \$ 614,512 4-Bid & Award 91% P.001741 Wingate Oaks Center SMART Program Renovations \$ 5,831,000 4-Bid & Award 15% P.001981 Winston Park ES SMART Program Renovations \$ 2,681,000 4-Bid & Award 5% Phase 5 - Construction P.001728 Annabel C. Perry PK8 SMART Program Renovations \$ 5,015,037 5-Construction 92% Atlantic Technical College Arthur Ashe SMART Program Renovations \$ 3,078,449 5-Construction 85%		•	_			
P.002000 Virginia Shuman Young Montessori SMART Program Renovations \$ 3,335,530 4-Bid & Award 5% P.002074 Watkins ES SMART Program Renovations \$ 3,120,000 4-Bid & Award 5% P.002131 Westglades MS SMART Program Renovations \$ 4,702,994 4-Bid & Award 5% P.002043 Westpine MS SMART Program Renovations \$ 5,830,151 4-Bid & Award 5% William T. McFatter Technical Broward Fire Academy SMART Program P.001965 Renovations (CC-A) \$ 614,512 4-Bid & Award 91% P.001741 Wingate Oaks Center SMART Program Renovations \$ 5,831,000 4-Bid & Award 15% P.001981 Winston Park ES SMART Program Renovations \$ 2,681,000 4-Bid & Award 5% P.001728 Annabel C. Perry PK8 SMART Program Renovations \$ 5,015,037 5-Construction 92% Atlantic Technical College Arthur Ashe SMART Program Renovations \$ 3,078,449 5-Construction 85%			_			
P.002074 Watkins ES SMART Program Renovations \$ 3,120,000 4-Bid & Award 5% P.002131 Westglades MS SMART Program Renovations \$ 4,702,994 4-Bid & Award 5% P.002043 Westpine MS SMART Program Renovations \$ 5,830,151 4-Bid & Award 5% William T. McFatter Technical Broward Fire Academy SMART Program P.001965 Renovations (CC-A) \$ 614,512 4-Bid & Award 91% P.001741 Wingate Oaks Center SMART Program Renovations \$ 5,831,000 4-Bid & Award 15% P.001981 Winston Park ES SMART Program Renovations \$ 2,681,000 4-Bid & Award 5% Phase 5 - Construction P.001728 Annabel C. Perry PK8 SMART Program Renovations \$ 5,015,037 5-Construction 92% Atlantic Technical College Arthur Ashe SMART Program Renovations \$ 3,078,449 5-Construction 85%	P.001952	Village ES SMART Program Renovations (CC-A)	\$	1,587,919	4-Bid & Award	15%
P.002074 Watkins ES SMART Program Renovations \$ 3,120,000 4-Bid & Award 5% P.002131 Westglades MS SMART Program Renovations \$ 4,702,994 4-Bid & Award 5% P.002043 Westpine MS SMART Program Renovations \$ 5,830,151 4-Bid & Award 5% William T. McFatter Technical Broward Fire Academy SMART Program P.001965 Renovations (CC-A) \$ 614,512 4-Bid & Award 91% P.001741 Wingate Oaks Center SMART Program Renovations \$ 5,831,000 4-Bid & Award 15% P.001981 Winston Park ES SMART Program Renovations \$ 2,681,000 4-Bid & Award 5% Phase 5 - Construction P.001728 Annabel C. Perry PK8 SMART Program Renovations \$ 5,015,037 5-Construction 92% Atlantic Technical College Arthur Ashe SMART Program Renovations \$ 3,078,449 5-Construction 85%						
P.002131 Westglades MS SMART Program Renovations \$ 4,702,994 4-Bid & Award 5% P.002043 Westpine MS SMART Program Renovations \$ 5,830,151 4-Bid & Award 5% William T. McFatter Technical Broward Fire Academy SMART Program P.001965 Renovations (CC-A) \$ 614,512 4-Bid & Award 91% P.001741 Wingate Oaks Center SMART Program Renovations \$ 5,831,000 4-Bid & Award 15% P.001981 Winston Park ES SMART Program Renovations \$ 2,681,000 4-Bid & Award 5% Phase 5 - Construction P.001728 Annabel C. Perry PK8 SMART Program Renovations \$ 5,015,037 5-Construction 92% Atlantic Technical College Arthur Ashe SMART Program Renovations \$ 3,078,449 5-Construction 85%	P.002000	Virginia Shuman Young Montessori SMART Program Renovations		3,335,530	4-Bid & Award	5%
P.002043 Westpine MS SMART Program Renovations \$ 5,830,151 4-Bid & Award 5% William T. McFatter Technical Broward Fire Academy SMART Program P.001965 Renovations (CC-A) \$ 614,512 4-Bid & Award 91% P.001741 Wingate Oaks Center SMART Program Renovations \$ 5,831,000 4-Bid & Award 15% P.001981 Winston Park ES SMART Program Renovations \$ 2,681,000 4-Bid & Award 5% Phase 5 - Construction P.001728 Annabel C. Perry PK8 SMART Program Renovations \$ 5,015,037 5-Construction 92% Atlantic Technical College Arthur Ashe SMART Program Renovations \$ 3,078,449 5-Construction 85%	P.002074	Watkins ES SMART Program Renovations	\$	3,120,000	4-Bid & Award	5%
William T. McFatter Technical Broward Fire Academy SMART Program P.001965 Renovations (CC-A) \$ 614,512 4-Bid & Award 91% P.001741 Wingate Oaks Center SMART Program Renovations \$ 5,831,000 4-Bid & Award 15% P.001981 Winston Park ES SMART Program Renovations \$ 2,681,000 4-Bid & Award 5% Phase 5 - Construction P.001728 Annabel C. Perry PK8 SMART Program Renovations \$ 5,015,037 5-Construction 92% Atlantic Technical College Arthur Ashe SMART Program Renovations (CC-A) \$ 3,078,449 5-Construction 85%	P.002131	Westglades MS SMART Program Renovations	\$	4,702,994	4-Bid & Award	5%
P.001965 Renovations (CC-A) \$ 614,512 4-Bid & Award 91% P.001741 Wingate Oaks Center SMART Program Renovations \$ 5,831,000 4-Bid & Award 15% P.001981 Winston Park ES SMART Program Renovations \$ 2,681,000 4-Bid & Award 5% Phase 5 - Construction P.001728 Annabel C. Perry PK8 SMART Program Renovations \$ 5,015,037 5-Construction 92% Atlantic Technical College Arthur Ashe SMART Program Renovations (CC-A) \$ 3,078,449 5-Construction 85%	P.002043	Westpine MS SMART Program Renovations	\$	5,830,151	4-Bid & Award	5%
P.001741 Wingate Oaks Center SMART Program Renovations \$ 5,831,000 4-Bid & Award 15% P.001981 Winston Park ES SMART Program Renovations \$ 2,681,000 4-Bid & Award 5% Phase 5 - Construction P.001728 Annabel C. Perry PK8 SMART Program Renovations \$ 5,015,037 5-Construction 92% Atlantic Technical College Arthur Ashe SMART Program Renovations (CC-A) \$ 3,078,449 5-Construction 85%		William T. McFatter Technical Broward Fire Academy SMART Program				
P.001981 Winston Park ES SMART Program Renovations \$ 2,681,000 4-Bid & Award 5% Phase 5 - Construction P.001728 Annabel C. Perry PK8 SMART Program Renovations \$ 5,015,037 5-Construction 92% Atlantic Technical College Arthur Ashe SMART Program Renovations (CC-A) \$ 3,078,449 5-Construction 85%	P.001965	Renovations (CC-A)	\$	614,512	4-Bid & Award	91%
P.001981 Winston Park ES SMART Program Renovations \$ 2,681,000 4-Bid & Award 5% Phase 5 - Construction P.001728 Annabel C. Perry PK8 SMART Program Renovations \$ 5,015,037 5-Construction 92% Atlantic Technical College Arthur Ashe SMART Program Renovations (CC-A) \$ 3,078,449 5-Construction 85%	P.001741					15%
Phase 5 - Construction P.001728 Annabel C. Perry PK8 SMART Program Renovations \$ 5,015,037 5-Construction 92% Atlantic Technical College Arthur Ashe SMART Program Renovations (CC-A) \$ 3,078,449 5-Construction 85%	P.001981		_			5%
P.001728 Annabel C. Perry PK8 SMART Program Renovations \$ 5,015,037 5-Construction 92% Atlantic Technical College Arthur Ashe SMART Program Renovations (CC-A) \$ 3,078,449 5-Construction 85%		-			<u> </u>	
Atlantic Technical College Arthur Ashe SMART Program Renovations P.001959 (CC-A) \$ 3,078,449 5-Construction 85%			4	E 01 F 02 7	E Construction	030/
P.001959 (CC-A) \$ 3,078,449 5-Construction 85%	P.UU1/28		>	5,015,03/	5-Construction	92%
	D 004050		_	2 272 775		0.55
P.UU1944 Banyan ES SMART Program Renovations (CC-A) \$ 2,205,979 5-Construction 50%		· · · · ·	_			
	P.001944	Banyan ES SMART Program Renovations (CC-A)	Ş	2,205,979	5-Construction	50%

D 001706	Daysians FC CAMART Programs Removations (CC A)	۲,	2 600 720	Construction	050/
P.001786 P.001646	Bayview ES SMART Program Renovations (CC-A)	\$ \$		5-Construction 5-Construction	95% 38%
	Blanche Ely HS SMART Program Renovations (CMAR)	_			
P.001974	Bright Horizons Center SMART Program Renovations (CC-A)	\$		5-Construction	8%
P.001661	Castle Hill ES SMART Program Renovations (CC-A)	\$		5-Construction	45%
P.001732	Chapel Trail ES SMART Program Renovations (CC-A)	\$		5-Construction	1%
P.001413	Coconut Creek ES SMART Program Renovations	\$	5,044,761	5-Construction	99%
D 004007	Colbert Museum Magnet (fka Colbert ES) SMART Program		4 =00 000		120/
P.001937	Renovations (CC-A)	\$	1,590,903	5-Construction	42%
	Cypress Bay HS SMART Program Renovations (CMAR)(Ph 1 -				
P.001774	Renovations)	\$	32,578,000	5-Construction	85%
	Cypress Bay HS SMART Program Renovations (CMAR)(Ph 2 -				2004
P.001774-2	Classroom Addition)		l Above	5-Construction	22%
P.001412	Cypress ES SMART Program Renovations	\$		5-Construction	99%
P.001972	Dave Thomas EC East SMART Program Renovations	\$		5-Construction	3%
P.001726	Dillard 6-12 SMART Program Renovations	\$		5-Construction	30%
P.002118	Discovery ES SMART Program Renovations (T&B)	\$	150,000	5-Construction	85%
P.001662	Dr. MLK Montessori Academy SMART Program Renovations (CC-A)	\$		5-Construction	85%
P.001722	Eagle Ridge ES SMART Program Renovations	\$		5-Construction	99%
P.001897	Embassy Creek ES SMART Program Renovations	\$		5-Construction	1%
P.001948	Everglades ES SMART Program Renovations (CC-A)	\$	2,344,500	5-Construction	10%
P.001985	Everglades HS SMART Program Renovations	\$	6,376,254	5-Construction	6%
P.001785	Fairway ES SMART Program Renovations	\$	7,510,900	5-Construction	25%
P.001902	Falcon Cove MS SMART Program Renovations (CMAR)	\$	22,788,000	5-Construction	10%
P.001865	Forest Glen MS SMART Program Renovations	\$	9,047,800	5-Construction	10%
P.001926	Forest Hills ES SMART Program Renovations (CC-A)	\$	2,419,601	5-Construction	25%
P.001839	Fort Lauderdale HS SMART Program Renovations	\$	3,772,887	5-Construction	9%
P.001863	Gator Run ES SMART Program Renovations	\$	4,116,323	5-Construction	25%
P.001745	Griffin ES SMART Program Renovations	\$	4,126,208	5-Construction	85%
	Gulfstream Academy of Hallandale Beach (FKA Hallandale Adult &				
P.001822	Community) SMART Program Renovations	\$	5,161,700	5-Construction	44%
P.001784	Hawkes Bluff ES SMART Program Renovations	\$		5-Construction	1%
P.001806	Hollywood Hills HS SMART Program Renovations (CMAR)	\$		5-Construction	14%
P.001743	James S. Rickards MS SMART Program Renovations	\$		5-Construction	8%
P.001886	Lake Forest ES SMART Program Renovations (CC-A)	\$		5-Construction	10%
P.001637	Lauderdale Lakes MS SMART Program Renovations	\$		5-Construction	70%
P.001999	Liberty ES SMART Program Renovations	\$		5-Construction	80%
P.001998	Maplewood ES Media Center (Bundled bid with P.001639)	\$		5-Construction	0%
P.001639	Maplewood ES SMART Program Renovations	\$		5-Construction	45%
P.002185	Marjory Stoneman Douglas HS Building Replacement (CMAR)		T SMART	5-Construction	40%
P.001964	McNab ES SMART Program Renovations (CC-A)	\$		5-Construction	15%
P.001904 P.002011	Mirror Lake ES SMART Program Renovations (CC-A)	\$		5-Construction	5%
P.002011 P.001996	Morrow ES SMART Program Renovations (CC-A)	\$		5-Construction	80%
	North Side ES SMART Program Renovations (CC-A)	\$			
P.001992		_		5-Construction	10%
P.001817	Nova HS SMART Program Renovations (CMAR)	\$		5-Construction	19%
P.001895	Oakland Park ES SMART Program Renovations	\$		5-Construction	90%
P.001712	Oakridge ES SMART Program Renovations	\$		5-Construction	25%
P.001864	Pembroke Pines ES SMART Program Renovations	\$		5-Construction	1%
P.001949	Pinewood ES SMART Program Renovations (CC-A)	\$		5-Construction	20%
P.001793	Pioneer MS SMART Program Renovations	\$		5-Construction	18%
P.002119	Plantation ES SMART Program Renovations (T&B)	\$		5-Construction	80%
P.001713	Pompano Beach ES SMART Program Renovations	\$		5-Construction	85%
P.001721	Pompano Beach MS SMART Program Renovations	\$		5-Construction	40%
P.001754	Quiet Waters ES SMART Program Renovations	\$		5-Construction	57%
P.001725	Ramblewood ES SMART Program Renovations	\$		5-Construction	30%
P.001866	Riverglades ES SMART Program Renovations	\$		5-Construction	4%
P.001987	Riverland ES SMART Program Renovations	\$	4,057,192	5-Construction	1%
P.001950	Rock Island ES SMART Program Renovations (CC-A)	\$	2,306,944	5-Construction	1%
P.001924	Sandpiper ES SMART Program Renovations (CC-A)	\$	921,942	5-Construction	45%
F.001324					

Seagull Alternative HS SMART Program Renovations (CC-A)	\$	2,455,082	5-Construction	22%
Silver Lakes ES SMART Program Renovations (CC-A)	\$	2,249,741	5-Construction	95%
Silver Ridge ES SMART Program Renovations	\$	3,032,700	5-Construction	10%
Silver Trail MS SMART Program Renovations	\$	6,203,150	5-Construction	57%
Stranahan HS SMART Program Renovations (CMAR)	\$	28,491,577	5-Construction	36%
Sunland Park Academy SMART Program Renovations (CC-A)	\$	1,379,100	5-Construction	71%
Sunrise MS SMART Program Renovations	\$	6,656,050	5-Construction	1%
Sunset Lakes ES SMART Program Renovations	\$	2,991,500	5-Construction	1%
Tamarac ES Media Center (CC-A)	\$	389,716	5-Construction	55%
Tamarac ES SMART Program Renovations	\$	2,462,657	5-Construction	56%
The Quest Center SMART Program Renovations (CC-A)(CC-CMAR)	\$	1,688,000	5-Construction	53%
Walker ES SMART Program Renovations (CC-A)	\$	3,428,090	5-Construction	5%
West Broward HS SMART Program Renovations (T&B)	\$	438,000	5-Construction	40%
West Hollywood ES SMART Program Renovations	\$	3,910,160	5-Construction	40%
Westchester ES SMART Program Renovations	\$	2,998,000	5-Construction	2%
Westwood Heights ES SMART Program Renovations (CC-A)	\$	4,237,269	5-Construction	80%
William E. Dandy MS SMART Program Renovations	\$	7,175,104	5-Construction	85%
	Silver Lakes ES SMART Program Renovations (CC-A) Silver Ridge ES SMART Program Renovations Silver Trail MS SMART Program Renovations Stranahan HS SMART Program Renovations (CMAR) Sunland Park Academy SMART Program Renovations (CC-A) Sunrise MS SMART Program Renovations Sunset Lakes ES SMART Program Renovations Tamarac ES Media Center (CC-A) Tamarac ES SMART Program Renovations The Quest Center SMART Program Renovations (CC-A)(CC-CMAR) Walker ES SMART Program Renovations (CC-A) West Broward HS SMART Program Renovations (T&B) West Hollywood ES SMART Program Renovations Westchester ES SMART Program Renovations Westwood Heights ES SMART Program Renovations (CC-A)	Silver Lakes ES SMART Program Renovations (CC-A) \$ Silver Ridge ES SMART Program Renovations \$ Silver Trail MS SMART Program Renovations \$ Stranahan HS SMART Program Renovations (CMAR) \$ Sunland Park Academy SMART Program Renovations (CC-A) \$ Sunrise MS SMART Program Renovations \$ Sunset Lakes ES SMART Program Renovations \$ Tamarac ES Media Center (CC-A) \$ Tamarac ES SMART Program Renovations \$ Walker ES SMART Program Renovations \$ Walker ES SMART Program Renovations (CC-A)(CC-CMAR) \$ West Broward HS SMART Program Renovations (T&B) \$ West Hollywood ES SMART Program Renovations \$ Westchester ES SMART Program Renovations \$ Westwood Heights ES SMART Program Renovations (CC-A) \$ Westwood Heights ES SMART Program Renovations (CC-A) \$ Westwood Heights ES SMART Program Renovations (CC-A) \$ Westwood Heights ES SMART Program Renovations (CC-A)	Silver Lakes ES SMART Program Renovations (CC-A)\$ 2,249,741Silver Ridge ES SMART Program Renovations\$ 3,032,700Silver Trail MS SMART Program Renovations\$ 6,203,150Stranahan HS SMART Program Renovations (CMAR)\$ 28,491,577Sunland Park Academy SMART Program Renovations (CC-A)\$ 1,379,100Sunrise MS SMART Program Renovations\$ 6,656,050Sunset Lakes ES SMART Program Renovations\$ 2,991,500Tamarac ES Media Center (CC-A)\$ 389,716Tamarac ES SMART Program Renovations\$ 2,462,657The Quest Center SMART Program Renovations (CC-A)\$ 3,428,000Walker ES SMART Program Renovations (CC-A)\$ 3,428,000West Broward HS SMART Program Renovations (T&B)\$ 438,000West Hollywood ES SMART Program Renovations\$ 3,910,160Westchester ES SMART Program Renovations\$ 2,998,000Westwood Heights ES SMART Program Renovations (CC-A)\$ 4,237,269	Silver Lakes ES SMART Program Renovations\$ 2,249,7415-ConstructionSilver Ridge ES SMART Program Renovations\$ 3,032,7005-ConstructionSilver Trail MS SMART Program Renovations\$ 6,203,1505-ConstructionStranahan HS SMART Program Renovations (CMAR)\$ 28,491,5775-ConstructionSunland Park Academy SMART Program Renovations (CC-A)\$ 1,379,1005-ConstructionSunrise MS SMART Program Renovations\$ 6,656,0505-ConstructionSunset Lakes ES SMART Program Renovations\$ 2,991,5005-ConstructionTamarac ES Media Center (CC-A)\$ 389,7165-ConstructionTamarac ES SMART Program Renovations\$ 2,462,6575-ConstructionThe Quest Center SMART Program Renovations (CC-A)\$ 1,688,0005-ConstructionWelker ES SMART Program Renovations (CC-A)\$ 3,428,0905-ConstructionWest Broward HS SMART Program Renovations (T&B)\$ 438,0005-ConstructionWest Hollywood ES SMART Program Renovations\$ 3,910,1605-ConstructionWestchester ES SMART Program Renovations\$ 2,998,0005-ConstructionWestwood Heights ES SMART Program Renovations (CC-A)\$ 4,237,2695-Construction

 Vendor ______
 Score out of #: _____

Program Manager - Owner's Representative (PMOR) Evaluation					
Deliverables	Exceeds Expectations 8 – 10 Points	Meets Expectations 4 – 7 Points	Falls Below Expectations 0 – 3 Points	Score	
1. Project Charters	Project Charters accompany every project and are invariably of high quality. Scope, timelines, responsibilities, and resources are accurately identified. Project Charters are thoroughly validated with key stakeholders at Project Charter meetings. Scope verification rarely uncovers inaccuracies. As a tool for keeping stakeholder expectations aligned, the Project Charters are indispensable.	High quality Project Charters accompany every project. Scope, timelines, responsibilities, and resources are accurately identified. Project Charters are sufficiently reviewed with key stakeholders. Inaccuracies uncovered during scope verification are typically minor.	Project Charters are inconsistent in quality. Producing a Project Charter is generally a compliance activity, produced only when requested. In many cases, there is no Project Charter associated with a project.		
2. Schedule Controls	Baseline schedules are completed on-time or ahead of schedule and are reliably error-free. Monthly Schedule Updates are fully transparent, accurate and on-time. Cost Loaded Schedules are accurate and on-time. There is high confidence that schedule controls capture true project status. Schedule controls flag risks early on such that corrective actions can be implemented successfully.	Baseline Schedules are completed with few inaccuracies. Monthly Schedule Updates are provided on-time and with few inaccuracies. Cost-Loaded Schedules are completed on-time and with few inaccuracies. Where inaccuracies are identified, they are corrected within a reasonable amount of time and used to inform continuous improvement measures. Schedule controls reliably capture project risks.	Baseline Schedules are often late, incomplete or inaccurate. Monthly Schedule Updates are late, incomplete or inaccurate. Cost Loaded Schedules are late, incomplete or inaccurate. There is little confidence that schedule controls accurately reflect what is happening in the field.		
3. Budgeting & Cost Control Process	Scope verification occurs before projects are advertised, accompanied by new budgets that reflect scope changes. The control process is highly responsive to scope adjustments.	Project Charters are high quality, accurately scoped and presented regularly at Project Charter meetings. Scope verification occurs before projects are advertised. Budget impact is clearly articulated and transparent.	Project Charters often miss key components such that the scope of work is often not correct. Stakeholders are not well informed of project scope or scope changes. The impact of scope changes on budget is often miscalculated or not transparent.		
4. Cost Estimating	Cost estimates based on construction documents are highly accurate and reliably within 3% of actual cost.	Cost estimates based on construction documents are reliably within +/- 10% of actual cost.	Cost estimates based on construction documents are unreliable and regularly deviate +/- 20% from actual cost.		
5. Change Management	Changes are identified and communicated in near real-time. The impact of changes is accurately calculated, and approvals for changes are regularly obtained on first pass. Options are well articulated; they objectively and convincingly present factors affecting cost, benefit, and value associated with each option.	Impact of changes is clearly communicated. Approval for changes is obtained in a timely manner. Change updates are provided at least once per month. Options are well articulated such that value vs. cost considerations are understood.	Change management is inconsistent. Often the impact of changes is not well communicated. Sometimes changes are misrepresented. Options do not sufficiently identify factors driving cost, benefit, or value.		
6. Cash Flow & Funding Source Management	Highly accurate expenditure projections are provided monthly. They are aligned with the Master Schedule and responsive to changes. Cash outflows are tightly coupled with project progress.	There is confidence that expenditure projections are accurate. They are provided on a monthly basis and provide a fair and reasonable view into the status of the capital program. Vendor terms are reliably honored.	There is reasonable concern that the gap between cash outflows and project progress is too large. Expenditure projections are often far off from actual capital program outlays.		
7. Bond Oversight Report: Quarterly	Quarterly reports are delivered on- time and transparent. Report formats are user-friendly and effectively highlight key areas of progress or concern. Schedule and budget data are accurate. Stakeholders are consistently satisfied and rarely express concerns over completeness, transparency, or accuracy.	Quarterly reports are delivered on-time and transparent. Report formats are user-friendly and effectively highlight key areas of progress or concern. Schedule and budget data are accurate. Stakeholders are consistently satisfied. Where concerns are raised, they are addressed in a timely manner.	Quarterly reports are not always provided on-time. The format is confusing, with information often missing, inaccurate, or hard to find. There is little confidence that the reports provide a fair portrait of program progress.		

8. Project	Vendor evaluations are always	Vendor evaluations are mostly	Vendor evaluations are not
Vendor	performed and consistently on time	performed on time at intervals defined	performed consistently on time at
Evaluations	at the intervals defined in SOP and	in the SOP and available for review, in	the intervals defined in SOP in
	readily available in eBuilder in	regard to technical performance of	regard to technical performance of
	regard to technical performance of	vendors to deliver scope of the project,	vendors to deliver scope of the
	vendors to deliver scope of the	at highest quality on schedule and	project, at highest quality on
	project, at highest quality on	within budget. Vendors are made	schedule and within budget.
	schedule and within budget.	aware of deficiencies and where	Vendors are not made aware of
	Vendors are made aware of	warranted put on notice of non-	deficiencies and where warranted
	deficiencies and where warranted	compliance	put on notice of non-compliance.
	put on notice of non-compliance		

Attributes	Exceeds Expectations 8 – 10 Points	Meets Expectations 4 – 7 Points	Falls Below Expectations 0 – 3 Points	Score
1. Communications Effectiveness	Communications are accurate, timely, and without omissions. The messages being conveyed are easily understood, and there is high confidence that communications are transparent. Delivery and format are clear, comprehensive, straightforward. Risks are identified early. The frequency of communications anticipates when information is needed and how it will be used.	Communications are mostly accurate and provided in a reasonably timely manner with few omissions. Risks are accurately identified and communicated. The information being conveyed is adequate and available when requested. Delivery and format are professional.	Communications are often misleading, inaccurate, or omit essential details. The delivery, format, and frequency of communications are often not reliable.	
2. Utilization of Tools	Tool functionality is fully exploited to bring discernible benefit to the District as evidenced by the quality and timeliness of the work performed. Tools are an essential enabler for tracking and reporting capabilities that extend beyond base requirements. Staff are well-trained on tools and use their knowledge to drive further improvements in tool functionality.	Tracking and reporting tools are used effectively to support all District requirements. Personnel are well trained on tools, and the timeliness and quality of their work is enhanced by tool functionality.	It is not clear how tool usage is improving quality or timeliness of work performed. Tool usage is peripheral to the work being done, and much of the work is still done manually.	
3. Leadership	Has excellent interpersonal skills. Inspires others to perform consistently at or above expectations at all times. Creates strong teams that are empowered to solve problems and adapt rapidly to changing conditions. Staff are motivated. Demonstrates a clear method behind problem-solving and consistently articulates more than one potential solution. Has a strong track record for making sound decisions?	Effectively sets clear expectations and acts quickly when expectations are not met. Demonstrates willingness to tackle difficult situations. Facilitates productive communications with stakeholders. Demonstrates a clear method behind problem-solving. Has a good track record for making sound decisions.	Not effective at managing teams. Expectations are unclear. Staff are not motivated or empowered to solve problems on their own. There is high staff turnover. Decisions appear to be made haphazardly. Has a spotty track record for making sound decisions.	
4. Customer Responsiveness	Understands and effectively leverages the District's institutional knowledge. Anticipates District requests and needs. Demonstrates a mastery of District processes. Always responds to requests in a timely and professional manner. Demonstrates a dedication to continuous improvement.	Response times are uniformly satisfactory and professional. There is high confidence that commitments to the District will be honored. There is evidence of efforts at continuous improvement.	Is often slow to respond to District requests. Often must be reminded to follow through with commitments made to the District. There is little evidence of a commitment to continuous improvement.	

Comments:	